chapter S-2.1, r. 4

Last amendment: 20 February 2025 Updated to 20 February 2025

Safety Code for the construction industry

Québec 🕷

Dépôt légal – 2025 Bibliothèque et Archives nationales du Québec

ISBN 978-2-551-27201-3

© Éditeur officiel du Québec, 2025

All rights reserved in all countries. No part of this publication may be translated, used or reproduced by any means, wether electronic or mechanical, including microreproduction, without the written authorization of the Éditeur officiel du Québec.

COMMISSION DES NORMES, DE L'ÉQUITÉ, DE LA SANTÉ ET DE LA SÉCURITÉ DU TRAVAIL

General information

1 844 838-0808

https://www.cnesst.gouv.qc.ca/en/client-services/ contact-us

Presentation

The Publisher has incorporated into the Regulation any amendments made between the date of the first publication in the *Gazette officielle du Québec*, Part 2, or Part 1 as the case may be, and the date of publication of this document.

This list of the amendments appears below. The date of coming into force is indicated in parentheses.

This publication has no official status.

Amendments which came into force since 1 January 1982:

Decision, 17 November 1983 (1984) G.O. 2, 849 (cf 84-02-08); (eff. 83-05-14); O.C. 21-85, 9 January 1985 (1985) G.O. 2, 331 (cf 85-02-03); O.C. 1959-86, 16 December 1986 (1987) G.O. 2, 188 (cf 87-01-24); O.C. 1960-86, 16 December 1986 (1987) G.O. 2, 202 (cf 87-01-24); O.C. 53-90, 17 January 1990 (1990) G.O. 2, 333 (cf 90-02-15); O.C. 54-90, 17 January 1990 (1990) G.O. 2, 335 (cf 90-02-15); O.C. 995-91, 10 July 1991 (1991) G.O. 2, 2913 (cf 91-08-15); O.C. 807-92, 27 May 1992 (1992) G.O. 2, 2869 (general, cf 92-07-02); (s. 4 par. 2 and s. 5 par. 4, cf 93-07-02); O.C. 213-93, 17 February 1993 (1993) G.O. 2, 1757 (cf 93-04-01);

O.C. 329-94, 9 March 1994 (1994) G.O. 2, 1335 (general, cf 94-04-21); (ss. 35 and 40, cf 95-04-21); S.Q. 1996, c. 74, section 54 (cf 96-12-23); O.C. 1279-98, 30 September 1998 (1998) G.O. 2, 4259 (cf 98-11-26); O.C. 1413-98, 28 October 1998 (1998) G.O. 2, 4441 (cf 98-11-26); O.C. 459-99, 21 April 1999 (1999) G.O. 2, 1115 (cf 99-05-20); O.C. 35-2001, 17 January 2001 (2001) G.O. 2, 1035 (cf 01-02-15, a. 15: 02-01-01); O.C. 885-2001, 4 July 2001 (2001) G.O. 2, 3888 (cf 01-08-02); O.C. 873-2003, 20 August 2003 (2003) G.O. 2, 2729 (cf 03-09-18); O.C. 119-2008, 13 February 2008 (2008) G.O. 2, 682 (cf 08-03-13); O.C. 425-2010, 12 May 2010 (2010) G.O. 2, 1313 (cf 10-06-10); O.C. 391-2011, 6 April 2011 (2011) G.O. 2, 970 (cf 11-05-05); O.C. 393-2011, 6 April 2011 (2011) G.O. 2, 975 (cf 11-05-05); O.C. 476-2013, 8 May 2013, 2013 G.O. 2, 1255 (cf 13-06-06); O.C. 606-2014, 18 June 2014, 2014 G.O. 2, 1413 (cf 14-07-17); O.C. 57-2015, 28 January 2015, 2015 G.O. 2, 94 (cf 15-02-26); S.Q. 2015, c. 13 (cf 15-06-03); O.C. 428-2015, 20 May 2015, 2015 G.O. 2, 1001 (cf 15-06-18); O.C. 513-2015, 10 June 2015 2015 G.O. 2, 1086 (cf 15-07-08);

O.C. 1005-2015, 11 November 2015 2015 G.O. 2, 3023 (cf 15-12-10);

O.C. 1078-2015, 2 December 2015 2015 G.O. 2, 3277 (cf 15-12-31); O.C. 1186-2015, 16 December 2015 2015 G.O. 2, 3477 (cf 16-01-14);

S.Q. 2015, c. 15, s. 237 (cf 16-01-01);

S.Q. 2016, c. 25, s. 45 (cf 16-11-10);

S.Q. 2018, c. 19, s. 19 (cf 18-10-17);

O.C. 640-2019, 19 June 2019 2019 G.O. 2, 1094 (cf 19-07-18);

O.C. 805-2020, 8 July 2020 2020 G.O. 2, 2007 (cf 20-08-06);

O.C. 890-2020, 19 August 2020 2020 G.O. 2, 2329 (cf 20-09-17);

O.C. 483-2021, 24 March 2021 2021 G.O. 2, 1207 (cf 21-04-22);

O.C. 48-2022, 12 January 2022 2022 G.O. 2, 187 (cf 22-02-10);

O.C. 645-2022, 30 March 2022 2022 G.O. 2, 1057 (cf 22-04-28);

O.C. 646-2022, 30 March 2022 2022 G.O. 2, 1059 (cf 22-10-28);

S.Q. 2021, c. 27, s. 267 (cf 23-01-01);

O.C. 820-2023, 10 May 2023 2023 G.O. 2, 923 (cf 23-06-08);

O.C. 781-2021, 2 June 2021 2021 G.O. 2, 167 (cf 23-06-16);

O.C. 1112-2023, 28 June 2023 2023 G.O. 2, 1776 (cf 23-07-27);

O.C. 820-2023, 10 May 2023 2023 G.O. 2, 923 s. 3.25.6 (part) (cf 23-12-09);

O.C. 1393-2024, 3 September 2024 2024 G.O. 2, 3562 (cf 24-10-03);

O.C. 63-2025, 23 January 2025 2025 G.O. 2, 555 (cf 25-02-20)

SAFETY CODE FOR THE CONSTRUCTION INDUSTRY

TABLE OF CONTENTS

page

DIVISION I – DEFINITIONS1				
DIVISION II – GENERAL PROVISIONS7				
2.1. 2.2. 2.3. 2.4. 2.5.	Scope			
2.6. 2.7. 2.8. 2.9. 2.10. 2.11. 2.12.	(Revoked). 12 Public safety			
2.13. 2.14. 2.15. 2.16. 2.17. 2.18. 2.19	demolition 25 Shoring parts 26 Damaged buildings 27 Hoisting apparatus 28 Piles, sheet-piling 39 (Revoked). 39 Transportation of workers 39 Prohibited work method and technique 42			
2.20.	Lockout and other energy control methods 42			
2.21.	Noise			
DIVISI	ON III – CONSTRUCTION SITES53			
3.1. 3.2. 3.3. 3.4. 3.5. 3.6. 3.7.	Access to construction site			
3.8. 3.9. 3.10. 3.11. 3.12. 3.13. 3.14.	and platforms 64 Guardrails 66 Scaffoldings 67 Construction equipment 80 Temporary heating 91 Steam supply 92 Supply of gas under pressure 93 Welding and cutting 94			

	page
3.15.	Diggings, excavations and trenches
3.16.	Handling and storage of materials
3.17.	(Revoked).
3.18.	Demolition103
3.19.	Work on stilts
3.20.	Sandblasting106
3.21.	Work in confined space 107
3.22.	Work in isolated locations108
3.23.	Work liable to produce
3.24.	asbestos dust emissions
3.25.	Work liable to produce crystalline silica dust
	cillissions
DIVISIO	ON IV
– HANE	DLING AND USE OF EXPLOSIVES 127
4.1.	General provisions
4.2.	Shot-firer's certificate
4.3.	Transportation of explosives
4.4.	Storage of explosives
4.5.	Drilling
4.6.	Loading of blastholes
4.7.	Firing140
4.8.	Work after blasting142
4.9.	(Revoked).
DIVISIO	ON V – WORK IN THE PROXIMITY
OF ELEC	CIRICAL LINES
5.1.	Interdictions 143
5.2.	Warning sign 145
5.5.	warning sign 145
DIVISIO CONCR	DN VI – SHORING OF ETE FORMWORK146
61	Shoring drawing 146
6.2	Concreting work 146
6.3.	Exception 146
6.4.	Materials
6.5.	Bearing surface
6.6.	Support parts
6.7.	Bracing
6.8.	Shoring of multistory structures150
6.9.	Shoring, dismantling and deforming150
DIVISIO	DN VII – PORTABLE TOOLS150
7.0.	General 150
7.1.	Special provisions relating
	to certain portable tools
7.1.1.	Explosive actuated tool
7.1.2.	Nailing gun
7.1.3.	Saws
7.1.3.1.	Circular saws
7.1.3.2.	Chainsaws

- 7.2. (Replaced).
- (Replaced). (Replaced). (Replaced. 7.3.
- 7.4.
- (Replaced). 7.5. 7.6.
- (Replaced).

DIVISION VIII

– UNDE	RGROUND WORK SITES	
8.1.	Fire prevention and protection	
8.2.	Ground stability	159
8.3.	Ventilation and air quality	
8.4.	Explosives	
8.5.	Drilling	
8.6.	Transportation of personnel and	
	materials in a shaft	
8.7.	Traffic	
8.8.	Loading and transportation	167
8.9.	Hygiene	
8.10.	Lighting	
8.11.	Communication and standard signals .	
8.12.	Emergency measures	170
8.13.	Minimum age	

DIVISION IX – (Revoked).

DIVISION X

– WORI	K ON ROADS OPEN TO TRAFFIC	174
10.1. 10.2. 10.3. 10.4.	Definitions Scope Traffic signs Other precautions	174 174 174 175
DIVISIO – WORK	ON XI X ABOVE OR NEAR WATER	176
SCHED NOT PE	U LE 0.1 – INCLINE OF A LADDER RMANENTLY FASTENED	179
SCHED SCAFFC WHEN A	ULE 0.2 – ANCHORS OF A DLDING LESS THAN 18 METRES A TARPAULIN OR NET IS USED	180
SCHED	ULE 1 – (Revoked).	
SCHED EXPLOS	ULE 2 – HANDLING AND USE OF SIVES	184
SCHED	ULE 3 – (Revoked).	
SCHED	ULE 4 – CODE OF SIGNALS	193
SCHED	ULE 5-PORTABLE GUARDRAILS	195
SCHED	ULE 6 –	196

SCHEDULE 7 – (Revoked).

page

SCHEDULE 8 – TRAINING AND CERTIFICA	ΔTE
OF A LOW VELOCITY EXPLOSIVE	
ACTUATED TOOL OPERATOR	197
SCHEDULE 9 – MOBILE CRANE LOG	198
SCHEDULE 10 – TOWER CRANE LOG	203
	200
INDEX	209
INDEX PROVISIONS LISTED IN SECTION 2 OF T	209 HE
INDEX PROVISIONS LISTED IN SECTION 2 OF T REGULATION RESPECTING OCCUPATIO	209 HE DNAL
INDEX PROVISIONS LISTED IN SECTION 2 OF T REGULATION RESPECTING OCCUPATIO HEALTH AND SAFETY AND APPLICABLE	209 HE DNAL 2 TO
INDEX PROVISIONS LISTED IN SECTION 2 OF T REGULATION RESPECTING OCCUPATION HEALTH AND SAFETY AND APPLICABLE CONSTRUCTION SITES	HE DNAL TO 251
INDEX PROVISIONS LISTED IN SECTION 2 OF T REGULATION RESPECTING OCCUPATIO HEALTH AND SAFETY AND APPLICABLE CONSTRUCTION SITES REGULATION RESPECTING	HE DNAL 2 TO 251
INDEX PROVISIONS LISTED IN SECTION 2 OF T REGULATION RESPECTING OCCUPATIO HEALTH AND SAFETY AND APPLICABLE CONSTRUCTION SITES REGULATION RESPECTING PREVENTION MECHANIMS SPECIFIC	HE DNAL 2 TO 251



Safety Code for the construction industry

chapter S-2.1, r. 4

An Act respecting occupational health and safety (chapter S-2.1, s. 223)

DIVISION I

DEFINITIONS

1.1. In this Code, unless otherwise indicated,

(1) "blasting accessory" means any explosive substance used in firing explosives;

(1.1) "CSA" means the Canadian Standards Association;

(1.2) "asbestos" means the fibrous form of mineral silicates belonging to rock-forming minerals of the serpentine group, that is, chrysotile, and the amphibole group, that is, actinolite, amosite, anthophyllite, crocidolite, tremolite or any mixture containing one or more of those minerals;

(1.3) "backup area" means a marked out space reserved for backup maneuvers of self-propelled vehicles;

(2) "material hoisting apparatus" means an apparatus designed to hoist materials, such as a crane, an overhead travelling crane, a platform lift truck or a fork lift truck;

(2.1) "personnel hoisting apparatus" means an apparatus designed to hoist people, such as a vehicle mounted aerial device, an articulated or telescopic self-propelled platform lift or a self-propelled scissor platform lift;

lift;(3) "ASTM" means the American Society for Testing and Materials;

(3.1) "impulse noise" means a noise of short duration (generally of less than one second), peaking at a high level and characterized by a sharp increase and rapid decrease in sound level. The parameter used to measure an impulse noise is the C-weighted peak sound pressure level;

(4) (paragraph revoked);

- (5) (paragraph revoked);
- (6) (paragraph revoked);

(7) "bolting" means a method for staying partitionwalls by means of bolts or other similar devices anchored in the surrounding rock; (7.0.0) "calculator" means a calculation tool that can be used to evaluate the daily noise exposure level ($L_{\rm EX,8h}$ or $L_{\rm ex,8h}$) for the purpose of reducing the duration of workers' daily exposure to noise;

(7.0) "CAN/CSA" means the Canadian Standards Association;

(7.1) "life line" means a synthetic fibre rope, a steel wire rope or a strap attached to an anchorage system and used to guide a rope grab;

(8) "high risk construction site" means

(a) an excavation site of 6 m or more in depth;

(b) a trench site, within the meaning of paragraph 34 of this section, 50 m or more in length;

(c) a site where water or sewer works are carried out over a length of 50 m or more;

(d) an underground work site;

(e) where work is carried out underwater or in a hyperbaric environment;

(f) a demolition site;

(g) the site of a building, a structure or part of a structure 15 m or more in height;

(*h*) a site where electrical transmission lines or their supporting structures are constructed or repaired;

(*i*) a site where work is carried out 3 m or less from an electrical line of over 750 V;

(j) a site where work is carried out above or near water;

(k) dredging works;

(*l*) a site where work in an electrical powerhouse or in an electrical transformer station is carried out;

(m) a site where work is carried out in a closed area;

(n) a site where explosives are used and handled;

(*o*) where high-risk work is carried out within the meaning of paragraph 3 of section 3.23.2 of this Code;

(8.1) "major construction site" means a site where not fewer than 500 workers are employed simultaneously;

(9) "underground work site" means a construction site where works related to the excavation of tunnels and shafts are performed;

(10) "rated load" means the maximum load established by the manufacturer;

(11) "non-combustible construction" means a construction whose structural members, including floors and joints, are made of non-combustible materials;

(12) (paragraph revoked);

(12.0) "lanyard" means a rope or strap fastened at one end to a safety harness and at the other end to an anchorage system or other component of a fall arrest connecting device;

(12.1) "digging" means any hole dug in the ground, including an excavation or a trench;

(13) "depot" means a building, construction or receptacle in which explosives are stored;

(13.1) (paragraph revoked);

(13.2) "dBA" means an A-weighted decibel measurement - the weighting reduces the significance of extreme frequencies, in particular low frequencies below 200 Hz, and increases the significance of frequencies around 2 500 Hz. A-weighting must be used for all measurements to evaluate $L_{\text{EX,8h}}$ or $L_{\text{ex,8h}}$;

(13.3) "dBC" means a C-weighted decibel measurement - the weighting reduces the significance of frequencies at or below 31 Hz and increases the significance of frequencies at or above 8 000 Hz. C-weighting must be used for all measurements to evaluate peak sound pressure level;

(14) "scaling" has the meaning assigned by paragraph 31;

(14.1) "pump jack scaffold" means a mast-climbing work platform that moves between two masts with the help of jacks;

(14.2) "mast-climbing work platform" means scaffolding constituted of a work platform that moves up and down between two or several masts with the help of a hoisting system and equipped with an anchoring system;

(14.3) "winch scaffolding" means a mast-climbing work platform whose masts are linked by braces or counter braces supporting a work platform that moves with the help of a hoisting device constituted of winches, pulleys and cables;

(15) "scaffold shoring" means an assembly of tubular scaffolding frames used for the shoring of concrete formwork;

(15.01) "motorized scaffolding" means a mast-climbing work platform constituted of a hoisting system made up of an electric, pneumatic or hydraulic motor or a natural gas or gasoline engine;

(15.1) "suspended scaffolding" means a platform suspended by one or more cables fixed to an anchoring point and moved vertically by means of a manual or motorized winch;

(16) (paragraph revoked);

(17) "electrical power company" means a person, firm, company, cooperative society or municipality operating an electric energy transportation or distribution network;

(17.1) "confined space" means a space which is not designed to be occupied by a human being, in particular a tank, silo, vat, caisson, shoring pile, stack or manhole;

(17.2) "non-destructive examination" means a radiographic, ultrasonic, magnetic particle or liquid penetrant examination carried out and interpreted by an equipment operator for non-destructive testing certified level 2 by the Natural Resources Canada National Non-Destructive Testing Certification Body under CAN/CGSB-48.9712 Standard, Non-Destructive Testing - Qualification and Certification of Personnel;

(18) "excavation" means a piece of land dug out to a depth of at least 1.2 m where the width of the base exceeds the depth;

(18.1) "explosives" means any substance fabricated, manufactured or used to produce an explosion or a detonation, such as gunpowder, blasting powder, dynamite, an explosive in solution, aqueous gelatin, a blasting agent or a blasting accessory;

(19) (paragraph revoked);

(20) "manufacturer" means the maker or his agent or where manufacture is done on the site, the employer;

(21) "safety factor" means the ratio of the ultimate load to the working load;

(21.1) "respirable asbestos fibre" means asbestos fibre having a ratio of length to diameter of more than 3:1; only fibres longer than 5 μ m must be taken into account for measurement purposes;

(21.2) "high-efficiency filter" means a filter capable of filtering particles of 0.3 μ m in size at an efficiency rate of at least 99.97%;

(22) "face" means the most advanced work point in a given direction in an underground excavation at a given level;

(23) (paragraph revoked);

(23.0) "free fall distance" means the vertical distance measured from the beginning of a fall, from the harness D-ring to which the fall arrest connecting device is attached, to the point where the fall arrest system begins to apply force to stop the fall;

(23.1) "ISO" means the International Organization for Standardization;

(24) "jumbo" means a hammer-carrying head designed for drilling at the face without having to repeatedly disassemble drilling machines;

(24.0) "fall arrest connecting device": all equipment, such as a lanyard, energy absorber, snap hook, connector, life line or rope grab, used to secure a safety harness to an anchorage system;

(24.1) (paragraph revoked);

(24.2) "friable material" means material that can be crumbled, pulverized or powdered by hand pressure when dry or that is crumbled, pulverized or powdered;

(25) "non-combustible material" means material that complies with CAN4-S114, Standard Method of Test for Determination of Non-Combustibility in Building Materials, applicable at the time of manufacture of the equipment;

(25.01) "dike" means a sandbag barricade, an earth mound or the equivalent located less than 50 cm from the depot and whose height is at least equal to the depot;

(25.1) "protective wall" means a partition made of plywood at least 9 mm thick or made of another rigid material of an equivalent or greater resistance, that is at least 1.8 m high and that is installed at not more than 100 mm above the ground;

(25.2) "NF EN" means the European standard, the French version of which (NF) is published in France by the Association française de normalisation;

(26) "NFPA" means the National Fire Protection Association;

(26.1) "daily noise exposure level" means the equivalent continuous sound pressure level (dBA) for an 8-hour working day. It results from measurements that include all the types of noise present, including impulse noises;

(26.2) "equivalent continuous sound pressure level (dBA)" means the A-weighted continuous sound pressure level measured over a given period of time. It is identical to the sound pressure level of a constant noise having the same total A-weighted sound energy over the same period of time. It results from measurements that include all the types of noise present, including impulse noises. In the formulas used to calculate daily noise exposure level, it corresponds to $L_{p,A,eqTe}$ or $L_{eq,i}$, which is the A-weighted equivalent continuous sound pressure level for the duration of the working day in hours (*Te* or T_w);

(26.3) "certified organization" means an organization certified by the Canadian Welding Bureau in accordance with the specifications of CSA Standard W178.1, Certification of Welding Inspection Organizations;

This section has been renumeroted.

(27) "two-blocking" means a situation where the hook block, the hoisting cable counterweight or other accessories attached to the hoisting cable come into contact with the tip of the boom or with the tip of the jib;

(28) "blasting mat" means a covering used to offer protection against stones or other objects that may be projected as a result of firing explosives; (28.1) "load-bearing part" means a part which bears or supports the loads inherent to the use of equipment;

(29) "low velocity explosive actuated tool" means any tool designed so that when it is used with the maximum explosive charge permitted by the manufacturer's specifications, it imparts to the fastener a speed of not more than 91.4 m per second and transmits to it kinetic energy of not more than 33 joules measured at a distance between 1.99 to 2.01 m from the mouth of the tool;

(29.01) "mobile elevating work platform" means an apparatus intended for moving persons, tools and materials to a working position and consisting of at least an extensible structure, a chassis and a work platform equipped with controls;

(29.1) "asbestos dust" means airborne asbestos particles or deposited asbestos particles liable to become airborne in the work area;

(29.2) "peak sound pressure level" means the instantaneous peak sound pressure level measured in C-weighted decibels;

(30) "shaft" means a passage dug underground whose longitudinal axis makes an angle greater than 20° with the horizontal;

(31) "scaling" means an action provoking the fall of rocks which tend to become separated from a partition wall or a vault;

(31.01) "SAE" means the Society of Automotive Engineers;

(31.1) "boatswain's chair" means a seat designed for a worker, supported by slings or by a structure and suspended from a cable fixed to an anchoring point;

(32) "hazard signs" means a yellow, orange or red rigid strip, such as a trestle, or flexible strip, at least 70 mm wide and installed, according to the configuration of the land or of the structure, at a height varying between 0.7 m and 1.2 m

(33) "fragile surface" means a surface not intended to support the weight of a worker, in particular a skylight, a canopy, a sunshade or a drywall ceiling;

(33.1) "flushing toilet" means a toilet having the following features:

(a) the bowl has a hatch or a siphon which separates it physically and visually from the drain pipe or the processing tank;

(b) waste is purged from the bowl toward the sewer system or into the processing tank by means of a device causing a flow of water or chemicals; (33.2) "chemical toilet" means a toilet without a bowl in which waste falls directly into a tank containing a chemical product to process the waste;

(34) "trench" means a piece of land dug to a depth of at least 1.2 m where the width of the base is equal or less than the depth. The width of the base is measured between the excavated walls or between an excavated wall and a structure;

(34.0.1) "work above or near water" means work carried out above or less than 2 m from a body of water or watercourse that is either more than 1.2 m deep and allows the use of a boat, or with a water flow of more than 0.51 m/s sufficient to carry a person;

(34.1) "blasthole" means a boring hole intended to be loaded with explosives;

(34.2) "misfire" means a blasthole loaded with explosives which did not fire;

(35) "tunnel" means an underground passage constructed without removing the material forming the vault and whose longitudinal axis makes an angle of 20° or less in relation to the horizontal;

(36) "ULC" means the Underwriters' Laboratories of Canada.

(36.1) "self-propelled vehicle" means any motor vehicle mounted on wheels, tracks or rails used to transport persons, objects or materials, or to pull or push trailers or materials, except an all-terrain vehicle or a hoisting apparatus;

(37) "loading area" means space that includes the place where workers load the drill holes, drill holes loaded or being loaded and any space occupied by the material and equipment necessary for the loading;

(38) "blasting area" means a place or space that presents a projection or blast hazard to a person or where a hazard exists because of the effects of the blasting.

```
R.R.Q., 1981, c. S-2.1, r. 6, s. 1.1; Decision 83-11-17, s. 1;
O.C. 1959-86, s. 1; O.C. 53-90, s. 1; O.C. 54-90, s. 1; O.C. 995-91, s. 1;
O.C. 807-92, s. 1; O.C. 329-94, s. 1; O.C. 1413-98, s. 1; O.C. 35-2001, s. 1;
O.C. 119-2008, s. 1; O.C. 425-2010, s. 4; O.C. 393-2011, s. 1;O.C. 476-
2013, s. 4; O.C. 606-2014, s. 1; O.C. 57-2015, s. 1; O.C. 428-2015, s. 1;
O.C. 513-2015, s. 1; O.C. 1078-2015, s. 1; O.C. 483-2021, s. 1; O.C. 646-
2022, s. 1; O.C. 781-2021, s. 1; O.C. 1393-2024, s. 1; O.C. 63-2025, s. 1.
```

DIVISION II

GENERAL PROVISIONS

§2.1. Scope

2.1.1. This Code applies to any work carried out on a construction site within the meaning of the Act respecting occupational health and safety (chapter S-2.1), except for facilities provided to workers by the employer for

administrative, lodging, eating or recreational purposes, to which this Code applies only where a formal provision exists to that effect.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.1.1; Decision 83-11-17, s. 2; O.C. 393-2011, s. 2.

2.1.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.1.2; O.C. 1959-86, s. 2.

§2.2. Responsibility and powers of inspectors

2.2.1.-2.2.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 2.

2.2.4. (*Revoked*).

Decision 83-11-17, s. 3; S.Q. 2021, c. 27, s. 267.

2.2.5. (*Revoked*).

Decision 83-11-17, s. 3; O.C. 329-94, s. 3.

2.2.6. To ascertain the soundness of a construction or installation, the inspector may require an attestation to that effect signed and sealed by an engineer or an architect.

O.C. 119-2008, s. 2.

§2.3. Equivalences

2.3.1. In the application of this Code, the nature, dimensions and arrangement of materials may differ from the rules established as long as the strength of materials and their use ensure as much safety as the prescribed rule.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.3.1.

2.3.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.3.2; O.C. 1959-86, s. 2.

§2.4. General obligations

2.4.1. (1) The principal contractor must notify the Commission des normes, de l'équité, de la santé et de la sécurité du travail in writing of the opening of a construction site at least 10 days before activities at the site begin.

The principal contractor must also notify the Commission in writing of the closing of a construction site at least 10 days before the work at the site is expected to end, except where the expected duration of the site is 1 month or less, in which case the Commission must be notified at least 10 days before activities at the site begin.

This subsection does not apply to major construction sites.

(1.1) The prescribed notice must state the following:

(a) the identification number assigned to the principal contractor by the Commission;

(b) the name and address of the principal contractor;

(c) the name and address of the owner if different from the principal contractor, and those of the architects, consulting engineers, superintendents and employers at the site;

(d) the municipal address of the site, its cadastral lot number and its location in relation to the nearest public road;

(e) the type of construction site;

(f) whether the planned construction site is a high risk construction site;

(g) the opening date of the construction site;

(*h*) the expected duration of the construction site;

(i) the anticipated number of workers;

(*j*) whether the expected duration of the construction site is 1 month or less, the expected closing date or the date when the work is expected to be completed;

(k) in the case of asbestos removal work or demolition work involving asbestos, the methods and procedures used as well as an attestation to the existence of a training and information program that complies with section 3.23.7.

(1.2) For emergency repairs to waterworks, a sewer, an electrical transmission or distribution line or gas pipeline, the Commission must be informed in writing, by telex, telegram or messenger, of the particulars prescribed in subsection 1.1 as soon as possible, where such work involves 10 or more workers.

No notice is required where such work involves fewer than 10 workers.

(1.3) For demolition work ordered by the public fire prevention service in the interests of public safety, the Commission must be informed in writing, by telex, telegram or messenger, of the particulars prescribed in subsection 1.1 at least 6 hours before the work begins.

(2) Before work mentioned hereunder begins, the employer shall send to the Commission all plans including the installation and disassembling procedures signed and sealed by an engineer of the manufacturer, respecting:

(a) the shoring of an excavation or trench 6 m or deeper;

(b) the shoring of concrete formwork;

(c) (paragraph revoked);

(d) any wood scaffolding 9 m or more in height;

(e) any metal scaffolding 18 m or more in height;

(f) any outrigger scaffolding or suspended outrigger scaffolding extending out more than 2.4 m from the finished face of a building;

(g) any temporary runway or platform designed to support workers and that is part of the forms;

(*h*) any platform, bucket or basket attached to a hoisting device for the transport of persons;

(i) any scaffolding used or installed on a vehicle or any equipment which may be moved;

(*j*) any anchoring device used for the installation of prefabricated concrete slabs or prefabricated buildings;

(*k*) any spreader bar used for the installation of prefabricated concrete slabs or prefabricated buildings;

(l) any suspended scaffolding or boatswain's chair;

(*m*) any mast-climbing work platform that must be anchored, except in the case of a pump jack scaffold.

(3) Before installing or putting up a tower crane, a material hoist, a job-site elevator or a mast-climbing transport platform, the employer must send to the Commission the installation plans signed and sealed by an engineer. These plans must also include the disassembling procedure.

(4) The copies of the compliance certificate signed by an engineer shall be sent to the Commission before the work begins.

(5) A certified copy of the plans mentioned in subsections 2 and 3, or of the certificates mentioned in subsection 4 shall be available at all times on the construction site.

(6) The employer shall:

(a) (paragraph revoked);

(b) ensure the good operation of safety devices and fire prevention apparatus;

(c) (paragraph revoked);

(d) ensure that any area of great depth has an easy means of evacuation and a ventilation system, where there is a possibility of gas, vapours or other dangerous substances or contaminants accumulating.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.4.1; Decision, 83-11-17, s. 4; O.C. 1959-86, s. 4; O.C. 54-90, s. 2; O.C. 807-92, s. 2; O.C. 119-2008, s. 3; O.C. 1005-2015, s. 1; O.C. 1393-2024, s. 2.

2.4.2. The employer shall ensure that:

(a) all necessary steps are taken to ensure public safety and the safety of workers;

(b) any worker on a construction site wears at all times clothing completely covering his torso and back;

(c) (paragraph revoked);

(d) workers do not involve themselves in games or competitions while at work;

- (e) paragraph revoked);
- (f) any worker knows:

i. this Code;

ii. any operation ensuring the good running order of apparatus and machinery for which he is responsible;

iii. the emergency measures to observe in case of fire, explosion, or other accidents;

(g) any worker is informed of the inherent dangers in his work;

(*h*) (paragraph revoked);

(*i*) management and control staff working mainly and usually on a construction site and workers working on a construction site who, on 18 July 2019, are not holders of a certificate issued by the Commission or by a body recognized by the Commission and have successfully completed the course on health and general safety on construction sites.

However, before 8 June 2023 a natural person who, to obtain a contractor's licence or to qualify a partnership or legal person for a licence, has passed the examination on construction site safety management skills required by the Regulation respecting the professional qualification of contractors and owner-builders (chapter B-1.1, r. 9) or who is exempted therefrom by that regulation or by a regulation under section 182 of the Building Act (chapter B-1.1), shall be exempted from successfully completing the course on health and general safety on construction sites;

(*j*) on an outlying construction site, all the required accommodations are provided for the inspector on the duration of this inspection work;

(k) the apparatus, equipment or tools used on a construction site conform to this Code.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.4.2; Decision, 83-11-17, s. 5; O.C. 21-85, s. 1; O.C. 1959-86, s. 5; O.C. 329-94, s. 4; S.Q. 1996, c. 74, s. 54; O.C. 1413-98, s. 2; S.Q. 2019, c. 19, s. 19; O.C. 640-2019, s. 1; O.C. 820-2023, s. 1.

2.4.3. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.4.3; O.C. 1959-86, s. 6.

2.4.4. On a construction site, traffic control, use of public ways, temporary electrical installation, housekeeping of sites, toilets and their accessories, public safety, access to the site, protection against fire, permanent ramps and guardrails, temporary heating, transportation, rescue on water, rescue following a fall and all other general safety measures are the responsibility of the principal contractor.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.4.4; O.C. 1959-86, s. 7; O.C. 428-2015, s. 2; O.C. 513-2015, s. 2; O.C. 63-2025, s. 2.

§2.5. (Revoked).

2.5.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.5.1; S.Q. 2021, c. 27, s. 267.

2.5.2. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.5.2; S.Q. 2021, c. 27, s. 267.

2.5.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.5.3; O.C. 1959-86, s. 8; S.Q. 2021, c. 27, s. 267.

2.5.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.5.4; O.C. 1959-86, s. 9; O.C. 606-2014, s. 2; S.Q. 2021, c. 27, s. 267.

§2.6 (Revoked).

2.6.1.-2.6.2. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 10.

§2.7. Public safety

2.7.1. Protective measure: A construction site shall be separated from any place to which the public has access by

(a) a covered passage if the works are carried out on more than one floor and if the sidewalk or the public way is less than 2 m from the site or from the projection of the latter at the level of the sidewalk or the public way; or

(b) protective wall if the sidewalk or the public ways is 2 m or more away and should there be any danger to the pedestrians.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.7.1; O.C. 1413-98, s. 3; O.C. 35-2001, s. 2.

2.7.2. Characteristics of covered passage: The covered passage shall:

(a) have a clear height of at least 2.1 m;

(b) have a clear width of at least 1.5 m or equal to the width of the sidewalk, whichever is less;

(c) be so designed and built as to support safely any load which may be reasonably applied to it. However, the passage shall be able to support a load at least 2,400 N/m² applied on the roof;

(d) have a weatherproof roof sloped towards the site;

(e) be totally enclosed on the site side and present a smooth surface on the inside of the passage;

(f) have a guardrail of 1,070 mm in height on the street side, when the passage is supported on posts on this side; and

(g) be sufficiently lighted when the street is lighted. R.R.Q., 1981, c. S-2.1, r. 6, s. 2.7.2; O.C. 35-2001, s. 3.

2.7.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.7.3; O.C. 1413-98, s. 4; O.C. 35-2001, s. 4.

§2.8. Traffic control on a construction site

2.8.1. General responsibilities of the principal contractor: Traffic of self-propelled vehicles must be controlled in order to protect any person on a site. To that end, the principal contractor must plan the traffic of those vehicles so as to restrict backup maneuvers and set up new safety measures to protect any person circulating on the site. The principal contractor must also give prior information of the safety measures prescribed to any person who must circulate on the site.

The principal contractor is responsible for ensuring that signs, including maximum speed indicators, are set up. The principal contractor must mark off traffic lanes, backup areas and work areas, as the case may be. The principal contractor is also responsible for ensuring that dust is kept down on all roads.

Where it is foreseen that activities on the construction site will occupy at least 10 construction workers simultaneously at a particular stage of the work, the principal contractor must, before work begins, develop a traffic plan that complies with the requirements of section 2.8.2.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.8.1; O.C. 329-94, s. 5; O.C. 1078-2015, s. 2.

2.8.2. Traffic plan: A traffic plan must indicate the safety measures taken to restrict backup maneuvers and those set up to protect persons circulating on a site. It must also determine the bidirectional telecommunications procedures or the code of hand signals related to backup maneuvers.

It must also contain a diagram indicating

(1) the location and size of all traffic lanes;

(2) the location of backup areas, if any;

(3) traffic signs;

(4) allowed maximum speeds;

(5) the positioning of a site signal person or flag person for road users.

The plan must be available at all times on the work site. The information it contains must be updated in the event of any change, in particular as regards the location of the backup areas.

O.C. 1078-2015, s. 3.

2.8.3. Training of site signal persons: Site signal persons direct operators of self-propelled vehicles, namely during backup maneuvers. Site signal persons must undergo training, given by an instructor, that deals with the following elements:

(1) the risks associated with the circulation of persons and self-propelled vehicles on a site; (2) the traffic rules and safety instructions on a site, including those provided for in the traffic plan, the marking of traffic zones and the instructions necessary to perform their task;

(3) work equipment appropriate for the signal person's duties such as high-visibility safety apparel and bidirectional means of telecommunications;

(4) the signal person's role and responsibilities;

(5) the positioning of site signal persons and the blind spots of self-propelled vehicles;

(6) the means of communication and the code of hand signals related to backup maneuvers.

O.C. 1078-2015, s. 3.

2.8.4. Site signal persons: Where site signal persons perform their duties, they must meet the following conditions:

(1) wear fluorescent yellow-green high-visibility safety apparel of Class 2 or 3 and of Level 2 that complies with CSA Standard Z96, High-Visibility Safety Apparel;

(2) use one of the means of communication provided for in the traffic plan and that was taught to them during training;

(3) remain visible to operators of self-propelled vehicles directed by them and stay clear of the trajectory of those vehicles.

O.C. 1078-2015, s. 3.

2.8.5. Backup maneuver: Where it is necessary for a self-propelled vehicle referred to in subsection 2 of section 3.10.12 to perform a backup maneuver in a zone where persons are present or circulating and where that backup maneuver may endanger their safety, the maneuver must be performed in a backup area where no person may walk or with a site signal person who must direct the operator throughout the maneuver.

Where a backup maneuver is directed by a signal person, the signal person must use a bidirectional means of telecommunications to guide the operator. Despite the foregoing, where the vehicle backs up over a distance of less than 10 metres, the signal person may use the code of hand signals indicated in the traffic plan, as the case may be.

O.C. 1078-2015, s. 3.

§2.9. Fall protection

2.9.1. Installation of guardrails: Whether or not workers are present, guardrails must be placed at a maximum distance of 300 mm from the open sides of every area, including the sides of a floor or a roof, from which workers could fall

(1) into a dangerous liquid or substance;

(2) onto a moving component;

(3) on equipment or material that constitute a danger;

(4) from a height of 1.2 m or more where they use a vehicle;

(5) from a height of 1.5 m or more where handling a load; or

(6) from a height greater than 3 m in other cases.

This section applies, with the necessary modifications, where a worker is at risk of falling through a fragile surface.

The use of a guardrail as a means of protection against falls is prohibited on a work surface with an incline greater than 19° (4/12).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.9.1; O.C. 329-94, s. 6; O.C. 35-2001, s. 5; O.C. 606-2014, s. 3; O.C. 63-2025, s. 3.

2.9.2. Exception: Despite the foregoing, where a guardrail is installed and a portion thereof must be removed during work because it is a nuisance or where it is prohibited or unachievable to install one, in particular on a ladder or a step-ladder, every worker must be protected using one of the following means of protection in the order of precedence below:

(1) by changing the process or the work position so that workers may work on the ground or on another surface from which they are not at risk of falling;

(2) by using a travel restraint system in accordance with section 2.10.16;

(3) by installing a safety net in accordance with section 2.9.3;

(4) by the worker wearing a safety harness secured to an anchorage system by a fall arrest connecting device, in accordance with sections 2.10.12 and 2.10.15.

The work area must then be delimited by a continuous barrier or trestles of a minimum height of 0.7 m, located at a distance varying between 0.9 m and 1.2 m from the place where workers are at risk of falling, or by a warning line complying with the requirements of section 2.9.4.1, to prevent access to the work area by persons not working therein.

In the case referred to in subparagraph 4 of the first paragraph, where the worker cannot remain in position without the help of the fall arrest connecting device, a means of positioning, such as a plank on brackets, a positioning tether or a strap, a suspension cable or a platform must be used.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.9.2; O.C. 995-91, s. 2; O.C. 35-2001, s. 5; O.C. 606-2014, s. 4; O.C. 63-2025, s. 4.

2.9.3. Safety net: Where a safety net is installed, it must:

(1) be installed in accordance with the manufacturer's instructions and so that, when used, the person who falls therein will not hit an obstacle under or above the net or be hit by an object;

(2) be placed as close as possible vertically to the work surface and so as to prevent a free fall of more than 3 m;

(3) be selected based on the environment in which it will be used so that it can resist damage that could be caused in particular by corrosion, welding or cutting operations, or weather-related conditions;

(4) be free of all foreign matter;

(5) have a notice indicating the manufacturer's name or trademark, the serial number, the year of manufacture and the minimum resistance;

(6) comply and be used in accordance with ANSI/ASSE Standard A10.11 Safety Requirements for Personnel and Debris Nets or standards NF EN 1263-1 Temporary Works Equipment - Safety Nets – Part 1: Safety Requirements, Test Methods and NF EN 1263-2 Safety Nets - Part 2: Safety Requirements for the Positioning Limits.

In case of conflict between the requirements provided for in the standards referred to in subparagraph 6 of the first paragraph and those provided for in this Regulation, the stricter standard applies.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.9.3; O.C. 995-91, s. 2; O.C. 35-2001, s. 5; O.C. 63-2025, s. 5.

2.9.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.9.4; O.C. 329-94, s. 6.

2.9.4.0. Despite section 2.9.1, a warning line complying with section 2.9.4.1 may be installed, during bridging or roofing work, on surfaces with a slope equal to or less than 15° (3/12), in order to replace the use of a guard-rail and delimit a work area.

In such a case, a means of protection against falls, in compliance with the hierarchy prescribed in the first paragraph of section 2.9.2, must be used by the worker outside the area delimited by the warning line.

O.C. 606-2014, s. 5; O.C. 63-2025, s. 6.

2.9.4.1. Warning line: A warning line must be

(1) continuous and installed on all sides of the work area that it delimits;

(2) placed at a distance of 2 m or more from any place where a worker may fall from a height;

(3) made of a rigid strip, a cable or a chain able to withstand a tractive force of at least 2.22 kN;

(4) equipped with flags made of high-visibility materials and placed at intervals of not more than 2 m;

(5) capable of withstanding a load of 100 N applied horizontally at the line's highest point or vertically at its midpoint between 2 stanchions;

(6) completed at each access point, storage area or hoisting area by a path formed by 2 parallel lines. However, when the path to a point of access to a work area is located at a distance of more than 5 m from it, the warning line does not have to be continued beyond that distance. In places where the access path starts at a roof edge, a guard-rail must be installed on the side of the roof, in compliance with section 2.9.1, so as to cover the first 3 metres on either side of the access path's starting point; and

(7) installed so that the line is

(a) located between 0.7 m above the work surface at the line's lowest point and 1.2 m above that surface at its highest point; and

(b) supported by stanchions placed at intervals of not more than 2.5 m;

(c) attached to each stanchion so that pushing on the line between 2 stanchions does not reduce the height of the line between adjacent stanchions by an equivalent amount.

O.C. 606-2014, s. 5; O.C. 63-2025, s. 7.

2.9.5. Any worker who, after a fall, is suspended in a safety harness or retained in a safety net must be rescued within a maximum of 15 minutes. The rescue procedures must prioritize the use of personnel hoists.

To that end, the principal contractor must develop a rescue procedure in cooperation with employers before the start of work requiring the use of a fall arrest safety harness or a safety net. The procedure must cover all activities on the construction site and be updated according to the evolution of the work on the site.

O.C. 63-2025, s. 8.

2.9.5.1. Before the start of the work referred to in the second paragraph of section 2.9.5, a training course on the rescue procedure provided for in that paragraph must be dispensed to the workers who will be required to carry it out.

The rescue procedure must be tested with drills that allow in particular for workers to become familiar with their roles, the communication protocol and the use of the determined rescue equipment. The complexity of the drills must vary according to the complexity of the work and the rescue to be performed.

The drills must be carried out before the start of the work referred to in the second paragraph of section 2.9.5 and repeated every 6 months for the duration of the work. $\overline{O.C. 63-2025. s. 8}$.

2.9.5.2. The principal contractor must ensure the availability on the construction site of the equipment necessary to perform a rescue following a fall for the duration of the work referred to in the second paragraph of section 2.9.5. The principal contractor must also ensure that at least 1 rescue attendant who was trained to rescue a worker suspended in a safety harness or retained in a safety net is present at all times on the work premises.

The training referred to in the first paragraph must be adapted to the complexity of the rescue to be performed.

The nature of the rescue attendant's work on the construction site must allow that person to intervene quickly and efficiently during a rescue. The rescue attendant may also act as first-aider if so provided in the rescue procedure.

O.C. 63-2025, s. 8.

2.9.5.3. Where the rescue is performed on ropes, the equipment must comply with standards NFPA 2500, ANSI Z359.4 or sections 2.10.12 and 2.10.15 and be available at all times on the construction site for the duration of the work referred to in the second paragraph of section 2.9.5.

O.C. 63-2025, s. 8.

§2.10. Personal protective equipment

2.10.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.1; O.C. 1959-86, s. 11.

2.10.2. In areas where there is danger of contact with moving parts:

(a) clothes shall fit well and have no loose parts;

(b) necklaces, bracelets or rings shall not be worn, except for medical bracelets which shall remain fastened to the wrist; and

(c) long hair shall be tucked under a bonnet in the hat. R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.2.

2.10.3. Protection of head: Any person on a construction site shall wear a safety hat designed and manufactured in accordance with CAN/CSA Standard Z94.1, Industrial protective headwear — Performance, selection, care, and use, applicable at the time of manufacturing.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.3; O.C. 890-2020, s. 1.

2.10.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.4; O.C. 329-94, s. 7.

2.10.5. Protection of eyes and face: Every worker must wear protective equipment that complies with the latest version of CAN/CSA Standard Z94.3 Eye and Face Protectors, within 24 months following its last update, when the worker's eyes or face are exposed to

- (a) moving particles;
- (b) hazardous substances;
- (c) intense light or heat radiation;
- (d) molten metal;
- (e) (paragraph revoked);
- (f) other risks of the same kind.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.5; O.C, 995-91, s. 3; O.C. 329-94, s. 8; O.C. 393-2011, s. 3.

2.10.6. Foot protection: Subject to section 2.1.1, every person on a construction site must wear class 1 protective footwear that complies with CAN/CSA Standard Z-195 Protective Footwear.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.6; O.C. 995-91, s. 4; O.C. 393-2011, s. 4.

2.10.7.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.1; O.C. 781-2021, s. 3.

2.10.7.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.2; O.C. 781-2021, s. 3.

2.10.7.3. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.3; O.C. 781-2021, s. 3.

2.10.7.4. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.4; O.C. 781-2021, s. 3.

2.10.7.5. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.5; O.C. 781-2021, s. 3.

2.10.7.6. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.6; O.C. 329-94, s. 11; O.C. 393-2011, s. 5; O.C. 781-2021, s. 3.

2.10.7.7. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.7; O.C. 781-2021, s. 3.

2.10.7.8. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.8; O.C. 393-2011, s. 6; O.C. 781-2021, s. 3.

2.10.7.9. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.7.9; O.C. 393-2011, s. 7; O.C. 781-2021, s. 3.

2.10.8. Protection of respiratory passages: The contaminants in the air of a work environment shall be eliminated at their emission point so that their concentration is reduced to a level lower than or equal to the permissible values indicated in Schedule I to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13).

The employer must provide a respirator in compliance with Division VI of the Regulation respecting occupational health and safety (chapter S-2.1, r. 13) in any situation where the values referred to in the first paragraph cannot be complied with.

During periods of work on equipment referred to in section 5 of the Regulation respecting occupational health and safety, or during temporary inspection or maintenance work or work of the same nature performed sporadically on another type of equipment or facility, an employer may provide such a device without the obligation to take other measures to eliminate or reduce contaminants.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.8; O.C. 885-2001, s. 366; O.C. 48-2022, s. 1.

2.10.9. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.9; O.C. 1960-86, s. 1; O.C. 329-94, s. 12; O.C. 885-2001, s. 367; O.C. 48-2022, s. 2.

2.10.10. Protection of hands:

(1) Mittens or gloves shall be worn in the handling of sharp-edged objects or objects with abrasive or rough surfaces, or of corrosive, poisonous or other substances which present similar risks.

(2) It is forbidden to wear mittens or gloves in the handling of mechanized apparatus except where there is a risk of frostbite. In such a case they shall:

(a) permit proper manipulation of the apparatus; and

- (b) be easily removable.
- (3) (paragraph revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.10; O.C. 483-2021, s. 2

2.10.11. Protection of the other parts of the body:

(1) A worker exposed to molten metal splashings or in contact with hazardous or infectious substances or using dangerous tools shall wear protective equipment such as hood, apron, leg pads and oversleeves.

(2) (Subparagraph revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.11; O.C. 329-94, s. 13.

2.10.12. Safety harness:

(1) A safety harness must comply with CSA Standard Z259.10 Full Body Harnesses and be secured to an anchorage system, in compliance with section 2.10.15, by a fall arrest connecting device that limits the maximum fall arrest force to 6 kN or the free fall distance to 1.8 m.

This fall arrest connecting device must consist of one or more of the following pieces of equipment, including as a minimum the equipment in subparagraph *a* or *b*:

(a) an energy absorber and a lanyard in compliance with CSA Standard Z259.11 Personal Energy Absorbers and Lanyards. The lanyard, including the energy absorber, must measure not more than 2 m in length;

(b) a self retracting lanyard in compliance with CSA Standard Z259.2.2 Self-Retracting Devices;

(c) a rope grab in compliance with CSA Standard Z259.2.5 Fall Arresters and Vertical Lifelines or CSA Standard Z259.2.4 Fall Arresters and Vertical Rigid Rails;

(d) a vertical life line in compliance with CSA Standard Z259.2.5 Fall Arresters and Vertical Lifelines or CSA Standard Z259.2.4 Fall Arresters and Vertical Rigid Rails.

A vertical life line must

i. be used by only 1 person;

ii. be shorter than 90 m; and

iii. never be brought into direct contact with a sharp edge.

(e) a connecting component, such as a spring hook, D-ring or snap hook in compliance with CAN/CSA Standard Z259.12 Connecting Components for Personal Fall Arrest Systems.

(2) (Revoked).

(3) Where a worker assigned to the erection or checking of power line towers wears a safety harness, the harness must be equipped with one of the following systems:

(a) an energy absorber to which are fastened 2 lanyards, including 1 that must be attached at all times;

(b) an energy absorber to which is fastened 1 lanyard attached by a rope grab to a vertical life line;

(c) a self retracting lanyard equipped with an energy absorber or fastened thereto.

Where the worker moves a life line or the sling of a self retracting lanyard by means of a pole anchor hook, the worker must be attached to the tower only by means of his or her work positioning strap or tether that the worker must fasten to a structural member above him or her.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.12; O.C. 1959-86, s. 12; O.C. 53-90, s. 2; O.C. 995-91, s. 5; O.C. 807-92, s. 3; O.C. 329-94, s. 14; O.C. 35-2001, s. 6; O.C. 606-2014, s. 6; O.C. 63-2025, s. 9.

2.10.13. Personal floatation device and life jacket:

(1) Any worker working above or near water must wear an individual floatation device or a life jacket during water transportation aboard an open boat or on the deck of a boat or in the case of a measure provided for in the rescue plan referred to in subparagraph 4 of the first paragraph of section 11.1. The foregoing also applies to a rescue attendant performing a rescue operation on water;

(2) a personal floatation device or a life jacket must

(a) be of the right size;

(b) be designed so as to keep head and face above water;

(c) enable to float without limb effort;

(d) be bright in colour and equipped with reflecting strips visible when in water;

(e) have a minimum floatability of 150 N (33 lbs) that is assured by buoyant materials or by an automatic inflation system activated by immersion;

(f) be approved by Transport Canada or by a body recognized by Transport Canada, as evidenced by the tag or approval stamp affixed to it.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.13; O.C. 329-94, s. 15; O.C. 513-2015, s. 3.

2.10.14. Safety belt: Where a worker is equipped with a safety belt, it can be used only to limit the movements of a worker or to keep him in his work position.

Such a belt must comply with CAN/CSA Standard Z259.1 Body Belts and Saddles for Work Positioning and Travel Restraint.

A safety belt shall not be used as individual protective equipment to stop the fall of a worker.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.10.14; O.C. 53-90, s. 3; O.C. 35-2001, s. 7; O.C. 606-2014, s. 7.

2.10.15. Anchorage system:

The fall arrest connecting device of a safety harness must be secured to

(1) a single point of anchorage with one of the following characteristics:

(a) a breaking strength of at least 18 kN; or

(b) designed and installed in accordance with an engineer's plan in compliance with CSA Standard Z259.16 Design of Active Fall-Protection Systems, and having one of the following characteristics:

i. a strength equal to twice the maximum fall arrest force as certified by an engineer; or

ii. certified in accordance with EN 795 Personal Protective Equipment against Falls – Anchor devices – published by the European Committee for Standardization or with CSA Standard Z259.15 Anchorage Connectors;

(2) a flexible continuous anchorage system (horizontal life line) with one of the following characteristics:

(a) in compliance with the following minimum standards:

i. a steel cable of a minimum diameter of 12 mm slackened to a minimum angle of 1 vertical to 12 horizontal, or 5° from horizontal;

ii. a maximum distance of 12 m between the end anchors;

iii. end anchors with a breaking strength of at least 90 kN;

iv. not to be used by more than 2 workers at a time;

(b) designed and installed in accordance with an engineer's plan in compliance with CSA Standard Z259.13 Manufactured Horizontal Lifeline Systems and CSA Standard Z259.16 Design of Active Fall-Protection Systems;

(3) a rigid continuous anchorage system designed and installed in accordance with an engineer's plan in compliance with CSA Standard Z259.16 Design of Active Fall- Protection Systems.

An anchorage system:

(1) must be designed so that the D-ring of the suspension point of a worker's safety harness cannot be moved horizontally by more than 3 m or an angle of 22° ;

(2) cannot be used by more than 1 person at a time, except in the case of a continuous anchorage system, such as a horizontal life line, or a rigid anchorage system, such as a rail; and

(3) must be designed so that properly attached personal protective equipment cannot be detached involuntarily.

The structure on which the anchorage system is installed must be able to withstand the effort exerted by the anchorage system in addition to the other efforts that it must ordinarily withstand.

An anchorage system with the characteristics described in subparagraph b of subparagraph 1 or 2 of the first paragraph, or in subparagraph 3 of that paragraph, must, before it is first brought into service, be inspected and tested by an engineer or a qualified person acting under the supervision of an engineer, to ensure that the system is in compliance with the design and installation plans.

O.C. 606-2014, s. 8; O.C. 63-2025, s. 10.

2.10.16. Travel restraint system: A travel restraint system must include:

(1) a safety harness complying with section 2.10.12 or a safety belt complying with section 2.10.14;

(2) a fall arrest connecting device without a shock absorber whose length does not allow the wearer to get any closer than 0.9 m from the open side and complying with section 2.10.12;

(3) an anchorage system complying with section 2.10.15, except as concerns the minimum resistance of the single anchorage, which can be 8 kN;

If the conditions provided for in the first paragraph are not complied with, the travel restraint system must be designed and installed in accordance with the plans and specifications of an engineer and comply with CSA Standard Z259.16 Design of Active Fall-Protection Systems.

Where the resistance of the anchorage system provided for in the plans and specifications of the engineer or in subparagraph 3 of the first paragraph does not comply with section 2.10.15, a mark is required on the anchorage system indicating that the use must be limited to travel restraint.

The travel restraint system cannot be used on surfaces with an incline greater than 15° (3/12).

O.C. 63-2025, s. 11.

§2.11. Electricity

2.11.1. An electrical appliance, electric tool or conductor shall be used only for the purposes for which it was designed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.1; O.C. 483-2021, s. 3.

2.11.2. An electrical appliance or electric tool shall be bonded or have double insulation.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.2; O.C. 483-2021, s. 3.

2.11.3. Electrical extension cords shall

(1) have a bonding conductor;

(2) be designed for outdoor use;

(3) be of the Hard Usage type for a circuit 300 volts or less, or of the Extra-Hard Usage type for a circuit 600 volts or less; and

(4) have a capacity at least equal to the value of the circuit overcurrent device.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.3; O.C. 483-2021, s. 3.

2.11.4. Where an extension cord is suspended, the suspension height shall allow free passage.

In addition, supports for suspending the extension cord shall not be conducting or sharp.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.4; O.C. 483-2021, s. 3.

2.11.5. Where an extension cord passes on a floor, it shall be protected to avoid any damage or reduce risks of tripping.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.5; O.C. 483-2021, s. 3.

2.11.6. An extension cord not in use shall be disconnected and stored.

In addition, an extension cord that has a broken, defective or repaired element shall not be used and shall be removed from the construction site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.11.6; O.C. 483-2021, s. 3.

2.11.7. Except where an energy control method provided for in subdivision 2.20 is applied, the components of an electrical circuit of more than 30 volts shall be protected to avoid any contact with a live element.

O.C. 483-2021, s. 3.

2.11.8. Subject to the provisions relating to alarm systems and fire pumps or any other provision to the contrary provided for in the Construction Code (chapter B-1.1, r. 2), the switch of a service box, a feeder or a branch circuit shall not be locked when it is in the energized position.

O.C. 483-2021, s. 3.

2.11.9. A 15A or 20A circuit at 125 volts supplying an appliance or a cord tool shall be protected by a Class A ground fault circuit interrupter.

O.C. 483-2021, s. 3.

2.11.10. A temporary electrical installation shall not be interconnected to the circuit of a permanent electrical installation, unless an appropriate warning is posted at all interconnection points or other locations that constitute a danger.

O.C. 483-2021, s. 3.

2.11.11. The distribution panel of a connection of a temporary outdoor electrical installation shall be weatherproof.

The ground in front and on each side of the panel shall be leveled, drained and free of obstructions to a distance of at least 1 metre.

O.C. 483-2021, s. 3.

§2.12. Precautions during construction or demolition

2.12.1. Any framework shall be calculated, constructed, placed, supported, wind-braced or guyed to resist any load which could be imposed during construction or demolition.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.12.1.

2.12.2. It is forbidden to leave without protection a wall, a chimney or a structure which could collapse during construction or demolition.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.12.2.

2.12.3.-2.12.5. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 391-2011, s. 1.

§2.13. Shoring parts

2.13.1. Telescopic steel jacks:

(1) On request, the employer shall provide the inspector with a copy of the drawings and informations concerning the ultimate capacity of the jacks, a table of the recommended capacities and the minimum length of penetration of the telescoping parts. He shall also be required to supply details of the joints of threaded parts used, the minimum lengths of thread engagement, the recommended methods of usage by the manufacturer and any other information that might affect safety.

(2) The calculation of jack capacity is based on the results obtained from tests carried out by an independent expert.

(3) The flanges at the jack extremities shall have a net bearing surface corresponding to the capacity of the jacks and in all instances shall have a minimum bearing surface of at least 100 cm² and be at least 6 mm thick and made of material currently used.

(4) The length of the penetration of the top part of the jack into the lower part of the jack shall be at least equal to the sixth of the jack length.

(5) The penetration of the threaded portion in the jack shall be equal to 1.5 times the diameter of the tube.

(6) The manufacturer's name or trade mark, the type of jack and the date of manufacture shall be legibly stamped on all jacks and detachable parts.

(7) The employer shall make sure that the defective parts are replaced by parts of the same quality as the original parts.

(8) The employer shall ensure that tests are made to check the capacity of the welding done on the jacks.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.13.1.

2.13.2. Timber shores:

(1) Rectangular shores shall be at least $89 \text{ mm} \times 89 \text{ mm}$ on each side.

(2) Circular shores shall have a minimum diameter of 100 mm.

(3) When shores are spliced end to end, the joint shall be reinforced so as to have the same strength as the shore itself. The joint shall be covered with a piece of the same width as the shore and at least 750 mm in length on 2 adjacent sides of the shore. It is not necessary for the shore parts to be placed end to end when tested assembly clamps are used.

(4) Each part of the spliced shore shall be braced independently in a horizontal manner and, in addition, braced diagonally.
(5) When wedges are used to facilitate the vertical adjustment and removal of shores, they shall be placed on each side of the shore. The vertical space filled by the wedges shall not be greater than the thickness of one wedge.

(6) The timber used for shores shall fulfil the following conditions:

(a) it shall be in good condition, shall be derived from species having long non-breaking fibres and shall be free from any defect that might impair its strength;

(b) it shall be of a quality equal to No. 1 spruce;

(c) if it is in the form of logs, it shall be debarked;

(d) it shall not be covered with paint or any opaque coating.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.13.2; O.C. 329-94, s. 16.

2.13.3. Scaffold shoring: The resistance of the materials making up the scaffold shorings and the support surface of the parts of these scaffolds shall be in accordance with Subdivisions 6.4 and 6.5.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.13.3.

2.13.3.1. All the extensible parts of a scaffold shoring, including adjustment screws, shall comply with one of the following requirements:

(a) bear a mark indicating that, beyond that mark, they are extended beyond the manufacturer's requirements;

(b) be equipped with a locking device that prevents their extension beyond the manufacturer's requirements.

A table of the allowable loads and corresponding extension heights shall be available on the construction site premises.

O.C. 1413-98, s. 6.

§2.14. Damaged buildings

2.14.1. Where a building or structure has been damaged in such a way that it might collapse, totally or partially:

(*a*) it shall be progressively propped or shored up so as to ensure the safety of persons; or

(b) any other equivalent safety measure shall be taken until the building or structure is demolished, dismantled or repaired.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.14.1; O.C. 329-94, s. 17.

§2.15. Hoisting apparatus

2.15.0.1. Definitions:

In this subdivision,

"experienced person" means a person who has acquired knowledge of things through practice and experience;

"qualified person" means a person who has acquired knowledge of things through recognized instruction that has been attested by a diploma;

"competent person" means a qualified and experienced person who has the required skills to properly judge a thing or perform a duty.

O.C. 1393-2024, s. 3.

2.15.1. General conditions:

(1) A hoisting apparatus and its accessories must

(a) be of safe construction and offer the required strength;

(b) be maintained in good working order so that using them does not endanger the safety of workers;

(c) be inspected, repaired and adjusted by a competent person before their first use after being purchased, leased or rented, or loaned;

(d) be inspected periodically according to the manufacturer's instructions;

(e) be subjected by the operator, each day they are used, to a visual inspection and an operational test according to the manufacturer's instructions;

(*f*) be easily and safely accessible, particularly by means of a ladder or steps with handrails;

(g) be equipped with hoisting brakes or restraining devices so designed and installed as to stop a load at least 1.5 times its rated load, unless the hoisting apparatus is specifically covered by a standard referenced in this Code, in which case that standard applies;

(*h*) provide the same degree of safety following any repair or changing of parts as it did when new;

(i) be used in accordance with the manufacturer's instructions to the extent that they do not contradict this Code.

(2) It is prohibited to use a hoisting apparatus or its accessories

(*a* if atmospheric conditions such as a storm could make it dangerous to use them;

(b) while they are undergoing repairs or maintenance;

(c) as an anchor point to protect a person situated outside of the equipment from falling from a height, subject to subsection 10 of section 2.15.12 for a personnel platform lift; and

(d) where the wind speed exceeds the limit specified by the manufacturer.

Despite the wind speed specified by the manufacturer, the operator of the hoisting apparatus must, while using it, take into account factors that could affect the stability of the equipment such as environmental conditions and the wind surface area of the items being handled.

An anemometer must be used to measure wind speed on the construction site at the height of the working level of the hoisting apparatus.

(3) It is prohibited to

(a) use a material hoisting apparatus to hoist persons, unless such use is provided for by the equipment's manufacturer or complies with section 3.10.7;

(b) modify a hoisting apparatus unless it is confirmed by an attestation signed and sealed by an engineer that the apparatus provides the same degree of safety after the modification as it did when it was new; and

(c) refuel a hoisting apparatus while any of its engines is running.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.1; O.C. 1393-2024, s. 4.

2.15.2. Posting of the rated load: The rated load shall be posted in a conspicuous place on a hoisting apparatus.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.2.

2.15.3. The employer shall ensure that the hoisting apparatus is not:

(a) loaded beyond the rated load; and

(b) submitted to sudden movements.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.3.

2.15.4. Boom: The boom of a hoisting apparatus not covered by CSA Standard Z150-1974 Safety Code for Mobile Cranes and its supplement No. 1-1977 or CSA Standard Z248-1975 Code for Tower Cranes shall be installed and built according to the specifications approved by an engineer.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.4.

2.15.5. Load-rating plate: A load-rating plate showing the nominal capacity of the tower crane, mobile crane or other similar apparatus shall:

(a) be so placed and lighted as to be easily read by the operator;

(b) provide information which complies with that provided by the manufacturer; and

(c) furnish all the necessary information for the safe operation of this equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.5.

2.15.6. Handling of loads:

(1) Before ordering the hoisting of a load, the signalman shall ensure that all cables, chains, slings or other moorings are properly attached to the load and that the hoisting does not present any hazard.

(2) The hoisting of loads shall be done vertically.

(3) When oblique hoisting is absolutely necessary, precautions dictated by the circumstances shall be taken, and this operation shall be performed in the presence of a representative of the employer.

(4) If the safety of any person is threatened by the uncontrolled movement or the swinging of a raised load, one or more guide ropes shall be used.

(5) (Subsection revoked).

(6) The crane operator shall not allow a worker to stand on a load, a hook or a sling suspended to a hoisting apparatus.

(7) Hooks used for hoisting loads and hooks attached to slings shall be equipped with a safety catch.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.6; O.C. 53-90, s. 4; O.C. 329-94, s. 18; O.C. 35-2001, s. 8.

2.15.7.1. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.1; O.C. 1393-2024, s. 5.

2.15.7.2. A mobile crane shall conform to CSA Standard Z150-1974 Safety Code for Mobile Cranes and its supplement No. 1-1977, with the exception of section 4.3.2.5.

A mobile crane log book shall be kept up-to-date in accordance with that standard and shall comply with Schedule 9.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.2; O.C. 1413-98, s. 7; O.C. 35-2001, s. 9.

2.15.7.2.1. Two-blocking situation protection device:

(1) Any mobile crane with a telescopic boom shall be equipped with a device designed to warn the operator that a two-blocking situation is about to occur, by means of warning lights and buzzer or an apparatus that stops the equipment. The protection device must be of the fail-safe design.

(2) A cable mobile crane must be equipped with a protective device complying with the device described in paragraph 1. Depending on the date on which the crane was manufactured, the device must be installed not later than:

Date on which the crane was manufactured	Final date on which the protection device must be installed		
As of the date of coming into force of this Regulation	At the time it was manufactured		
From 1 January 1995 to the day preceding the date of coming into force of this Regulation	On the date of coming into force of this Regulation		
From 1 January 1990 to 31 December 1994	31 December 2001		
From 1 January 1985 to 31 December 1989	31 December 2002		
From 1 January 1980 to 31 December 1984	31 December 2003		
From 1 January 1975 to 31 December 1979	31 December 2004		
From 1 January 1970 to 31 December 1974	31 December 2005		

Before 1 January 1970

31 December 2006.

(3) This section does not apply to a cable mobile crane equipped with a shovel, a dragging bucket, a clamshell bucket or a pile hammer and that is used for purposes other than the hoisting of loads and to a cable mobile crane that is used to carry out the following foundation work, including all handling work required to that end: pile driving, installation of drilled or excavated caissons, installation of sheet-piling, setting up of shoring devices, below-grade work, drilling of tie rods or anchors and ground improvement techniques such as dynamic compacting and vibroflotation.

O.C. 53-90, s. 5; O.C. 35-2001, s. 10.

2.15.7.2.2. Overload protection device:

(1) Any mobile crane used for hoisting loads must be equipped with one of the following overload protection devices: a rated capacity limiter, a load moment system or a load indicating device. Depending on the date on which the crane was manufactured, the load indicating device must be installed not later than:

Date on which the crane was manufactured	Final date on which the load indicating device must be installed
From 1 January 1980 to 31 December 2004	31 December 2005
From 1 January 1975 to 31 December 1979	31 December 2010.
(2) Any mobile crane m	anufactured on or after 1 Janu-

(2) Any mobile crane manufactured on or after 1 January 2005 must be equipped with a load indicating device or a load moment system. The devices must be of the fail-safe design.

(3) Load indicating devices and load moment systems may be equipped with a function allowing them to be bypassed.

(4) Load indicating devices must comply with Standard SAEJ376 APR85 Load Indicating Devices in Lifting Crane Service.

(5) Rated capacity limiters and load moment systems must comply with Standard SAEJ159 APR94 Load Moment System.

(6) This section does not apply to a cable mobile crane equipped with a shovel, a dragging bucket, a clamshell bucket or a pile hammer and that is used for purposes other than the hoisting of loads and to a cable mobile crane that is used to carry out the following foundation work, including all handling work required to that end: pile driving, installation of drilled or excavated caissons, installation of sheet-piling, setting up of shoring devices, below-grade work, drilling of tie rods or anchors and ground improvement techniques such as dynamic compacting and vibroflotation.

(7) In this section,

"load indicating device" means a device that measures and displays to the crane operator's sight the weight of the load hoisted by the crane; (dispositif indicateur de charge)

"load moment system" means a device that gathers information relating to the load hoisted by the crane and that transmits signals warning the operator that the crane is nearing its limiting conditions; (contrôleur d'état de charge)

"rated capacity limiter" means a system that prevents a crane from hoisting and moving a load higher than its limiting conditions and that consists of a load moment system and a device acting on the movements of the crane so that they do not exceed the load rating chart of the crane. (dispositif limitateur de charge)

O.C. 35-2001, s. 10.

2.15.7.3. A mobile crane with a luffing boom transformed and used for purposes other than the hoisting of loads such as a shovel, a dragging bucket, a clamshell bucket or a pile hammer shall be equipped with:

(a) bumpers or boom stops; and

(b) a high boom angle switch.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.3.

2.15.7.4. A tower crane shall conform to CSA Standard Z248-1975 Code for Tower Cranes.

A tower crane log book shall be kept up-to-date in accordance with that standard and shall comply with Schedule 10.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.4; O.C. 1413-98, s. 8.

2.15.7.5. A general purpose top running electric overhead travelling crane, except a single girder overhead travelling crane, must conform to CSA Standard B-167-1964 General Purpose Electric Overhead Travelling Cranes.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.5; O.C. 1959-86, s. 13.

2.15.7.6. A lift truck shall conform to CSA Standard B335.1-1977 Low Lift and High Lift Trucks.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.7.6.

2.15.7.7. (Revoked).

O.C. 1413-98, s. 9; O.C. 1393-2024, s. 6.

2.15.8. (1) The rated capacity of a lifting jack shall be indicated in a legible and indelible manner.

(2) Any lifting jack shall be equipped with a positive stop to prevent overstop or with a stop indicator in order to determine the limit of travel.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.8; O.C. 329-94, s. 19.

2.15.9. Hoisting apparatus shall be assembled, maintained and dismantled under the supervision of experienced workers and in accordance with the manufacturer's recom-mendations.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.9.

2.15.10. Minimum age: No work performed by means of a motorized hoisting apparatus may be performed by a worker under 18 years of age.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.15.10; O.C. 329-94, s. 20.

2.15.11. Material hoist:

(1) A material hoist manufactured before 1987 must comply with CSA Standard Z256-72, Safety Code for Material Hoists, including all updates.

(2) A material hoist manufactured in or after 1987 must comply with CAN/CSA Standard Z256, Safety Code for Material Hoists, applicable at the time it was manufactured.

O.C. 1393-2024, s. 7.

2.15.12. Personnel hoisting apparatus:

(1) A personnel hoisting apparatus must be equipped with "deadman" type control levers, an emergency stop button within reach of the workers being transported and a device that prevents the cabin from falling in case of a defect in the electric or hydraulic supply.

(2) A personnel hoisting apparatus must be driven and operated according to the manufacturer's instructions.

(3) A personnel hoisting apparatus must be used solely to move persons, tools and the materials necessary to carry out their work, without exceeding its rated load and in accordance with the manufacturer's specifications.

(4) The work platform of the personnel hoisting apparatus must be enclosed by a guardrail.

(5) It is prohibited for workers situated on the work platform of a personnel hoisting apparatus to use a guardrail, planking, ladder or any other object on or inside the platform to extend their reach or the height to which they are able to reach.

(6) When using an apparatus with multidirectional elevation whose work platform can be horizontally extended from the carrier chassis, a worker must wear a safety harness secured by means of a fall arrest connecting device to an anchorage system provided for by the hoisting apparatus's manufacturer or, failing that, to an anchor, in accordance with subsection 1 of the first paragraph of section 2.10.15.

(7) The operator of a personnel hoisting apparatus that is moving must

(a) limit the travelling speed based on conditions such as the type of ground, visibility, the grade of the terrain, the presence of people and any other factor that could result in collision, injury or overturning;

(b) remain at a safe distance from obstacles, downhill grades, drop off or holes, ramps or any other danger;

(c) make sure to have a good view of the ground and the path to be travelled;

(d) make sure that every person in the concerned work area is informed of the movement of the equipment and there is no one in its path; and

(e) visually inspect the forward range of movement of the platform.

During the visual inspection carried out in accordance with subparagraph *e* of paragraph 7, if the operator notes that nearby overhead structures or obstacles present a risk to persons situated on the platform of being wedged in or crushed, the operator must take the necessary measures to eliminate that risk. Where it is impossible for persons to remain standing on the platform, for example because the equipment is moving under a doorway, the operator must operate the equipment from the ground.

(8) A register of inspections and repairs must be kept by the owner of the personnel hoisting apparatus.

(9) The manufacturer's operating manual for the personnel hoisting apparatus must be stored in the apparatus in the weather-proof compartment.

(10) It is prohibited to use a personnel hoisting apparatus, other than a job-site elevator or a mast-climbing work platform, to transfer persons from one level to another in order to access a work site situated outside of that work site, except where the following conditions are met:

(*a*) after analyzing the risks related to access to the work site, a competent person has determined that access to the work site is impossible by means of a ladder, a stairway, scaffolding, an elevator or a mast-climbing work platform;

(b) an engineer has confirmed in writing that a personnel hoisting apparatus can be used safely for that purpose; and

(c) the hoisting apparatus is used in accordance with a work procedure signed by an engineer that takes into account the manufacturer's recommendations as well as CSA B354.7, Mobile Elevating Work Platforms - Safety Principles, Inspection, Maintenance and Operation. The procedure must be specific to the work site.

Despite the foregoing, a personnel hoisting apparatus may be used to rescue persons as part of a rescue plan.

(11) In the absence of any manufacturer's specifications, a personnel hoisting apparatus must not be used where wind speed exceeds 45 km/h.

(12) A personnel hoisting apparatus must be equipped with an audible warning device that is activated whenever the apparatus is engaged in motorized movement on the ground.

O.C. 1393-2024, s. 7.

2.15.13. Vehicle-mounted aerial device:

(1) A vehicle-mounted aerial device must meet one of the following requirements:

(a) it must be designed and manufactured in accordance with CSA Standard C225, Vehicle-Mounted Aerial Devices, applicable at the time it was manufactured; or (b) it must be designed and manufactured in accordance with ANSI/SAIA Standard A92.2, Vehicle-Mounted Elevating and Rotating Aerial Devices, applicable at the time it was manufactured.

(2) It is prohibited to use a vehicle-mounted aerial device for purposes other than those for which it was specifically designed.

O.C. 1393-2024, s. 7.

2.15.14. Job-site elevator: Every job-site elevator must be designed and manufactured in accordance with CSA Standard Z185, Safety Code for Personnel Hoists.

O.C. 1393-2024, s. 7.

2.15.15. Digger derrick:

(1) A digger derrick manufactured after 1 January 1987 must be designed and manufactured in accordance with ANSI/ASSP Standard A10.31, Safety Requirements, Definitions and Specifications for Digger Derrick, applicable at the time it was manufactured.

A "digger derrick" means an apparatus equipped with a hydraulic boom, mounted on a carrier vehicle and designed specifically to drill holes and install posts in the ground and, by means of an aerial device, install the material to be supported by the posts.

(2) It is prohibited to use a digger derrick to hoist loads other than those for which it was specifically designed. $\overline{0.C. 1393-2024. s. 7.}$

2.15.16. Mobile elevating work platform:

(1) A mobile elevating work platform must be manufactured in accordance with CSA Standard B354.6, Mobile Elevating Work Platforms - Design, Calculations, Safety Requirements, and Test Methods, applicable at the time it was manufactured.

(2) A mobile elevating work platform must undergo a structural inspection in accordance with CSA Standard B354.7, Mobile Elevating Work Platforms - Safety Principles, Inspection, Maintenance and Operation, to ensure that the integrity of its critical components and its stability remain as they were when it was new:

(a) 10 years after the date of manufacture and every 5 years thereafter;

(b) after any suspected, potential or real damage sustained in an incident and liable to affect its structural integrity and its stability; and

(c) after a change of owner.

O.C. 1393-2024, s. 7.

2.15.17. Mobile elevating work platform operator training: A mobile elevating work platform may be used, at the construction site, only by an operator who is adequately trained and familiarized with the type and group of equipment, as defined by CSA Standard B354.6, Mobile Elevating Work Platforms - Design, Calculations, Safety Requirements, and Test Methods. A mobile elevating work platform operator is adequately trained if

(1) the operator has received an initial training, for each type of equipment, whose content is equivalent to CSA Standard B354.8, Mobile Elevating Work Platforms - Operator (Driver) Training. In addition,

(a) the training must comprise a theoretical portion, a practical portion and an evaluation for each type and group of equipment covered by the training. The training must also address work methods for traveling safely under structures to prevent any person situated on the platform from being wedged in or crushed;

(b) the practical portion must include, at a minimum, one hour at the controls of the equipment for each worker;

(c) the evaluation must comprise a theoretical portion and a practical portion at the controls of the equipment which must show that the worker has acquired the necessary skills to safely operate the equipment;

(d) the training must be provided by

i. an accredited instructor in accordance with the Regulation respecting the accreditation and ethics of training bodies, training instructors and training services (chapter D-8.3, r. 0.1); or

ii. an instructor qualified by a training body recognized by the Commission; and

(e) an attestation of training indicating the type and group of equipment covered by the training must be issued to the worker by the training body or the instructor that provided the training;

(2) the operator receives, every five years following the initial training, refresher training including, at a minimum, a practical examination;

(3) the operator is familiarized, at the construction site by a qualified or experienced person, with the following elements:

(a) the location of the manufacturer's operating manuals;

(b) the manufacturer's specific warnings and instructions;

(c) the functions of specific controls;

(d) the function of each specific safety device; and

(e) the specific operating features.

O.C. 1393-2024, s. 7.

2.15.18. Mast-climbing transport platform:

A mast-climbing transport platform must be designed and manufactured in accordance with CSA Standard B354.12, Design, Calculations, Safety Requirements, and Test Methods for Mast Climbing Transport Platforms. In addition,

(1) the mast-climbing transport platform must

(*a*) be covered with a roof in accordance with section 4.4.3.2 of that standard; and

(b) be enclosed by a fixed guardrail at least 1.06 m high consisting of full-length wire mesh that can block a 25 mm ball;

(2) the hoistway doors must

(a) be at least 2 m high;

(b) be equipped with a side protector at least 0.6 m wide on both sides of the door; and

(c) be equipped with a mechanical interlock system that prevents the platform from moving when the hoistway door is open;

(3) where the floor of the loading dock is less than 2 m from the ground, the hoistway door providing access to the platform at that level may

(a) be at least 1.06 m high consisting of full-length wire

mesh that can block a 25 mm ball;

(b) be equipped with a side protector at least 0.6 m wide on both sides of the door consisting of full-length wire mesh that can block a 25 mm ball; and

(c) be kept closed by means of a latch;

(4) The loading dock must be enclosed by a guardrail that complies with section 3.8.3 of this Code;

(5) A mast-climbing transport platform may not be used at a height of more than 55 m measured from the ground;

(6) A fence at least 1.8 m high must enclose the work area around the mast-climbing transport platform installations.

O.C. 1393-2024, s. 7.

2.15.19. Mast-climbing transport platform maintenance and use: A mast-climbing transport platform must be used and maintained in accordance with CSA Standard B354.13, Safe Use and Best Practices for Mast Climbing Transport Platforms (MCTPs).

O.C. 1393-2024, s. 7.

2.15.20. Mast-climbing transport platform operator training: A mast-climbing transport platform may be used only by an operator trained and familiarized with the type of equipment used on the construction site, in accordance with CSA Standard B354.14, Training for Mast Climbing Transport Platforms (MCTPs).

O.C. 1393-2024, s. 7.

§2.16. Piles, sheet-piling

2.16.1. Handling: Piles and sheet-piling shall at all times be firmly held up with a device other than the hand of the worker, during their installation and removal.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.16.1.

2.16.2. Handling area: No worker who is not assigned to the handling of piles and sheet-piling shall be in that area.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.16.2; O.C. 329-94, s. 21.

2.16.3. Pile hammer: When not in use, any pile hammer shall not be kept in a raised position.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.16.3.

2.16.4. Pile heads: Any pile head, before being driven in, shall be:

- (a) cut at right angle; and
- (b) free from debris, bark or chips.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.16.4.

§2.17 (Revoked).

2.17.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.17.1; O.C. 329-94, s. 23.

§2.18. Transportation of workers

2.18.1. Any motor vehicle used for the transportation of workers must conform to any provision of the Highway Safety Code (chapter C-24.2) in the same way as any vehicle used on a public highway.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.1; O.C. 1959-86, s. 14.

2.18.2. Vehicle driver: The employer shall make sure that the driver of the vehicle used for the transportation of workers:

(a) has the permit required by the Highway Safety Code (chapter C-24.2);

(b) informs the person concerned of all defects of his vehicle; and

(c) does not transport anyone if the reported defects have not been repaired, or if he judges that his vehicle does not offer all the safety guarantees.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.2; O.C. 1959-86, s. 15.

2.18.3. The employer shall ensure that any transported worker:

(a) is seated on seats or benches at his disposal during transportation, except in the case of motor vehicles specially equipped for the transportation of persons standing up; and

(b) boards or gets out of the vehicle only when the latter is completely stopped, and uses the boarding devices that are at his disposal.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.3.

2.18.4. Vehicle arrangement:

(1) Any motor vehicle primarily or regularly used for the transportation of workers shall:

(a) be provided with handholds, steps and other accessories so placed and arranged that the workers can safely get in or out of the vehicle;

(b) be provided with a device by which passengers can communicate with the driver, when the driver's compartment is separated from the passengers' compartment;

(c) never transport a number of passengers that may jeopardize the safe driving of the vehicle or the safety of passengers; and

(d) have the floor and the interior space reserved for passengers, in good condition and free from nails, screws, splinters, and other objects likely to be injurious to passengers.

(2) Trucks and buses primarily or regularly used for the transportation of workers shall be:

(a) constructed and accommodated for that purpose;

(b) provided with seats or benches firmly secured in place;

(c) covered with a roof and protected on the sides with solid walls having a minimum height of 1.2 m;

(d) provided with 2 Bengal flares or 2 flashing lights. In case of a break-down on the road or within 3 m of the road, one of these devices shall be placed 30 m in front of the vehicle and the other one 30 m behind;

(e) provided with a portable fire extinguisher in compliance with section 3.4.4 of this Code. The driver shall have the necessary knowledge concerning the type of fire extinguisher with which the trucks and buses are equipped and how it is operated;

(f) provided with a first-aid kit in the driver's compartment.

(3) Trucks primarily or regularly used for the transport of the workers shall be provided with seats or benches offering the following characteristics: (a) a minimum width of 450 mm for each passenger;

(b) so placed as to allow for a longitudinal aisle at least 600 mm when the seats are facing each other and 300 mm wide in other cases;

(c) have a height between 380 mm and 480 mm above the floor;

(d) have a minimum depth of 300 mm; and

(e) provided with backrests extending to a height of at least 500 mm above the level of the seats or benches if these are not placed near the truck-rack or walls.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.4; O.C. 329-94, s. 24.

2.18.5. Special provisions:

(1) Flat-bed trucks and pickups, occasionally used for the transportation of workers can be put into service, if the following conditions are observed:

(a) space allowing, workers shall ride in the driver's compartment and there shall not be more than 3 persons for a single cab;

(b) flat-bed trucks shall be protected on their sides with racks or walls having a minimum height of 900 mm;

(c) workers shall sit on benches placed near the sides of the platform and firmly fitted to the floor, and having a height between 380 mm and 480 mm;

(d) tailgates shall be kept closed while travelling; and

(e) the truck shall be equipped with a tarpaulin to protect workers against bad weather.

(2) Any truck with a closed body, used for the transportation of workers, shall have:

(a) the floor of the workers' compartment airtight so as to prevent intoxication by exhaust gases;

(b) the tailpipe extended beyond the end of the body;

(c) openings specially arranged to allow natural ventilation and lighting of the vehicle for daytime driving; and

(d) artificial lighting inside the workers' compartment for night time driving.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.5.

2.18.6. Prohibited transportation: Transportation of persons is forbidden in trailers and semi-trailers.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.6.

2.18.7. Simultaneous transportation of workers and materials:

(1) Vehicles used for the transportation of workers shall not carry dangerous pesticides and flammable substances, unless these substances are carried in receivers designed for this purpose and outside the compartments occupied by the drivers or passengers. (2) All sharp tools carried in the driver's or the passengers' compartment shall be:

(a) placed in boxes or covered containers; or

(b) protected by scabbards covering the sharp edge and secured to the body of the vehicle outside of the aisles.

(3) It is permitted to carry simultaneously in the same compartment, the workers, and:

(a) small materials, as long as a stowing device prevents any movement of materials likely to injure the passengers;

(b) bulk material, if a safe device prevents this material from filling the place reserved for passengers.

(4) No worker shall ride on a vehicle load while the vehicle is moving.

R.R.Q., 1981, c. S-2.1, r. 6, s. 2.18.7.

§2.19. Prohibited work method and technique

2.19.1. It is prohibited to use a liquified gas to freeze a water pipe unless the gas is physiologically inert and nonflammable.

O.C. 995-91, s. 6; O.C. 1413-98, s. 10.

2.19.2. When freezing a water pipe in a trench or in an obstructed space by using a liquified, physiologically inert and nonflammable gas, the following measures shall be taken:

(a) the trench or the obstructed space shall be ventilated by an exhaust ventilation system so that the concentration of oxygen in the air is at all times equal to or greater than 19.5% at the place occupied by the worker to carry out his work;

(b) a device to measure the concentration of oxygen in the air shall be available on the work site at all times.

For the purposes of this section, an obstructed space is a space marked off by a wall, a partition wall, a ceiling or any other physical obstacle where the circulation of air with the outside is insufficient to maintain a percentage of oxygen in the obstructed space equal to or greater than 19.5% in air volume.

O.C. 1413-98, s. 10.

2.19.3. A worker freezing a water pipe shall wear mittens or gloves as a protection against chilblain.

O.C. 1413-98, s. 10.

§2.20. Lockout and other energy control methods

2.20.1. In this subdivision,

"danger zone" means any zone situated inside or around a machine and which poses a risk for the health, safety or physical well-being of workers;

"energy control method" means a method designed to maintain a machine out of working order in such a way that the working order cannot be altered without the voluntary action of every person having access to the danger zone;

"individually keyed" means a special layout of the components of a lock making it possible to open it with a single key;

"lockout" means an energy control method designed to install an individually keyed lock on an energy isolating device or on any other device allowing for the control of energy such as a lockout box.

O.C. 1186-2015, s. 1.

2.20.2. Before undertaking any work in the danger zone of a machine, such as erecting, installing, adjusting, inspecting, unjamming, setting up, decommissioning, maintaining, dismantling, cleaning, servicing, refurbishing, repairing, altering or unlocking, lockout, or, failing that, any other method that ensures equivalent safety must be applied in accordance with this subdivision.

This subdivision does not apply

(1) where work is carried out in the danger zone of a machine that has a specific control mode as defined in section 2.20.13;

(2) where a machine is unplugged within the reach and under the exclusive control of the person who uses it, where the machine has a single energy source and where there remains no residual energy after the machine is unplugged.

O.C. 1186-2015, s. 1.

2.20.3. Lockout must be carried out by every person having access to the danger zone of a machine.

O.C. 1186-2015, s. 1.

2.20.4. Where the principal contractor intends to apply an energy control method other than lockout, the principal contractor must first ensure the equivalent safety of that method by analyzing the following:

(1) the machine features;

(2) identification of the health and safety risks when using the machine;

(3) the estimate of the frequency and seriousness of the potential employment injuries for each risk identified;

(4) the description of prevention measures that apply for each risk identified, the estimate of the level of risk reduction thus obtained and the assessment of residual risks.

The results of the analysis must be recorded in a written document.

The method referred to in the first paragraph must be developed from the elements mentioned in subparagraphs 1 to 4 of the first paragraph.

O.C. 1186-2015, s. 1.

2.20.5. The principal contractor must, for every machine situated on the construction site, ensure that one or more procedures describing the energy control method are developed and applied.

The procedures must be easily accessible on the sites where work is carried out in written form intelligible to every person having access to the danger zone of a machine.

Where the duration of a construction site lasts more than 1 year, the procedures must be reviewed periodically so as to ensure that the energy control method remains efficient and safe.

O.C. 1186-2015, s. 1.

2.20.6. A procedure describing the energy control method must include the following:

(1) identification of the machine;

(2) identification of the person responsible for the energy control method;

(3) identification and location of every control device and of every energy source of the machine;

(4) identification and location of every cutoff point of every energy source of the machine;

(5) the type and quantity of material required for applying the method;

(6) the steps required to control the energy;

(7) where applicable, the measures designed to ensure the continuity of application of the energy control method during a staff rotation, in particular the transfer of required material;

(8) where applicable, the applicable characteristics, such as the release of residual or stored energy, the required personal protective equipment or any other complementary protection measure.

O.C. 1186-2015, s. 1.

2.20.7. Where lockout is the method applied, the steps required to control energy for the purposes of paragraph 6 of section 2.20.6 must include:

(1) deactivation and complete shutdown of the machine;

(2) elimination or, if that is impossible, control of any residual or stored energy source;

(3) lockout of the machine's energy source cutoff points;

(4) verification of lockout by using one or more techniques making it possible to reach the highest level of efficiency;

 $\underbrace{(5) \text{ safely unlocking and re-operating the machine.}}_{\text{O.C. 1186-2015, s. 1.}}$

2.20.8. Before applying an energy control method, the principal contractor must ensure that the persons having access to the danger zone of the machine are trained and informed on the health and safety risks related to work carried out on the machine and on the prevention measures specific to the energy control method applied.

O.C. 1186-2015, s. 1.

2.20.9. An employer or a self-employed worker must obtain written authorization from the principal contractor before undertaking work in the danger zone of a machine. The principal contractor must make sure that the employer or self-employed worker will apply an energy control method that complies with this subdivision.

O.C. 1186-2015, s. 1.

2.20.10. Where one or more employers or selfemployed workers carry out work in the danger zone of a machine, it is the principal contractor's responsibility to coordinate the measures to be taken to ensure the application of the energy control method, in particular by determining their respective roles and their means of communication.

O.C. 1186-2015, s. 1.

2.20.11. The principal contractor must provide lockout material including individually keyed locks, except if an employer or self-employed worker is responsible therefor pursuant to section 2.20.10.

The name of the person who installs an individually keyed lock must be clearly indicated on the individually keyed lock. Despite the foregoing, the principal contractor may provide persons having access to the danger zone of a machine with individually keyed locks with no name indication, if the principal contractor keeps a record thereof.

The record contains at least the following information:

(1) identification of each individually keyed lock;

(2) the name and telephone number of each person to whom a lock is given;

(3) where applicable, the name and telephone number of the employer of each worker to whom a lock is given;

(4) the date and time at which the lock is given;

(5) the date and time at which the lock is returned.

O.C. 1186-2015, s. 1.

2.20.12. Where a lock is forgotten or a key is lost, the principal contractor may, with the agreement of the person who carried out lockout, authorize the lock to be cut after ensuring that it does not involve any danger for the health, safety and physical well-being of that person.

Where the agreement of the person who carried out lockout is not obtained, the principal contractor must, before authorizing the lock to be cut, inspect the danger zone of the machine accompanied by a representative of the certified association of which the person is a member, if he or she is available on the work site or, failing that, by a worker present on the work site designated by the principal contractor.

Every instance of a lock being cut must be entered in a written document kept by the principal contractor for at least one year following the day on which the applicable energy control method is altered.

O.C. 1186-2015, s. 1.

2.20.13. Where, for setting, maintenance, inspection or other work on a machine, a guard has to be displaced or removed or a protective device has to be neutralized, and where it is necessary for the purpose of those operations for the machinery or part of it to be put into operation, worker safety must be ensured using a specific control mode which

(1) disables all other control modes;

(2) allows operation of elements presenting a risk for worker health and safety only by continuous actuation of an enabling device, a two-hand control device or a holdto-run control device;

(3) allows operation of the elements presenting a risk for worker health and safety only in reduced risk conditions for instance, at reduced speed, under reduced power or force or in a step-by-step mode, for example, with a limited movement device;

(4) prevents voluntary or involuntary action on the machine's sensors from triggering a function presenting a risk for worker health and safety.

O.C. 1186-2015, s. 1; O.C. 1112-2023, s. 8.

2.20.14. This subdivision applies, with the necessary modifications, to any work on an electrical installation.

O.C. 1186-2015, s. 1.

§2.21. Noise

2.21.1. This subdivision sets noise exposure limits, means for evaluating the daily noise exposure level and peak sound pressure level in an establishment, and the standards applicable.

The provisions of this subdivision are intended to eliminate or reduce noise at source or, at a minimum, to reduce workers' exposure to noise.

They also specify the reasonable means that an employer must implement to eliminate or reduce noise at source, comply with the noise exposure limits, and reduce workers' exposure to noise. They specify the workplace situations in which the wearing of hearing protectors is necessary.

For the purposes of this subdivision, "workplace situation" means a trade or a representative function of a worker or group of workers that includes all the tasks and activities of the worker or group of workers, and takes into account the workplace.

O.C. 781-2021, s. 2.

2.21.2. The noise exposure limits are as follows:

(1) daily noise exposure level: 85 dBA, as defined using one of the following formulas:

(a) under ISO Standard 9612:2009, Acoustics – Determination of occupational noise exposure – Engineering method:

 $L_{\text{EX,8h}} = L_{p,\text{A,eqTe}} + 10 \, \log[T_{\text{e}}/T_{0}] \, \text{dB},$

where $T_e =$ effective duration of the working day, in hours;

 T_0 = reference duration, 8 h;

(b) under CSA Standard Z107.56-13, 2014, Measurement of noise exposure:

 $L_{ex.8h} = L_{ea.t} + 10 \log (T_w/8),$

where $T_w =$ effective duration of the working day, in hours.

(2) peak sound pressure level $(L_{p,Cpeak})$: 140 dBC, using the following formula, provided for in ISO Standard 9612:2009, Acoustics – Determination of occupational noise exposure – Engineering method:

 $L_{p,\text{Cpeak}} = 10 \, \log[p^2_{\text{Cpeak}}/p^2_0] \, \text{dB},$

where the reference value, p_0 , is 20 µPa.

O.C. 781-2021, s. 2.

2.21.3. When purchasing or replacing a tool, vehicle, piece of machinery, machine or piece of equipment, the employer must implement reasonable means so that the one that produces the least noise is acquired.

The reasonable means referred to in the first paragraph must not compromise any other element of worker health or safety.

O.C. 781-2021, s. 2.

2.21.4. When planning and performing work, the employer must identify workplace situations that may exceed the noise exposure limits and implement reasonable means to eliminate or reduce noise, in particular by taking into consideration one or several of the following means:

(1) eliminate or reduce noise at source;

(2) limit the propagation of noise by enclosing a machine or piece of equipment;

(3) reduce a worker's exposure, in particular by isolating a workstation;

(4) maintaining a machine or piece of equipment and keeping it in proper working order.

When it is not possible to respect the exposure limits, the employer must implement all the reasonable means identified, even if they do not allow the noise to be reduced sufficiently to respect the exposure limits.

O.C. 781-2021, s. 2.

2.21.5. The employer must reduce the workers' daily exposure to noise, in accordance with section 2.21.6, or provide them with hearing protectors in accordance with the rules established in section 2.21.10,

(1) during the time needed to implement a reasonable measure;

(2) during the time needed to repair or maintain a machine or piece of equipment;

(3) when it is not possible to respect the exposure limits; or

(4) in the situations referred to in section 2.21.7. O.C. 781-2021, s. 2.

2.21.6. To determine the reduction in the workers' daily exposure to noise, the employer must

(1) ensure, if a worker is affected by a single workplace situation comprising a single task or activity that may exceed the exposure limits during the working day, that the worker is not exposed to the equivalent continuous sound pressure level (dBA) specified in the following table for longer than the time indicated:

Equivalent continuous sound pressure level (dBA)	Maximum permitted daily duration	
82	16	
83	12	
85	8	Hours
88	4	
91	2	
94	1	
97	30	Minutes
100	15	
103	7	
106	4	
109	2	
112	1	
115	28	Seconds
118	14	
121	7	
124	3	
127	1	
130-140	< 1	

(2) determine, if a worker is affected by a workplace situation comprising more than one task or activity that may exceed the exposure limits during the working day, a reduction in the daily exposure to noise using the calculator published by the Commission on its website. The $L_{ex,8h}$ or $L_{Ex,8h}$ daily exposure level calculated in this way must be consistent with the daily noise exposure limit.

This section shall not have the effect of permitting a work period that is longer than the period authorized by a law or regulation, a collective agreement, an order in council or a contract of employment.

O.C. 781-2021, s. 2.

2.21.7. The wearing of hearing protectors is mandatory when

(1) the noise level to which a worker is exposed exceeds the exposure limits according to an evaluation based on a measurement made in accordance with section 2.21.8 or using a Type I or Type II integrating sound level meter or a Type II dosimeter;

(2) it is not possible to converse with another person in a normal speaking voice, in other words without raising one's voice or shouting, at a distance of approximately one metre or one arm's-length from the other person, or if impulse noise is present. The evaluation of noise using an integrating sound level meter or dosimeter must be performed by a person with the necessary knowledge acting in compliance with best practices. The person must be available throughout the working day. In addition, the integrating sound level meter or dosimeter must be correctly calibrated on site, before and after the measurement is made, in accordance with the manufacturer's instructions for the instrument used.

O.C. 781-2021, s. 2.

2.21.8. The measurement of the daily noise exposure level and peak sound pressure level must be made in view of the recommendations made in ISO Standard 9612:2009, Acoustics – Determination of occupational noise exposure – Engineering method, – or in CSA Standard Z107.56-13, 2014, Measurement of noise exposure.

In addition, the integrating sound level meter or dosimeter used for the measurement must be a model recommended in one of the two standards.

O.C. 781-2021, s. 2.

2.21.9. The measurement of the daily noise exposure level and peak sound pressure level must be made by

(1) a professional or technician with training in occupational hygiene or specialized training in acoustics; or

(2) a person who masters best practices in the field of noise measurement.

This section shall not have the effect of preventing the employer from designating a person to assist the person referred to in the first paragraph, provided the latter person retains entire responsibility for the measurements made pursuant to section 2.21.8.

O.C. 781-2021, s. 2.

2.21.10. The employer must provide hearing protectors that meet the performance and selection requirements in the following clauses of CSA Standard Z94.2-2014, Hearing Protection Devices – Performance, Selection, Care, and Use:

- (1) 3 to 8.2.1;
- (2) 8.2.4 to 9.1;
- (3) 9.3.4;
- (4) 9.4;
- (5) 9.5.3 to 9.6.1;
- (6) 9.6.3 to 9.7.1;
- (7) 9.8.3;
- (8) 9.9 to 10.3.5;
- (9) 11.2.3 to 11.2.5;

- (10) 12 to 12.2.6.3;
- (11) tables 1 to 6;
- (12) annexes A, B and D.

For the purposes of Clause 9.6.4.3 of the standard, the result of a measurement performed in accordance with section 2.21.8 may be used as a measurement of a worker's noise exposure, namely the $L_{ex,8h}$ or $L_{EX,8h}$ equivalent.

That measurement is not mandatory where the employer selects hearing protectors according to the single-number ratings method provided for in the standard.

The employer may also provide hearing protectors that meet

(1) the performance requirements set out in the following clauses of Hearing protectors - general requirements or Safety requirements and testing, as the case may be:

(a) 1 to 6 and annexes A and ZA of Part 1: Ear-muffs, NF EN 352-1;

(b) 1 to 6 and annexes A and ZA of Part 2: Earplugs, NF EN 352-2;

(c) 1 to 6 and annexes A and ZA of Part 3: Ear-muffs attached to an industrial safety helmet, NF EN 352-3;

(d) 1 to 7 and annexes A, B and ZA of Part 4: Leveldependent ear-muffs, NF EN 352-4;

(e) 1 to 7 and annexes A, B and ZA of Part 5: Active noise-reduction ear-muffs, NF EN 352-5;

(f) 1 to 7 and annexes A, B and ZA of Part 6: Ear-muffs with electrical audio input, NF EN 352-6;

(g) 1 to 7 and annexes A, B and ZA of Part 7: Leveldependent earplugs, NF EN 352-7; and

(2) the selection requirements set out in the following clauses of the NF Standard EN 458:2016, Hearing protectors - Recommendations for selection, use, care and maintenance - Guidance document:

- (*a*) 3 to 4;
- *(b)* 6 to 6.2.1;
- (c) 6.2.3 to 6.5;
- (d) 6.8 to 6.9.2;
- (e) annexes A to E.

For the purposes of Clause 6.2.3.2 and Annex B of NF Standard EN 458:2016, the result of a measurement made in accordance with section 2.21.8 may be used as a measurement of the peak sound pressure level.

A hearing protector meets the requirements of this section if it conforms to the most recent or second most recent version of a standard named in the section and has not exceeded the manufacturer's expiry date, if any.

O.C. 781-2021, s. 2.

2.21.11. The hearing protectors provided for a worker must attenuate noise in such a way that the worker is not exposed to levels that exceed those established in section 2.21.2.

O.C. 781-2021, s. 2.

2.21.12. The employer must provide workers with theoretical and practical training on hearing protectors that addresses

(1) their use and the elements to be taken into consideration in selecting hearing protectors in response to different workplace situations;

(2) the adjustment of hearing protectors;

(3) the inspection of hearing protectors;

(4) the maintenance of hearing protectors;

(5) the risks associated with noise and the importance of wearing protectors during any exposure to noise;

(6) the methods used to evaluate noise levels pursuant to section 2.21.7.

O.C. 781-2021, s. 2.

2.21.13. The employer must post or disseminate in another way any report of a measurement made pursuant to section 2.21.8 not later than 15 days after the report becomes available.

The report must be readily accessible to workers in a visible place until the closure of the worksite or for a minimum period of 3 months, whichever date occurs first. $\overline{O.C.781-2021, s. 2}$.

2.21.14. The employer must include and update in the prevention program, or if none in a register, the following entries and documents:

(1) the workplace situations where limits may be exceeded, as identified when the work was planned;

(2) the reasonable means implemented;

(3) the reports on the measurements made pursuant to section 2.21.8, if applicable.

The employer must keep the information specified in the first paragraph for a period of 10 years, and must keep the other information until the closure of the worksite. The employer must also make it available to the Commission, to workers and their representatives, to the safety representative and to the health and safety committee.

O.C. 781-2021, s. 2.

DIVISION III

CONSTRUCTION SITES

§3.1. Access to construction site

3.1.1. Access to the construction site shall be at all times limited to personnel authorized by the site head.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.3.1; O.C. 329-94, s. 29.

§3.2. Housekeeping of site

3.2.1. Any construction site, including all means of access or egress, shall be kept in a tidy condition and no hazard shall be allowed to develop from the storage of material or equipment, the accumulation of waste or the condition of any material or piece of equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.1; O.C. 329-94, s. 29.

3.2.2. Waste material:

(1) Waste material shall never be thrown from one level to another.

(2) Waste material shall be cleared away progressively:

(a) by means of appropriate containers;

(b) by an inclined or vertical chute emptying into a truck box, a container or an enclosure off limits to workers and strictly reserved for motorized equipment. The box, container or enclosure shall be large enough to hold all waste material emptied into it.

The chute shall:

i. be enclosed if inclined more than 45°;

ii. be closed at its entrance when not in use; and

iii. have a stop-block at its entrance to stop wheel-barrows; or

(c) by means of a hoisting apparatus, if large pieces or objects are involved.

(3) (Subsection revoked).

(4) Waste material shall be transported or so disposed as to create no inconvenience.

(5) Lumber, brick, blocks, stone, steel and other salvaged materials shall be stacked in an orderly fashion and in accordance with the rules of the trade.

(6) (Subsection revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.2; O.C. 53-90, s. 6; O.C. 329-94, s. 25 and 29.

3.2.3. Protruding nails and other protruding parts:

(1) Nails protruding from a piece of wood or any scrap shall be pulled out or nailed down unless such material is piled or placed in a container to be transported outside the site. (2) During dismantling, protruding nails from a reusable piece of wood shall be removed immediately.

(3) The tie rods for concrete formwork and other protruding parts shall be cut away as soon as possible after the deforming.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.3; O.C. 329-94, s. 26 and 29.

3.2.4. Maintenance and arrangement of the work site: Traffic lanes, passages and any work site in general shall:

(a) be kept free of any obstacle and well-cleared;

(b) be kept free of all snow, ice, oil or grease;

(c) be sprinkled with sand or any other non-skid product in order to prevent slipping and falling;

(d) not be obstructed with equipment, tools or material;

(e) not be cluttered with waste products from a mechanical tool;

(f) be illuminated by natural or artificial lighting so as to provide the following lighting levels:

i. 50 lux on traffic lanes and passages used by workers;

ii. 250 lux for rough work, such as casing, concrete pouring or carpentry;

iii. 550 lux for finishing work using a machine or tool, such as a circular saw, a nailing gun or a welding machine;

iv. 800 lux for precision mechanical work;

(g) have a minimum width of 600 mm between machines, installations or piles of materials or goods. This width shall be increased according to the danger present or the size of the parts to be handled;

(*h*) (paragraph revoked);

(*i*) have no opening at floor or roof level, unless the opening is surrounded by guard-rails or closed by a load resistant cover for any loads to which it may be subjected, but not less than 2.4 kN/m2. If the cover or guard-rails interfere with the carrying out of the work, the cover or the guard-rails may be removed and replaced, for the duration of the work, by installing a continuous barrier or trestles of a minimum height of 0.7 m, at a distance varying between 0.9 m and 1.2 m from the opening, or a warning line complying with the requirements in section 2.9.4.1.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.4; O.C. 329-94, s. 27 and 29; O.C. 1413-98, s. 11; O.C. 35-2001, s. 11; O.C. 606-2014, s. 9; O.C. 640-2019, s. 2.

3.2.5. Hazard signs: A danger zone shall be marked off by hazard signs:

(a) below an outrigger scaffolding, a suspended scaffolding or a boatswain's chair; and

(b) at any place where a material hoisting apparatus, a personnel hoisting apparatus with an extensible mast or a concrete pump is used.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.5; O.C. 995-91, s. 7; O.C. 329-94, s. 29; O.C. 1413-98, s. 12; O.C. 1393-2024, s. 8.

3.2.6. Drinking water:

(1) The employer shall supply his workers with drinking water by putting water fountains or clean individual cups and a sufficient supply of water at their disposal.

(2) If the water does not come from a municipal aqueduct, it shall be analysed at the beginning of the works and every 3 months thereafter.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.6; O.C. 329-94, s. 29.

3.2.7. Toilets: As of the first day of work, a toilet must be placed at the disposal of workers. If 25 or more workers are to occupy the site at the same time, a flushing toilet must be placed at their disposal, even if all the workers are not yet present on site. If there never will be more than 24 workers on site, a chemical toilet may be at their disposal.

One toilet is provided per group of 30 workers or fewer.

If a flushing toilet is not connected to a waterworks or sewer system in accordance with the Construction Code (chapter B-1.1, r. 2), it must collect the waste in a tank to process the waste chemically and be built in accordance with the standard Sanitation - Nonsewered Waste-Disposal Systems - Minimum requirements, ANSI Z4.3-1995 (R. 2005), published by the American National Standards Institute.

The obligation to place a toilet at the disposal of workers is met if the workers are authorized to use the sanitary facilities of an establishment located at a distance that complies with section 3.2.7.1.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.7; O.C. 329-94, s. 28 and 29; O.C. 428-2015, s. 3.

3.2.7.1. Toilets must be located at a distance not exceeding 150 m (500 ft.) from the work area and must not be more than 4 storeys higher or lower than the work area.

O.C. 428-2015, s. 3.

3.2.7.2. A toilet must be

(1) within easy access;

(2) free from any obstacle or obstruction that could prevent the toilet from being used;

(3) built so that the occupant is sheltered from view and protected from the weather and falling objects;

(4) provided with natural or artificial lighting;

- (5) equipped with a seat and cover;
- (6) provided with toilet paper;
- (7) heated to a minimum of 20°C; and
- (8) ventilated.

The toilet must be maintained in good operating condition and in a clean and sanitary condition and be kept free of vermin, rodents and insects.

Any cracked or damaged toilet seat must be replaced immediately.

O.C. 428-2015, s. 3.

3.2.8. Sinks: The employer shall provide the workers who handle corrosive or toxic materials with sinks or showers that comply with section 3.2.15 and that enable them to wash with clean water, and likewise provide them with paper towels or individual hand towels.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.8; O.C. 329-94, s. 29; O.C. 393-2011, s. 8.

3.2.8.1. Flushing toilet accessories: A sink containing clean and temperate water must be placed at the disposal of workers for each flushing toilet. The sink must be maintained in good operating condition and in a clean and sanitary condition and the following products must be placed at the disposal of workers:

- (a) soap or another cleansing agent;
- (b) a hand dryer, roller towels or paper towels;

(c) where paper towels are used, waste paper baskets for their disposal.

A notice indicating that the water is not suitable as drinking water must be displayed so that it is visible to workers, where applicable.

O.C. 428-2015, s. 4.

3.2.9. Lunch room: On any work site where there are 10 or more workers and where the work lasts more than 7 days, the employer must make a room available to the workers where they may take their meals. The room must

(a) not have any dimension of less than 2.3 m;

(b) have 1.1 m^2 of floor area per person who eats in the room;

(c) be heated to a minimum of 20° C;

(d) be properly ventilated and lighted and prohibited to smoking;

(e) be provided with hooks for clothing;

(f) be furnished with sufficient tables and chairs for the number of workers who may eat there at the same time;

(g) be supplied with covered recipients for waste; and

(h) be maintained in good, clean and sanitary condition.

The room must not serve for the storage of materials, equipment or tools.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.1.9; O.C. 329-94, s. 29; O.C. 428-2015, s. 5.

3.2.10. Changing room: The employer must provide a changing room to workers working in an underground work site or to those who must wear specific clothing, used exclusively for

(1) abrasive blasting;

(2) work liable to produce asbestos dust emissions where moderate-risk work is carried out in accordance with paragraph 2 of section 3.23.2;

(3) work carried out in heat stress that exceeds the continuous work curve in the "Permissible heat exposure limit values" graph in Schedule V to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13); and

(4) (paragraph revoked).

O.C. 393-2011, s. 9; O.C. 1005-2015, s. 1.

3.2.11. Changing room features: The changing room referred to in section 3.2.10 must be in an area separate from the work area, have a minimum level of illumination of 250 lux and be maintained at a minimum temperature of 20 °C, be provided with drinking water, installations to dry work clothes and individual lockers to store clothes. In addition, the storage space of each locker must be at least 0.14 m³ and there must be a clearance of at least 600 mm in front of each row of lockers.

O.C. 393-2011, s. 9.

3.2.12. Double changing room: The employer must provide workers with a changing room for street clothes and another one for work clothes, between which a shower room is set up so that workers may shower before wearing their street clothes,

(1) for work liable to produce asbestos dust emissions where high-risk work is carried out according to paragraph 3 of section 3.23.2; or

(2) where workers are exposed to lead, mercury or beryllium or their compounds, in the form of vapour or dust.

O.C. 393-2011, s. 9.

3.2.13. Layout of the facilities: The facilities referred to in section 3.2.12 must be contiguous to the work area, each changing room and the shower room must be in separate, communicating rooms used exclusively for the purposes referred to in section 3.2.12. The changing rooms must comply with section 3.2.11 and the showers with section 3.2.15.

O.C. 393-2011, s. 9.

3.2.14. Showers: The employer must provide showers to workers working in an underground work site or to those who must wear specific clothing, used exclusively for

(1) abrasive blasting;

(2) work liable to produce asbestos dust emissions where high-risk work is carried out in accordance with paragraph 3 of section 3.23.2;

(3) work where workers are exposed to lead, mercury or beryllium or their compounds, in the form of vapour or dust;

(4) work carried out in heat stress that exceeds the continuous work curve in the "Permissible heat exposure limit values" graph in Schedule V to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13); and

(5) (paragraph revoked).

O.C. 393-2011, s. 9; O.C. 1005-2015, s. 1.

3.2.15. Shower features: The showers referred to in section 3.2.14 must be installed separately for each sex, not less than 1 shower per 10 workers or fewer of each sex, in rooms that are maintained at a minimum temperature of 20 °C and that have a minimum level of illumination of 250 lux. They must be supplied with temperature controlled potable water, be provided with soap, individual towels and hand towels, and be cleaned and disinfected at least once per shift, when they have been used.

O.C. 393-2011, s. 9.

§3.3. Temporary structures

3.3.1. Ramps, platforms, runways, scaffolding, and other temporary structures shall be designed and constructed according to recognized methods so as to avoid any risk of collapse or any other danger of accident.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.2.1; O.C. 329-94, s. 29.

3.3.2. Any temporary structure shall be sufficiently braced in order to withstand all loads that might be applied during construction, restoration or demolition.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.2.2; O.C. 329-94, s. 29.

3.3.3. Any temporary structure constructed to support part of a permanent structure until this structure is self-supporting, shall be designed, constructed, supported and braced to withstand all loads which might be applied.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.2.3; O.C. 329-94, s. 29.

3.3.4. No change which might affect the structure of a building shall be undertaken before ensuring that the constituent elements will not be submitted to stresses higher than those prescribed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.2.4; O.C. 329-94, s. 29.

3.3.5. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.2.5; O.C. 1959-86, s. 16; O.C. 329-94, s. 29; O.C. 119-2008, s. 4.

§3.4. Fire protection

3.4.1. Fire hoses equipped with nozzles, portable fire extinguishers, automatic sprinkler valves, hydrants and all fire fighting equipment shall be:

- (a) certified;
- (b) easily accessible;
- (c) located at suitably marked stations;
- (d) protected from mechanical damage;
- (e) maintained in good operating condition;
- (f) protected from freezing; and
- (g) accessible to firemen at all times.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.1; O.C. 329-94, s. 30.

3.4.2. Where a standpipe system with valves equipped with hoses and nozzles, an automatic water extinguisher system or a combined system is installed in a building, a standpipe system with valves equipped with hoses and nozzles shall be installed progressively within 2 stories of the highest storey under construction.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.2; O.C. 53-90, s. 7.

3.4.3. Portable fire extinguishers: Portable fire extinguishers shall be placed:

(a) in any workshop;

(b) in any storage building for combustible materials or flammable liquids;

(c) in places where welding or cutting operations with a torch are being carried on and for a reasonable time thereafter;

(d) where temporary oil or gas-fired heat generators are used;

(e) during the use of a tar or asphalt kettle;

(f) during the storage or handling of flammable liquids; and

(g) near the exits on each storey having a floor area of 500 m^2 or less in a building where construction works are in progress, and an additional portable fire extinguisher for each additional 500 m^2 of floor area or any fraction thereof.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.3; O.C. 329-94, s. 31.

3.4.4. In addition to the standards prescribed in section 3.4.1, portable fire extinguishers shall:

(1) be of type ABC approved by Underwriters' Laboratories of Canada (ULC);

(2) be loaded;

(3) comply with:

(a) NFPA Standard 10-1975 Portable Fire Extinguishers, where the extinguishers are manufactured before 1 January 1989; or

(b) NFPA Standard 10-1988 Portable Fire Extinguishers, where the extinguishers are manufactured on or after 1 January 1989;

(4) bear the name of the person responsible for their maintenance and indicate the date of the most recent inspection;

(5) be easily accessible.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.4; O.C. 329-94, s. 32.

3.4.5. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.5; O.C. 329-94, s. 33.

3.4.6. If the building is provided with a water supply system of at least 300 litres per minute under a minimum pressure of 80 kPa at the highest point, hoses equipped with nozzles may replace the portable water extinguishers. The minimum hose diameter shall be 19 mm and the nozzle shall be of the spray type with a minimum opening of 10 mm.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.4.6.

§3.5. Ladders and step-ladders

3.5.1. When there are no stairways, ramps, passages, scaffolding, elevators or mast-climbing transport platforms ladders shall be used to gain access to a work area.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.1; O.C. 329-94, s. 34; O.C. 1393-2024, s. 9.

3.5.2. A ladder shall:

(a) be designed, built, maintained and used so as not to endanger the safety of workers;

(b) always be used in such a way that the loads applied do not cause any part of the ladder to be stressed beyond allowed stresses; and

(c) be suitable for the type of work to be performed, with regard to its type as well as to its length and accessories.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.2.

3.5.3. Commercial ladders: The employer shall ensure that any commercially manufactured ladder that he uses complies with CAN3-Z11-M81 Portable Ladders, except to the extent that the standard is modified by this subdivision.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.3; O.C. 329-94, s. 35.

3.5.4. Site-fabricated ladders:

(1) Any site-fabricated ladder shall have:

(a) a maximum length of 4.8 m, measured along the side rails;

(b) a maximum spacing of 400 mm between the side rails, unless the site where the ladder is used precludes this. In such a case, the width of the ladder may be reduced accordingly;

(c) a maximum spacing of 300 mm between the cleats. Such spacing shall be uniform within a flight.

(2) Any wooden ladder must have:

(a) 2 side rails of at least:

i. 38 mm by 89 mm for single ladders; or

ii. 38 mm by 140 mm or 89 mm by 89 mm for double-width ladders;

(b) cleats:

i. of not less than 38 mm by 89 mm; and

ii. resting on filler blocks of not less than 38 mm by 38 mm.

(3) Any double-width ladder shall:

(a) have 3 rails evenly spaced;

(b) be at least 1.5 m in width and not more than 2 m in width;

(c) have cleats in one piece that extend the full width of the ladder;

(d) be made of materials whose size is corresponding to those listed in subsections 1 and 2; and

(e) be solidly fastened in place.

(4) The wood used for a ladder shall fulfil the following conditions:

(a) it shall be in good condition, shall be derived from species having long non-breaking fibres and shall be free from any defect that might impair its strength;

(b) it shall be of a quality equal to No. 1 spruce;

(c) if it is in the form of logs, it shall be debarked;

(d) it shall not be covered with paint or any opaque coating.

(5) Where it is foreseen that a site-fabricated ladder will exceed the permitted maximum length of 4.8 m, the ladder must be designed by an engineer, as attested to by a plan or certificate signed and sealed by the engineer.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.4; O.C. 329-94, s. 36; O.C. 606-2014, s. 10.

3.5.5. The maximum length of an extension ladder having 2 or more sections shall be 15 m, measured along the side rails.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.5; O.C. 329-94, s. 37.

3.5.6. Ladder utilization: Any ladder shall:

(*a*) rest on a firm footing and the top shall be propped on its 2-side rails;

(b) be firmly held in place by one or more persons if it is not permanently fastened and if its length is equal to or superior to 9 m;

(c) be protected against any jolt or sliding which might upset it;

(d) when not permanently fastened, be so inclined, in compliance with Schedule 0.1, that the horizontal distance between the base of the ladder and the vertical plane of its top support is approximately between 1/4 and 1/3 of the length of the ladder between its supports;

(e) if used as a means of access:

i. be securely fastened in place;

ii. extend at least 900 mm above the upper landing; and

iii. have a minimum clear space of 150 mm below any rung;

(f) be so placed that there is adequate free space at the bottom of the ladder;

(g) (paragraph revoked);

(*h*) not be lashed to another ladder end to end;

(i) if installed in a continuous vertical position, except ladders permanently installed with hoop protection:

i. have rest platforms with handrails, at intervals not greater than 6 m; and

ii. be off-centre at each platform to ensure protection for the upper part;

(*j*) not be used near an exposed electrical circuit when it is made of metal or is metal-reinforced;

(k) be of a length sufficient to allow a person to work without having to use the top 2 rungs; and

(l) be lifted and lowered by the user in such a way that he faces it.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.6; O.C. 329-94, s. 38.

3.5.6.1. Notwithstanding section 3.6.1, a ladder may be used in an elevator shaft or hoistway to temporarily serve one of the 2 means of egrees provided for in that section, for a maximum height of 2 storeys.

In addition, the ladder may be used upon the following conditions:
(a) the elevator or the hoist shall not be in the process of being installed;

(b) the opening of the shaft at the level of the first landing shall be completely closed;

(c) the opening of the second landing shall allow only the free passage of persons;

(d) the ladder shall cover the opening of the second landing so as to prevent persons from falling;

(e) the ladder shall be built as work progresses, but without adversely affecting construction work.

O.C. 329-94, s. 39.

3.5.7. Step-ladder: If the worker uses a step-ladder the employer shall make sure:

(a) that it complies with CAN3-Z11-M81 Portable Ladders, except to the extent that the standard is modified by this subdivision;

(b) that it is not used near an exposed electrical circuit when it is made of metal or is metal-reinforced;

(c) that its legs are fully spread and that the retaining device is locked; and

(d) that the top and the rest are not used as a step.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.7; O.C. 329-94, s. 40.

3.5.8. As long as the permanent or temporary stairways are not finished or ready to be used, ladders shall be utilized and kept in good condition.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.8.

3.5.9. Ladders shall:

(a) be set aside until repaired, when one of their parts is broken or defective;

(b) not be repaired by using splice-pieces or binding; and

(c) be stored away and protected against bad weather in a dry place.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.5.9.

§3.6. Stairs

3.6.1. Any building under construction shall be provided with 2 means of egress by means of permanent or temporary stairs which shall:

(a) be built for the entire height from the ground or underground to the working level as soon as the work has attained a height of 2 storeys or 6 m above ground;

(b) be extended as work progresses without adversely affecting construction work;

(c) never be extended at less than 2 storeys from the upper working level; and

(d) be built of steps with a depth more than 150 mm. R.R.Q., 1981, c. S-2.1, r. 6, s. 3.6.1.

3.6.2. Construction:

(1) Stairs and landings shall be designed and constructed to safely support a live load of 4.8 $kN/m^2.$

(2) Stairs shall have:

(a) uniform treads and risers in any one flight;

(b) a vertical distance between landings or floors not exceeding 3.6 m;

(c) an angle of rise not exceeding 50° from the horizontal; and

(d) guardrails in accordance with Subdivision 3.8 and securely fastened and supported on the open sides of the stairs and landings.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.6.2.

3.6.3. Steel frame stairs: In addition to the requirements of section 3.6.2, skeleton steel stairs shall have temporary treads and landings:

(a) of contiguous planking extending the full width and breadth of these elements; and

(b) be securely fastened in place.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.6.3.

3.6.4. Temporary stairs:

(1) Outside of buildings, temporary stairs shall have:

- (a) steps with a width of at least 500 mm;
- (b) a maximum slope of 55°;
- (c) landings with a depth of at least 500 mm;
- (d) a handrail; and
- (e) uniform steps in the same flight.

(2) Inside buildings, temporary stairs shall have a width of at least 750 mm or at least equal to that of the permanent stairs to be installed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.6.4.

§3.7. Temporary ramps, runways and platforms

3.7.1. Temporary ramps, runways and platforms, other than scaffolding platforms, shall:

(a) be designed, constructed and maintained to safely support loads that may be applied to them;

(b) be at least 480 mm wide;

(c) be securely fastened in place;

(d) have braces that tie their vertical and horizontal supports and ensure stability;

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

(e) when they are open-worked and placed more than 1.8 m above the floor or the ground, not include openings or holes such as a sphere of 30 mm can go through;

(f) have a free space of 2 m above and below, unless the danger is indicated;

(g) where they include welded elements, be welded by welders having an "O" or "V" class certificate from the Canadian Welding Bureau or a Class A or B qualification certificate in pressure vessel welding issued by Minister of Employment and Social Solidarity;

(h) bear a plate giving their rated load, maximum weight (including rated load), name of the manufacturer, date of manufacture, reference to plans submitted and in the case of a ramp, a runway or a platform made after 24 January 1987, identification of the welder;

(i) have guard-rails complying with Subdivision 3.8.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.7.1; O.C. 1959-86, s. 17; O.C. 35-2001, s. 12; O.C. 606-2014, s. 11.

3.7.2. Ramp:

(1) Any ramp shall:

(a) have a slope not exceeding 300 mm of vertical rise to each 900 mm of horizontal run; and

(b) have cross cleats if the slope exceeds 300 mm of vertical rise to each 2.4 m of horizontal run. These cleats shall:

i. be spaced at regular intervals not exceeding 450 $\,\rm mm;$ and

ii. measure at least 25 mm \times 50 mm, nominal measurements.

(2) The requirements of subsection 1 do not apply to a ramp which is installed in the stairwell of a building 2 storeys or less, inasmuch as the ramp:

(a) has a slope not exceeding 45° ; and

(b) has cleats measuring 50 mm \times 50 mm, nominal measurements, spaced at regular intervals not exceeding 300 mm.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.7.2.

3.7.3. Platforms and other similar structures hooked to concrete forms shall be designed, constructed and installed with respect to the work to be performed and the hazards involved. These structures shall:

(a) be supported on solid foundations;

- (b) be securely anchored at the rest point; and
- (c) never be overloaded.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.7.3.

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

§3.8. Guardrails

3.8.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.8.1; O.C. 995-91, s. 8; O.C. 1413-98, s. 15; O.C. 35-2001, s. 13.

3.8.2. Resistance:

(1) A guardrail shall be designed to:

(a) resist a concentrated horizontal force of 900 N applied to any point of the top plate; and

(b) resist a concentrated vertical force of 450 N applied to any point of the top plate.

(2) Where there is a concentration of workers, as well as other areas where a guardrail may be submitted to unusual pressures, the guardrail shall be reinforced accordingly.

(3) Where equipment or materials may fall from one work level to another, precautions shall be taken to avoid this, unless there is a guardrail strengthened for this purpose.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.8.2.

3.8.3. Construction:

(1) Any guardrail shall be between 1 m and 1.2 m above the surface on which the worker is working.

(2) A wooden guardrail shall consist of:

(a) top plate not less than 40 mm thick by 90 mm wide, supported on posts of the same dimension spaced at intervals of not more than 1.8 mm and placed so that the 90 mm width of the post is on the axis of the width of the top plate;

(b) an intermediate rail not less than 75 mm wide at midway and securely fastened to the inner side of the posts; and

(c) a toe-board at least 90 mm high and securely fastened to the inner side of the posts.

(3) A guardrail of steel wire ropes shall be maintained rigid by means of a turn-buckle and consist of:

(a) a wire rope at least 10 mm in diameter for the toprail and the intermediate rail;

(b) steel posts spaced at intervals of not more that 3 m; and

(c) a toe-board of at least 90 mm high and securely fastened to the inner side of the posts.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.8.3; O.C. 329-94, s. 42.

3.8.4. Metal guardrails: Metal guardrails shall be designed, constructed, installed and maintained in such a way as to offer a resistance and a safety equal or superior to that required for wooden guardrails.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.8.4.

3.8.5. When wire mesh is required on a guardrail, it shall:

(a) be of gauge No. 16 or larger;

(b) have mesh of 40 mm or less;

(c) fill all the area between the toe-board and the top-rail; and

(d) be installed on the inside of the intermediate rail. R.R.Q., 1981, c. S-2.1, r. 6, s. 3.8.5.

§3.9. Scaffoldings

3.9.1. Generality: Scaffoldings constructed in accordance with this subdivision shall be required in places where workers cannot carry out their work safely from the ground or from a solid construction. However, ladders may be used for work requiring less than one hour.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.1.

3.9.2. Construction: Scaffoldings shall:

(a) be designed, constructed, braced, windbraced and maintained in order to support the loads and stresses to which they are subjected and be wind resistant; and

(b) rest on the ground or on foundations which offer sufficient resistance.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.2.

3.9.3. Materials:

(1) Materials used for scaffolding shall be free from any defects which might impair their resistance.

(2) The lumber used for scaffolding shall be:

(a) in good condition, composed of long non-breaking fibres and free from any defect which might impair its strength;

(b) of a quality equivalent to No. 1 spruce;

(c) barked, if it is rough timber; and

(d) without paint or any opaque coating.

(3) The metal parts of a scaffolding shall:

(a) not be weakened by rust, nor by any corrosive material; and

(b) if tubular elements are used, not be weakened by the effects of cold or heat.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.3; O.C. 53-90, s. 8; O.C. 329-94, s. 43.

3.9.4. Erection and dismantling:

(1) The erection and dismantling of scaffoldings shall be performed under the supervision and control of a qualified person.

(2) All parts of a scaffolding shall be inspected by a qualified person before its erection.

(3) During erecting and dismantling operations, all safety measures shall be taken to prevent objects from falling.

(4) Tools appropriate to the type of scaffolding shall be placed at the disposal of the workers.

(5) Any worker who erects or dismantles scaffolding must be protected against falls in accordance with subdivision 2.9 of Division II.

(6) During the erection or dismantling of scaffolding, any other work shall:

(*a*) be performed on only those sections which conform to this Code; and

(b) not be performed under the sections being used, unless an overhead protection for workers is provided to catch objects which fall from a higher level.

(7) During its erection or dismantling, no scaffolding shall be left in such a state that it may create a hazard.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.4; O.C. 35-2001, s. 14; O.C. 63-2025, s. 13.

3.9.5. Uprights:

(1) Scaffolding uprights shall rest on the ground or on foundations capable of supporting the maximum load without excessive settlement or deformation.

(1.1) When scaffolding uprights rest on uneven ground, a safe and efficient means, such as screw jacks, must be used to ensure the plumbness of the scaffolding.

The use of construction waste material to fill the unevenness in the ground is prohibited.

(2) The inner row of uprights shall be erected as close as possible to the building or structure.

(3) The wooden uprights shall:

(a) be fastened to the ground or foundation in such a way as to avoid any displacement of the base;

(b) be vertical; and

(c) not be cracked at the foot.

(4) Should the uprights be spliced, they shall be fitted together in such a way that the spliced portion be at least as resistant as the portion which is directly beneath it.

(5) When 2 stationary scaffoldings, resting on the ground, meet at the angle of a building or structure, an upright shall be placed at the intersection of the extended exterior longitudinal girders.

(6) The parts making up the uprights shall be linked up in such a way that no torque or stress capable of causing buckling can come about.

(7) The distance between 2 juncture points of a structure likely to buckle shall not be longer than:

(a) 30 times the smallest dimension of the wooden member involved; or

(b) 70 times the diameter of the tubular member involved.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.5; O.C. 119-2008, s. 5.

3.9.6. Stringers:

(1) Two stringers located at the same level can only be joined at right angles with an upright.

(2) Any stringer shall be in a horizontal position.

(3) Any stringer shall be sufficiently long to be sustained by at least 3 uprights.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.6.

3.9.7. Putlogs: Putlogs supporting the platform shall:

(a) be fixed in a greater vertical dimension; and

(b) be sufficiently long so they extend at least 75 mm beyond the braces of the inside and outside rows of uprights.

(c) be attached to the main structure by one end if they have a single row of uprights. They shall be embedded at least 100 mm in the masonry.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.7; O.C. 640-2019, s. 3.

3.9.8. Platforms: The components of the platforms shall be laid in such a way as to prevent any toppling or sliding. In addition, scaffolding platforms shall:

(1) have a uniform surface between two fulcrum points;

(2) have a minimum free width of 470 mm;

(3) if made of lumber, be constructed of planking:

(a) stamped by a body certified by the Canadian Lumber Standards Association as being of quality equivalent to No. 1 spruce;

(b) measuring at least 50 mm \times 250 mm;

(c) long enough to extend at least 150 mm but not more than 300 mm beyond the end supports; if they are fixed end to end, their extremities shall rest on individual fulcrum points;

(d) laid in such a way as the span between 2 fulcrum points does not exceed:

i. 3.0 m if they are tested and stamped in accordance with Division 11 of CAN/CSA Standard S269.2-M87 Access Scaffolding for Construction Purposes;

ii. 2.1 m in any other case;

(e) whose deflection in the middle of the span does not exceed L/80 where L is the distance between 2 fulcrum points;

(4) if made of manufactured products, be:

(a) built in accordance with CAN/CSA Standard S269.2-M87 Access Scaffolding for Construction Purposes;

(b) worked so as to avoid slipping;

(c) maintained to avoid corrosion and deterioration;

(5) before each use, be visually inspected in order to detect any alteration that may affect its solidity;

(6) have a slope of less than $1:5 (11^{\circ} \text{ from the horizontal plane})$; and

(7) be located less than 350 mm from a wall or other platform if there is no guardrail.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.8; O.C. 53-90, s. 9; O.C. 329-94, s. 44; O.C. 35-2001, s. 15; O.C. 119-2008, s. 6; O.C. 820-2023, s. 2.

3.9.9. Scaffolding assemblies: Scaffolding assemblies shall be performed in the following manner:

(a) if nails are used, they shall:

i. be of a length and size appropriate to the thickness of the assembled pieces;

ii. be sufficiently numerous in relation to the weight being supported;

iii. not be subject to a drawing effect;

iv. be completely driven in, except for double-headed nails in which case the first head shall be completely driven in;

v. have their points flattened should they extend beyond the exterior surface of the last piece assembled; and

vi. be pulled out of any dismantled wood;

(b) when nuts are used, wrenches shall be:

i. the appropriate size, to avoid sliding on the head of the nut;

ii. such that while tightening, the elasticity limit of the metal with which the nut is made is not exceeded; and

iii. equipped with a strap to enable the workmen to suspend them from his wrist or belt;

(c) when devices such as clamps or metallic points are utilized, these devices shall:

i. be specially designed for this purpose; and

ii. be so fixed as not to slip under the pressure brought to bear upon them.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.9.

3.9.10. Mooring:

(1). Scaffoldings whose height is greater than 3 times the shortest side dimension of their base shall be firmly

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

anchored to a building or structure by means of anchors, or to the ground by means of guys.

(2). The building or structure to which the scaffolding is anchored shall resist to the load induced by scaffolding and anchors.

The anchors shall

(a) resist to traction and compression side loads applied to the scaffolding. The loads must be equal to 225 N/m of platform length without being less than 1,0000 N;

(b). be installed

i.. in accordance with the recommendations of the scaffolding's manufacturer; or

ii.. on the vertical, at intervals not exceeding 3 times the smallest side dimension of the scaffolding and, on the horizontal, at least every 2 uprights;

(c). be uniformly distributed in a staggered arrangement if possible.

In addition to those requirements, when a safety tarpaulin or net is installed on a scaffolding, the number and type of anchors shall comply with an engineer's plan or with the manufacturer's recommendations or, in the case of scaffoldings less than 18 metres in height, shall comply with those provided for in tables 1 and 2 of Schedule 0.2, depending on whether it is a tarpaulin or net and on the region where the scaffolding is installed.

(3) Scaffolding anchored to the ground by means of guys shall be installed in accordance with the recommendations of the scaffolding's manufacturer or with the plans for the installation of the scaffolding.

(4) This section does not apply to a suspended scaffolding, a boatswain's chair, a ladder jack scaffolding, a suspended scaffolding used by bricklayers or a mast-climbing work platform referred to in sections 3.9.22 to 3.9.25.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.10; O.C. 640-2019, s. 4.

3.9.11. Means of access: Scaffoldings more than 1.5 m from the ground, or from any other fixed support, shall have means of access, safe and clear of any obstruction, with:

(a) landings at every 6 m; and

(b) a stair for scaffoldings 18 m in height or over.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.11.

3.9.12. Inspection: Scaffoldings shall be inspected by a qualified person:

(a) at least once every 3 months;

(b) following any failure of the material;

(c) after any abnormal pressure or any incident which might have adversely affected the structure; and

(d) when re-used after any lengthy work interruption. R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.12.

3.9.13. Hoisting cables for scaffoldings:

(1) Any hoisting cable for scaffoldings shall:

(*a*) be designed for this use and of the appropriate diameter. This diameter not being less than:

i. 19 mm for fibre cables; or

ii. 8 mm for metallic cables;

(b) have a safety factor of 10;

(c) be provided with the appropriate sockets when spliced into eyeholes or links. If cable terminal clamps are used, they shall be of a size appropriate to the diameter of the metallic cable and so placed that the band of the U is on the supporting end of the cable;

(d) be bound at both ends to avoid unravelling of the strands;

(e) be shielded from projections of the building;

(f) be adequately protected when corrosive substances are used nearby;

(g) be kept in a cool and dry place protected against chemical or corrosive vapours;

(h) if used with a hoisting apparatus having a friction drum, be long enough to reach the ground or be prevented from coming out of the hoisting apparatus by twisting the free end around an eyelet and fixing it by means of a cable clamp.

This last method must be used for any work above a body of water or watercourse; and

(*i*) if used with a hoisting apparatus having a winding drum, be fixed to the drum with a fastener having a minimum resistance of 80% of the breaking point of the hoisting cable.

(2) Any fibre cable:

(a) shall not be used in any of the following cases:

i. when the suspension points are more than 30 m above ground;

ii. on winch drums;

iii. near corrosive or chemical substances unless it has been appropriately treated;

(b) shall not drag on rough surfaces;

(c) shall be kept in good condition by:

i. drying it out; and

ii. protecting it from frost; and

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

(d) shall be replaced after 2 years of services or before, if the cable is frayed and if the strands are discoloured or blackened and are beginning to crumble by producing a whitish dust.

(3) Synthetic fibre cable may be used in place of fibre cable if the same standards are fulfilled and if it has an equivalent strength.

(4) Any metallic cable shall:

(a) comply with the rules of the Hoisting and Rigging Safety Manual of the Construction Safety Association of Ontario, translated by the Commission des normes, de l'équité, de la santé et de la sécurité du travail and published by Les Publications du Québec;

(b) be made of a minimum of 6 strands of 19 wires;

(c) have a hemp core or an equivalent flexible core if the cable is wound around the hoisting drum in a single layer, or a steel core in other cases; and

(d) be maintained in good condition by:

i. following all the manufacturer's instructions;

ii. handling it in such a manner as to avoid any kink;

iii. frequently applying an appropriate lubricant which will help keep it flexible and protect it from rust; and

iv. using it only on pulleys or drums which are smooth and without any ragged edges.

(5) Unless a test shows that nowhere along the cable has the breaking point been lowered to less than 90% of the original point, the hoisting cable shall be replaced if:

(a) 4% of the total number of wires in the cable are broken in one strand through the cable pitch, approximately 6 1/2 times the diameter of the cable, when used on a winding drum, or 2% on a friction drum;

(b) the original diameter, measured on an unstretched cable, has been reduced by:

i. 0.8 mm for cables of 8 mm to 15 mm in diameter; or

ii. 1.2 mm for cables of 15 mm to 25 mm in diameter;

(c) the exterior wires present a worn aspect covering more than 50% of their diameter;

(d) corrosion is more than superficial.

(6) Any hoisting cable, when being used on an apparatus, shall not be left lying on the ground, but kept in a receptacle.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.13; O.C. 329-94, s. 45; O.C. 393-2011, s. 10; O.C. 513-2015, s. 4.

3.9.14. Utilization:

(1) The employer shall see that no one works on a scaffolding:

(a) which does not conform to this Code;

(b) during a storm or a high wind period; or

(c) when the platform is covered with ice, snow or sleet, unless the said surface has been covered with an anti-slip substance.

(2) Workmen shall not work on scaffoldings with different levels, unless protection has been provided over those working below in order to stop tools or other objects which may fall from a higher level.

(3) A hoisting apparatus forming part of the installation shall be installed and operated according to the manufacturer's instructions.

(4) In assembling metal scaffolding, it is prohibited to use any piece that has been reformed or straightened in such a way that its strength is diminished.

(5) No vehicle or equipment which can be moved and on which a scaffolding is installed or erected, may be moved when a worker is on the scaffolding.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.14; O.C. 1959-86, s. 18; O.C. 329-94, s. 46.

3.9.15. Outrigger scaffolding: The outrigger beams for an outrigger scaffolding shall:

(a) have their inboard portion from fulcrum point to point of anchorage not less than 1.5 times the length of the outboard portion;

(b) be supported on the solid parts of the construction;

(c) be securely anchored at the fulcrum point and well braced against upsetting;

(d) be securely anchored at the inboard end in order to prevent any vertical or horizontal movement; and

(e) be inspected before each use.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.15.

3.9.16. Suspended scaffolding:

(1) Any suspended scaffolding shall:

(a) be off-limits to persons under age 18;

(b) at the working level, be firmly and appropriately attached to the structure if the guardrail installed on the structure side is to be removed; and

(c) not be attached to another scaffolding or to the structure by a bridge;

(d) be used with a safety harness secured by a fall arrest connecting device to an anchorage system, the whole in accordance with sections 2.10.12 and 2.10.15. However, when the suspended scaffolding is hung from 4 hoisting cables, the anchorage system may be installed on the platform. Where a rope grab fastened to a vertical life line is used, it must be a Class ADP rope grab.

(2) A mechanically operated suspended scaffolding shall conform to CAN3-Z271-M84 Safety Code for Suspended Powered Platforms.

(3) A manually operated suspended scaffolding shall also conform to the standard mentioned in subsection 2, with he necessary modifications.

The winch of the scaffolding shall be fitted with at least 2 independent braking devices, including one automatic brake, that shall serve only for that purpose.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.16; O.C. 807-92, s. 4; O.C. 606-2014, s.12.

3.9.17. Boatswain's chair:

(1) Any boatswain's chair shall:

(a) be prohibited to persons under 18 years of age;

(b) be built of material able to support 115 kg with a safety factor of 4;

(c) be not less than 600 mm wide and 250 mm deep;

(d) be suspended from a hoisting ring by a metallic structure or by means of slings at the 4 corners and crossing beneath the seat; and

(e) be equipped with:

i. retaining devices preventing any swaying;

ii. stirrups so as to avoid numbness of the legs;

iii. (subparagraph revoked);

iv. a back; and

v. reinforcement blocks at each corner which extend at least 230 mm beyond the front of the chair.

(2) The boatswain's chair shall conform to CAN3-Z271-M84 Safety Code for Suspended Powered Platforms, with the necessary modifications.

(3) Where the boatswain's chair is moved by means of a manually operated winch, that winch must be equipped with at least 2 independent braking devices, including an automatic brake, that shall serve only for that purpose.

(4) A worker in a boatswain's chair must wear a safety harness secured by a fall arrest connecting device to an anchorage system, the whole in accordance with sections 2.10.12 and 2.10.15. Where a rope grab fastened to a vertical life line is used, it must be a Class ADP rope grab.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.17; O.C. 807-92, s. 5; O.C. 35-2001, s. 16; O.C. 606-2014, s. 13.

3.9.18. The use of a ladder jack scaffolding shall be prohibited.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.18; O.C. 329-94, s. 47; O.C. 483-2021, s. 4.

3.9.19. Mobile scaffolding: Any mobile scaffolding shall:

(a) be equipped with a braking device which prevents the scaffolding from moving;

(b) have the brakes applied when it is being used;

(c) be rigid and stable; and

(d) not be moved when a workman is on it, unless:

i. he is advised of the move;

ii. the smallest dimension of the base of the scaffolding is at least equal to 1/3 its height.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.19.

3.9.20. Metal scaffolding:

(1) The use of a metal scaffolding shall conform to the instructions of the manufacturer.

(2) The tubes or runners shall be assembled in such a way that when they are taken down, whatever remains of the scaffolding will not constitute a hazard.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.20.

3.9.21. Suspended scaffolding used by bricklayers: Any suspended scaffolding for the use of bricklayers shall:

(a) be attached to an outrigger beam capable of withstanding at least 4 times the maximum utilization load without exceeding the allowable stress of the materials used. This outrigger beam shall, in addition to be conform to section 3.9.15, be solidly fixed in one of the following ways:

i. by an anchorage to the concrete slab on which it rests;

ii. between 2 concrete slabs, by beams or columns braced perpendicular to the vertical and horizontal planes;

iii. to the beams of the metal framework;

(b) in addition to conforming to paragraphs a and c of subsection 1 of section 3.9.16, have a wooden guardrail as specified in Subdivision 3.8;

(c) include hoisting apparatus:

i. in sufficient number and assembled so that the safety factor of the cables is always more than 10;

ii. attached 2 by 2 to the same outrigger beam; and

iii. equipped with a double safety device which continually locks the winch against any reverse motion and with cables which conform to section 3.9.13;

(d) include a platform:

i. in conformity with section 3.9.8;

ii. made of pieces of wood 50 mm \times 250 mm bolted at each end on a putlog;

iii. equipped with steel putlogs whose circular ends are 25 mm in diameter and 50 mm in length and perforated 15 mm at the outside edge for a 10 mm pin; and

(e) assembled so that an end of one span is never more than 200 mm higher than the other.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.9.21; O.C. 1413-98, s. 16.

3.9.22. Mast-climbing work platform: Every mast-climbing work platform shall:

(1) be designed in conformity with the plans of an engineer; a copy of the plans must be available on request;

(2) be equipped with a plate on each hoisting system clearly stating the maximum load allowed on the platform;

(3) be equipped with bearers resting on base plates or mudsills, as shown on the engineer's plans or in the manufacturer's manual;

(4) have a distance between masts in conformity with the engineer's plans;

(5) have the sections of each mast linked together using the means specified in the engineer's plans or in the manufacturer's manual;

(6) have masts whose plumbness complies with the lesser of the following measures: the gap between a plumb line and any point lower than

(a) the engineer's plans or the manufacturer's manual; or

(b) 12 mm for 3 m in height, 19 mm for 6 m in height or 38 mm for the total height of the scaffolding;

(7) be loaded in conformity with the engineer's plans or the manufacturer's manual which must identify the loading zones;

(8) have masts anchored at the height and using an anchoring system in conformity with the engineer's plans or the manufacturer's manual;

(9) be equipped with an anchoring system designed for that purpose in conformity with the engineer's plans or the manufacturer's manual when safety tarpaulins are installed; and

(10) be used by workers trained to

(a) identify and prevent the hazards associated with the erection, dismantling and use of the scaffolding; and

(b) work on the platform safely.

O.C. 119-2008, s. 7.

3.9.23. Pump jack scaffold: In addition to the standards prescribed in section 3.9.22, every pump jack scaffold shall:

(1) be designed to support, in addition to the dead weight, 3 times the maximum load allowed without damage to any of its components;

(2) be used by not more than 2 workers at the same time between 2 masts;

(3) have masts made out of a material other than wood;

(4) be equipped with jacks designed to prevent any accidental slippage of the platform alongside the masts; and

(5) when the work platform passes in front of an intermediate anchoring point, be equipped with a new anchor installed one and a half metres under the platform, before removing the intermediate anchoring.

O.C. 119-2008, s. 7.

3.9.24. Winch scaffolding: In addition to the standards prescribed in section 3.9.22, every winch scaffolding shall:

(1) be designed and manufactured in compliance with ANSI/SIA Standard A 92.9-1993 Mast-Climbing Work Platforms;

(2) when climbing, have bracing affixed under the work platform before the above bracing is removed;

(3) be equipped with an arresting device to stop and support the platform in case of overspeed; the device must be designed to stop the work platform with twice the maximum load allowed by limiting the height of the fall to 300 mm, without any break in the components under stress;

(4) in addition to the instructions of the manufacturer, be inspected in compliance with the following minimum conditions:

(*a*) before each use in compliance with section 7.4.3 of ANSI/SIA Standard A 92.9-1993 Mast-Climbing Work Platforms by a qualified person; and

(b) quarterly by a qualified mechanic;

(5) undergo every year a visual examination of its welds by a welding inspector holding a certificate issued by the Canadian Welding Bureau or a welding supervisor in the employ of a company certified in compliance with the requirements of CSA Standard W47.1 Certification of Companies for Fusion Welding of Steel; and

(6) undergo every 5 years a non-destructive examination of the bearing parts by an organization certified by the Canadian Welding Bureau in compliance with the requirements of CSA Standard W178.1 Certification of Welding Inspection Organizations.

O.C. 119-2008, s. 7.

3.9.25. Motorized scaffolding: In addition to the standards prescribed in section 3.9.22, every motorized scaffolding manufactured after 13 March 2008 shall:

(1) be designed and manufactured in compliance with ISO Standard 16369 - Mast-Climbing Work Platforms, or CSA Standard B354.9, Design, calculations, safety requirements, and test methods for mast climbing work platforms (MCWPs) applicable at the time of the manufacture;

(2) have the sides of the work platform next to a mast protected over a height of at least 2 m to prevent access to the mast when the speed of the work platform exceeds 2.5 m per minute;

(3) be equipped with a device preventing the fall of the work platform in case of malfunction of the hoisting system;

(4) have at least one plate on which the following particulars appear in French:

- (a) the name of the manufacturer;
- (b) the date of manufacture;
- (c) the climbing speed;
- (d) the self-bearing height;
- (e) the electric power supply; and
- (f) the load table;

(5) in addition to the manufacturer's instructions, be inspected in compliance with the terms and conditions of CSA Standard B354.10/B354.11, Safe use and best practices for mast climbing work platforms (MCWPs)/ Training for mast climbing work platforms (MCWPs), and according to the following frequencies:

(a) before each use; and

(b) every 6 months or every 120 hours of use, whichever comes first, by a qualified mechanic;

(6) undergo every year a visual examination of its welds by a person holding for at least 5 years a certificate issued by the Canadian Welding Bureau or a welding supervisor in the employ of a company certified in compliance with the requirements of CSA Standard W47.1 Certification of Companies for Fusion Welding of Steel; and

A document confirming the examination shall be kept. The document shall contain the name and signature of the examiner and the date of the examination.

Where an examination reveals an anomaly or a sign of wear, the examination provided for in subparagraph 7 shall be performed before the motorized scaffolding maybe again used.

(7) undergo, within 10 years after the date of manufacturing, and, then, every 5 years, a non-destructive

examination of the load-bearing parts, specified by the manufacturer, in compliance with the requirements of CSA Standard W178.1 Certification of Welding Inspection Organizations.

The structure shall also be analyzed using ultrasound.

A document confirming the examination and analysis shall be kept. The document shall contain the name and signature of the examiner and the date of the examination.

In addition, a complete instruction manual for every motorized scaffolding, written in French, must be at the disposal of users to ensure the scaffolding is used safely.

O.C. 119-2008, s. 7; O.C. 483-2021, s. 5.

3.9.26. Bracket scaffolding: Every bracket scaffolding must:

(1) be designed in conformity with plans signed and sealed by an engineer; a copy of the plans must be available on request; and

(2) undergo every 5 years a non-destructive examination, other than a visual examination, of its welds by an organization certified by the Canadian Welding Bureau in compliance with the requirements of CSA Standard W178.1 Certification of Welding Inspection Organizations.

O.C. 606-2014, s. 14.

§3.10. Construction equipment

3.10.1. Any self-propelled vehicle or equipment used on a construction site shall:

(a) be maintained so that its use shall not compromise the safety of workers;

(b) if driven by a source of energy other than manual, be inspected by an experienced person before being used for the first time on the site and, afterwards, on a daily basis, whenever it is used;

(c) be easily and safely accessible by means of either a ladder or steps with handrails;

(d) have the motors stopped when refuelling;

(e) not be used if weather conditions could render its use hazardous or during repairs or maintenance;

(f) provide a safety equal to the original safety following any repair or changing of parts;

(g) be used according to the manufacturer's instructions; and

(h) not be modified without an engineer's signed and sealed authorization specifying that such modification provides a safety equivalent to that provided by the apparatus when new.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.1; O.C. 1393-2024, s. 10.

3.10.2. Self-propelled vehicles:

(1) Any self-propelled vehicle shall be:

(a) provided with efficient brakes; and

(b) provided with horns that shall be used when approaching pedestrians, doors, curves and hazardous areas. This requirement does not apply to crawler type dozers or to skidders.

(2) The operators of self-propelled vehicles shall be protected against falling objects or material by means of canopies, protective screens, cabs, or structures.

(3) Self-propelled vehicles equipped with a winch at the back to haul materials shall have a protective screen between the winch and the operator.

(4) Persons other than the operator are prohibited from boarding self-propelled vehicles, unless these vehicles are equipped with seats and accessories for their protection.

(5) (Subsection revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.2; O.C. 1959-86, s. 19; O.C. 807-92, s. 6; O.C. 1393-2024, s. 11.

3.10.2.1. All-terrain vehicles

The use of all-terrain vehicles on a construction site is permitted only under the following conditions:

(a) the vehicle is mounted on at least 4 wheels;

(b) it is equipped with a portable fire extinguisher complying with section 3.4.4 of this Code;

(c) it is equipped with a yellow flag at least 0.05 m^2 in area placed at least 1.5 m off the ground;

(d) the driver of the vehicle has the skill and knowledge required to safety use the vehicle as well as the extinguisher with which it is equipped;

(e) the driver shall wear the following pieces of individual protective equipment:

i. a motorcycle or snowmobile protective helmet conforming to the Protective Helmets Regulation (chapter C-24.2, r. 6);

ii. safety glasses or a face shield that comply with the latest version of CAN/CSA Standard Z94.3 Eye and Face Protectors, within 24 months following its last update, or a visor designed to be attached to a protective helmet;

iii. flexible leather gloves or gloves made of a material that ensures a snug fit at the wrists and a good grip on the vehicle's controls.

It is prohibited to install a winch on an all-terrain vehicle.

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

In this section, "all-terrain vehicle" means a pleasure vehicle designed for driving elsewhere than on public highways and having a net mass not exceeding 450 kg.

O.C. 807-92, s. 7; O.C. 329-94, s. 49; O.C. 393-2011, s. 11.

3.10.3. Heavy equipment:

(1) The heavy equipment referred to in subparagraph a or b must have a rollover protective structure that complies with the standards referred to in those subparagraphs:

(a) Bulldozer, loader and skidder on wheels or caterpillars, grader, scraper and roller: ISO 3471 Earth-Moving Machinery - Roll-Over Protective Structures - Laboratory Tests and Performance Requirements, applicable at the time of manufacture of the equipment;

(b) Agricultural and industrial tractor with engine power greater than 15 kW: SAE J1194 Rollover Protective Structures (ROPS) for Wheeled Agricultural Tractors, applicable at the time of manufacture of the equipment.

(2) The heavy equipment referred to in subparagraph a of subsection 1, with the exception of rollers, must be protected from falling objects by a protective structure that complies with ISO 3449, Earth-Moving Machinery - Falling Object Protective Structures - Laboratory Tests and Performance Requirements, applicable at the time of manufacture of the equipment.

(3) (Subsection revoked).

(4) A rollover protective structure shall bear a plate indicating:

(a) the name of the manufacturer or his code;

(b) its serial number;

(c) a reference to the number of the standard used for its conception; and

(d) the model and the make of the heavy equipment for which the structure has been designed.

(5) The operator of a heavy equipment equipped with a protective structure, except for the grader, shall wear a safety belt when the vehicle is in motion.

(6) The caterpillars of bulldozers working perpendicular to the axis of a slope shall be equipped with metal parts placed so as to prevent sliding in the axis of the slope.

(7) When rollers are used on a steep slope, one shall:

(a) be careful to keep to the newly laid paving;

(b) be careful never to come in contact with bituminous bonding; or

(c) provide for a device capable of stopping the roller in case of danger.

(8) This section does not apply to a bulldozer on caterpillars equipped with a side boom and a counterweight and used for installing gas or oil pipelines.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.3; Decision 83-11-17, s. 6; O.C. 393-2011, s. 12; O.C. 606-2014, s. 15.

3.10.3.1. A loader manufactured after 23 April 1980 shall be designed and installed so that persons entering or leaving the cabin may not be struck or caught by a moving part.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.3.1.

3.10.3.2. Loader used for handling purposes

Subject to section 3.10.3.3, the use of a loader to handle materials, other than earthwork materials, is allowed under the following conditions:

(a) the handling of materials shall be carried out with equipment, excluding the bucket, designed by the manufacturer for the handling of materials; and

(b) the handling of materials shall be carried out in accordance with a method prepared in writing by the employer and indicating that no worker may be located under the load arms, any other part of the loader equipment being used for lifting, or under the load during handling.

O.C. 1413-98, s. 17.

3.10.3.3. Earth-moving machinery used for lifting purposes

The use of a hydraulic shovel, backhoe loader or loader for lifting purposes, whether or not it is modified, is allowed only for water, sewer, culvert or blasting works, if the following conditions are complied with:

(a) lifting must be carried out in accordance with a work method prepared in writing by the employer, available on the work site. The method must comply with the requirements provided for in section 2.15.6 and indicate that no worker may be located under the boom, arm, load arm or bucket of the machine or under the load during lifting; and

(b) the hydraulic shovel, backhoe loader or loader shall also be equipped with a load hook-up device designed to avoid any accidental unhooking. The device shall be designed by the equipment manufacturer or be recommended by the manufacturer.

(c) the requirements provided for in section 2.15.1 must be complied with.

The use of a sling or a mooring rope attached to the bucket teeth to lift a load is prohibited.

O.C. 1413-98, s. 17; O.C. 428-2015, s. 6.

3.10.3.3.1. Hydraulic shovel used for assembly on the ground during electric power line installation

The use of a hydraulic shovel on caterpillars for assembling components on the ground during the installation of an electric power line is allowed if the conditions of section 3.10.3.3 and the following conditions are complied with:

(1) the shovel must be equipped with load lowering control devices on the boom and the arm that comply with the standard Earth-moving machinery - Hydraulic excavator and backhoe loader boom-lowering control device - Requirements and tests, ISO 8643, published by the International Organization for Standardization, with the necessary modifications, and visual or audible overload indicator that complies with the standard Earth-moving machinery - Safety - Part 5: Requirements for hydraulic excavators, NF EN 474-5, published by the Association française de normalisation (AFNOR);

(2) a nominal load table that complies with the standard Earth-moving machinery - Hydraulic excavators - Lift capacity, ISO 10567, published by the International Organization for Standardization, must be installed so that the operator can read it;

(3) hoisting must be executed on a level surface having a bearing capacity sufficient to support the equipment and the load hoisted without significant settlement;

(4) the bucket of the hydraulic shovel must be removed to hoist a load.

O.C. 428-2015, s. 7.

3.10.3.4. Earth-moving machinery used for installing posts

The use of a hydraulic shovel or of a backhoe loader for the installation of posts is allowed if the following conditions are complied with:

(1) the lift cylinders of the boom, arm or bucket of the machine are provided with load lowering control devices that comply with one of the following standards:

(a) for machinery manufactured before 5 May 2011: ISO 8643: 1988 Earth-Moving Machinery - Hydraulic Excavator and Backhoe Loader Boom-Lowering Control Device - Requirements and Tests, with the necessary modifications;

(b) for machinery manufactured on or after 5 May 2011: ISO 8643 Earth-Moving Machinery - Hydraulic Excavator and Backhoe Loader Boom-Lowering Control Device -Requirements and Tests, with the necessary modifications;

(2) the outriggers are provided with control devices installed in accordance with one of the standards referred to in subparagraph a or b of paragraph 1.

O.C. 1413-98, s. 17; O.C. 393-2011, s. 13.

3.10.4. Utilization:

(1) All construction equipment must be used by an experienced person or under their supervision.

(2) No person shall operate a vehicle on a construction site unless:

(a) he holds a licence to use the vehicle on a public highway in accordance with the Highway Safety Code (chapter C-24.2); or

(b) he is the holder of a document attesting to his ability to drive the vehicle off public highways.

(3) Paragraph b of subsection 2 above does not apply to a person who is:

(a) in training; and

(b) accompanied by a person who meets the requirements provided for in subsection 2.

(4) Any manoeuvre shall be carried out so that no workman is exposed to a hazard. No load shall pass over the heads of workers and no worker shall stand under a load or under a part of a hoisting apparatus that could fall on him.

(5) The controls of construction equipment shall not be left unattended when the following parts of the equipment are raised:

(a) the bucket of a power shovel, backhoe or other excavating machine;

(b) the blade of a bulldozer;

(c) the load of a lift truck, crane or any other hoisting machine; or

(d) the hammer of a pile-driving device.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.4; O.C. 1959-86, s. 20; O.C. 35-2001, s. 17; O.C. 1393-2024, s. 12.

3.10.5. Workers acting as signal persons:

(1) (paragraph revoked);

(2) If the operator of a crane, mechanical shovel or any other construction equipment does not have an unrestricted view during any maneuver, except a backup maneuver, one or more workers shall assist the operator:

(a) by watching the movement of the equipment or of the load when it is out of the operator's sight;

(b) subject to section 3.24.2, by communicating with the operator by means of a code of hand signals, or by a telecommunications system when conditions so require or the operator judges it necessary.

(3) When a worker is required to ensure safe maneuvering as prescribed in subsection 2, he must be placed in full view of the operator, with a full view of the intended path of travel of the vehicle and off the intended path of travel.

(4) The operator must obey any stop signal.

(5) When the operator can no longer see the worker, he must stop his vehicle or his maneuver.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.5; O.C. 1959-86, s. 21; O.C. 329-94, s. 51; O.C. 1078-2015, s. 4.

3.10.6. Access to construction equipments:

(1) No unauthorized persons shall be allowed to climb onto construction equipment.

(2) The employer shall see that no worker remains in or on a vehicle if he might be endangered during the loading or unloading of the vehicle or while objects are being handled above the said vehicle.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.6.

3.10.7. Lifting of workers:

(1) *(Revoked);*

(2) The lifting of a worker with an apparatus designed exclusively for the lifting of materials is allowed provided that:

(a) the apparatus is equipped with a platform;

(b) there are not more than 5 workers on the platform;

(c) every worker wears a safety harness secured by a fall arrest connecting device to an anchorage system, the whole in accordance with sections 2.10.12 and 2.10.15;

(d) the total load does not exceed 70% of the rated load of the hoisting apparatus;

(e) the operator of the hoisting apparatus uses the stabilizers when the apparatus is equipped with them;

(f) the hoisting apparatus is not displaced when a worker is on the platform;

(g) there is communication without intermediaries between a worker on the platform and the operator of the hoisting apparatus;

(*h*) the employer submits the plans, the description of the hoisting apparatus and the utilization process to the Commission, except for a mobile crane or a lift truck; and

(*i*) ladders, stepladders, scaffolds or other accessories are not used on the platform.

(3) The platform required in subsection 2 shall:

(a) have a non-skid floor;

(b) be built in such a way as to allow a clearance of 2 m above any point of the floor;

(c) be equipped with a metal guardrail on the 4 sides. The intermediate rail may be replaced by a wire mesh;

(d) have a minimal width of 500 mm;

(e) have a minimum safety factor of 4 for the structural elements;

(f) comply with the plans required in paragraph h of subsection 2 of section 2.4.1;

(g) if it has welded elements, be welded by a welder holding a Class "O" or "V" certificate of the Canadian Welding Bureau or a Class A or B qualification certificate in pressure vessel welding issued by Minister of Employment and Social Solidarity; and

(*h*) have a plate indicating the rated load of the platform, the total weight of the platform (including the rated load), the manufacturer's name, the date of manufacture and a reference to the plans submitted. Identification of the welder shall be mentioned for any platform manufactured after 23 April 1980.

(4) During the hoisting of a worker with a mobile crane:

(a) the crane shall conform to CSA Standard Z150-1974 Safety Code for Mobile Cranes and its supplement No. 1-1977;

(b) the platform shall be suspended or held in such a way that:

i. the slope of the floor does not exceed 1/5 in the worst loading conditions; and

ii. the flexible suspension members of the platform and the supporting hitch or pin have a minimum safety factor of 10;

(c) an additional link shall tie the supporting hitch of the platform to a point located above the hook; and

(d) the mobile crane shall be equipped with an upper limit switch for the hook or with a boom allowing the lifting of the platform at least to 3 m above the highest work level.

(5) During the hoisting of a worker with a lift truck:

(*a*) the lift truck shall conform to CSA Standard B335.1-1977 Low Lift and High Lift Trucks;

(b) the platform shall encircle the forks and be secured to the apron of the lift truck;

(c) the total load shall not exceed 50% of the rated load of the lift truck;

(d) when the platform is equipped with a hoisting control, it shall be possible to stop the lift truck from this control and the latter shall have priority on any other control; and

(e) when the platform is not equipped with a hoisting control, the operator of the lift truck shall remain at the controls during the work.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.7; O.C. 1959-86, s. 22; O.C. 329-94, s. 52; O.C. 35-2001, s. 18; O.C. 606-2014, s. 16; O.C. 1393-2024, s. 14.

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

3.10.8. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.8; O.C. 329-94, s. 53; O.C. 1413-98, s. 18; O.C. 35-2001, s. 19; O.C. 606-2014, s. 17; O.C. 1393-2024, s. 15.

3.10.9. Material hoisting apparatus:

(1) Any material hoisting apparatus used at a construction site must have an outrigger beam capable of supporting 4 times the apparatus' rated load. The beam must comply with section 3.9.15.

(2) The hoisting cable shall always be in good condition and never used as a sling.

(3) Any worker who pulls loads onto a floor at platform level must be protected against falls in accordance with subdivision 2.9 of Division II.

(4) The projection on the ground level of the space used for hoisting operations, the loading area and the hoisting mechanism area shall be encircled by a barrier and forbidden to any worker other than the one doing the loading.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.9; O.C. 329-94, s. 54; O.C. 35-2001, s. 20; O.C. 1393-2024, s. 16; O.C. 63-2025, s. 14.

3.10.9.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.9.1; O.C. 329-94, s. 55; O.C. 1324-2024, s. 17.

3.10.10. Cabin:

(1) If the operator of a self-propelled vehicule runs the risks of being injured by flying objects, or by falling objects or material, a cabin or a protective structure shall be installed.

(2) The cabins shall be:

(a) located and equipped so as to provide the operator with the best possible view;

(b) closed and provided with safety glass windows on all side if used outside; and

(c) adequately heated during cold weather.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.10; O.C. 1393-2024, s. 18.

3.10.11. Lighting: When a self-propelled vehicle is in operation after nightfall or whenever circumstances so require, it shall be provided with:

(a) headlights in front and at least one red light in the rear; and

(b) sufficient lighting installed so as not to hamper the operator but allowing him to carry out his duties and to read easily the indicators on the dashboard.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.11.

3.10.12. Warning devices:

(1) Any self-propelled vehicle shall be equipped with a warning device that shall:

(a) be installed within reach of the operator;

(b) have a sound which is special to that type of equipment and different from any other signal on the site; and

(c) be loud enough to override surrounding noises.

(2) The following must be equipped with an automatic reset backup alarm for the reverse gear:

(a) any self-propelled vehicle used mainly on a site where the operator's view, through the rear window, is obstructed;

(b) any earth-moving machinery as defined in ISO Standard 6165:2012, Earth-moving machinery – Basic types – Identification and terms and definitions;

(c) any truck having a nominal capacity of 2,250 kg or more, as defined in subsection 5 of this section.

(3) The automatic reset backup alarm device referred to in subsection 2 must have the following features:

(a) have a distinct sound and a noise intensity that is superior to the surrounding noise and to the noise of the equipment on which it is installed;

(b) be visible from the rear of the vehicle and face backward;

(c) if the device is electric, it must comply with SAE Standard J994, Alarm – Backup – Electric Laboratory Performance Testing.

(4) In addition to the requirements provided for in subsection 3, the backup alarm device installed on earth-moving machinery must comply with ISO Standard 9533:2010, Earth-moving machinery – Machine-mounted audible travel alarms and forward horns – Test methods and performance criteria.

(5) For the purposes of this section, "nominal capacity" means the gross vehicle weight rating certified by a motor vehicle manufacturer less the net mass of the vehicle.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.12; O.C. 1078-2015, s. 5.

3.10.13. Safety and protective devices:

(1) Moving parts of machines shall be protected by safety devices. They shall be designed, constructed and used so as:

(a) to ensure a positive protection and to prevent any access to the danger zone during their operation;

(b) not to hamper the operator;

(c) to operate automatically or with a minimum of effort;

(d) to allow easy maintenance and repair of the machine;

(e) not to have sharp edges, asperities or burrs and to constitute a source of hazards; and

(f) to be solidly fixed.

(2) If moving parts are not protected due to their location or by suitable guards, it shall be assured that:

(a) the projecting extremities of transmission shafts are protected by sleeves or sockets;

(b) the revolving collars are cylindrical and without projecting parts;

(c) the jet screws, belts, keys, grease cups and other projections in revolving parts are flushed or covered to avoid any contact; and

(d) unused keyways are filled up or enclosed in a guard. R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.13.

3.10.14. Abrasive wheels: Any abrasive wheel:

(a) 50 mm or more in diameter, shall be equipped with a guard compatible with the work being performed and offering the most efficient protection against accident;

(b) shall be tightened between 2 protective discs of a minimum diameter of at least 1/3 its diameter unless mounted on chucks cemented on back metallic plates or firmly fixed on back by any other mean;

(c) shall have its tool holder or its work stand firmly fixed and never more than 3 mm from the abrasive wheel;

(d) shall be inspected before being mounted in order to detect cracks, excessive looseness or other defects;

(e) shall not operate at a speed higher than the maximum speed specified by the manufacturer; and

(f) shall not be used if it is cracked, broken or unbalanced.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.14.

3.10.15. Saws:

(1) Bandwheels on bandsaws and the return portion of the blades shall be placed in guards, and the working side of the saw blade shall be protected between the rollers or guide blocks and the casing of the upper band wheel.

(2) Any stationary circular saw shall be provided with a protective hood covering the exposed parts up to the base of the teeth.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.15; O.C. 483-2021, s. 6.

3.10.16. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.16; O.C. 483-2021, s. 7.

3.10.17. Internal combustion motor:

(1) No internal combustion motor shall be used near sites where there are dusts or flammable or explosive vapours.

(2) Air impurities produced inside any closed structure shall be removed at the emission point so that their concentration is reduced to a level below the limit values specified in Schedule I to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13).

(3) As soon as the ventilation system stops, any internal combustion motor shall be immediately stopped and re-started only when the ventilation system starts to operate again.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.17; O.C. 885-2001, s. 377; O.C. 805-2020, s. 1.

3.10.18. Repairs: Before beginning repairs, all raised parts shall be well blocked.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.18.

3.10.19. Long loads: When a load extends more than 900 mm from the rear of a truck, the end of the load shall be indicated by:

(a) a square red flag during the day, at least 300 mm along its sides; or

(b) a flashing amber light at night.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.10.19.

§3.11. Temporary heating

3.11.1. Any equipment providing a temporary heating supply system shall be safely installed and operated.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.1.

3.11.2. It is prohibited to use a heating system which burns gasoline or naphtha.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.2.

3.11.3. Any oil or fuel or gas heating system with the exception of those where the flame is in direct contact with the air, shall have a vent pipe.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.3.

3.11.4. Any fuel-fired heating system shall be:

(a) so located, protected and used that there is no risk of igniting:

i. tarpaulins or any other similar temporary shelter; or

ii. wood or any other combustible material adjacent thereto;

(b) used in an enclosed area only if there is:

i. sufficient air supply for normal combustion; and

ii. sufficient ventilation;

(c) protected from any damage or overturning;

(d) located so as not to block the means of egress;

(e) connected to a metal chimney, if used to burn a solid fuel, in order to discharge the products of combustion outside the building; and

(f) connected to the liquid fuel tank by means of piping well protected against any damage.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.4.

3.11.5. An electric air-heater must be certified in accordance with CAN/CSA Standard C22.2 No. 46, Electric Air-Heaters, applicable at the time of its manufacture.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.5; O.C. 393-2011, s. 14.

3.11.6. Oil heating equipments shall be installed in accordance with CSA Standard B139 — 1971 Installation Code for Oil Burning Equipment, with the exception, however, of clause 12.3 of that Code.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.6.

3.11.7. Any radiator, device or equipment operated with natural gas and used on a construction site shall:

(*a*) comply with CAN/CSA Standard B.149.1-M91 Natural Gas Installation Code, if it is fired with natural gas, and with CAN/CSA Standard B.149.2-M91 Propane Installation Code, if it is fired with propane gas; and

(b) not discharge gases into the work environment that could cause gas concentrations to exceed the standards prescribed in section 2.10.8.

Subparagraph b of the first paragraph also applies to oil radiators used on construction sites.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.7; O.C. 1413-98, s. 19.

3.11.8. The installation of solid fuel heating equipment, including the mounting, clearances and air supply of such equipment, must comply with CAN/CSA Standard B365 Installation Code for Solid-Fuel-Burning Appliances and Equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.8; O.C. 606-2014, s. 18.

3.11.9. Hot air and air retake ducts shall be constructed with non-flammable material and be sufficiently supported.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.11.9.

§3.12. Steam supply

3.12.1. Supplying of steam shall be done by means of equipments safely installed and put into operation.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.12.1.

3.12.2. Steam supply lines shall be clearly identified.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.12.2.

3.12.3. No one shall disconnect or repair a section of a steam line under pressure before ensuring that all precautions have been taken to prevent any hazard from occurring.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.12.3.

3.12.4. Any steamhose of a pile driving engine shall be held back by a steel cable or chain to avoid any whiplash.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.12.4.

3.12.5. Any steam supply system shall be insulated or protected so that no worker comes into contact with the piping.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.12.5.

§3.13. Supply of gas under pressure

3.13.1. The piping through which compressed air or gas circulates shall be protected against any shock or shall be clearly identified as to the nature of its content.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.13.1; O.C. 1413-98, s. 20.

3.13.2. Before disconnecting or repairing a connection or a section of piping through which gas under pressure circulates, the piping supply shall be cut off and that piping shall be completely depressurized.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.13.2; O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.3. Air piping: Connections for flexible piping sections through which compressed air circulates shall be equipped with one of the following devices:

(a) a chain or steel rope attached to each side of the connection;

(b) a self-locking device; or

(c) a locking device.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.13.3; O.C. 1959-86, s. 23; O.C. 1413-98, s. 20.

3.13.4. It is prohibited to store oxygen cylinders less than 6 m from a flammable gas cylinder or a combustible such as oil or grease, unless the oxygen cylinder is separated from it by a wall made of non-combustible material, of a minimum height of 1.5 m and of a fire resistance of not less than 30 minutes.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.13.4; O.C. 1413-98, s. 20.

3.13.5. Any gas under pressure cylinder shall:

(a) comply with the Act respecting pressure vessels (chapter A-20.01) and its regulations;

(b) be kept away from any heat source liable to increase the temperature of the contents above 55 °C;

(c) be used only for the purposes for which it is intended; and

(d) be handled in such a way as not to damage it.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.13.5; O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.6. A gas under pressure cylinder shall be used only under the following conditions:

(*a*) it shall be placed upright so that the pressure limiting device is in constant and direct contact with the gaseous phase; and

(b) it shall be fastened to a rigid structure or held on a cart designed for the transportation of such cylinder, or on the vehicle it supplies.

O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.7. When a gas under pressure cylinder is not in use,

(a) it shall be held in place upright, with the valves on top; and

(b) the protective cap shall be in place.

O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.8. Gas under pressure cylinders connected in series by a manifold shall be supported and maintained together, forming a unit, with the use of a rack or another installation designed for that purpose and the valves and safety devices shall be protected against shock.

O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.9. It is prohibited to lift gas under pressure cylinders by the collar or the protective cap.

O.C. 1413-98, s. 20; O.C. 805-2020, s. 3.

3.13.10. Any propane gas cylinder that is not connected for use, shall be stored outside and protected from unauthorized handling:

(a) in a place where no vehicles or mobile equipment are used or moved; or

(b) in an area protected by a continuous barricade or trestles of a minimum height of 0.7 m.

O.C. 1413-98, s. 20.

§3.14. Welding and cutting

3.14.1. Welding and cutting operations are prohibited close to combustible materials or in places containing flammable or explosive dusts, gases, or vapours, unless precautions are taken to prevent fire or explosions.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.14.1.

3.14.2. Any electric or gas welding or cutting operation and the installation, handling and maintenance of the equipment used for such operations shall comply with CAN/CSA Standard W117.2-M87 Safety in Welding, Cutting, and Allied Processes, except section 7.8.2.2.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.14.2; O.C. 1413-98, s. 21.

3.14.3. No cutting, welding or other work requiring an open flame shall be performed on a vessel, reservoir, pipe or other container where a flammable or explosive substance may be present, unless:

(a) air samples have been taken to indicate that the work may be done without danger; or

(b) procedures have been taken in accordance with subdivision 9.8 of CAN/CSA Standard W117.2-M87 Safety in Welding, Cutting, and Allied Processes.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.14.3; O.C. 1413-98, s. 22.

3.14.4. Protective screens against radiation shall be installed where electric arc welding or cutting operations are liable to constitute a danger to the health, safety and physical well-being of persons other than the welder.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.14.4; O.C. 1413-98, s. 23.

3.14.5. A welding torch using oxygen and a combustible gas shall be equipped with a gas-check valve and with a flame arrester at the combustible gas supply point and the oxygen supply point located in the torch handle.

O.C. 329-94, s. 61; O.C. 1413-98, s. 23.

3.14.6. The chassis or hood of a welding machine supplied by a public power utility shall be grounded in accordance with CSA Standard C22.10-1992 Canadian Electrical Code, Part I (16th Edition) plus the Québec Amendments (O.C. 1674-91, 91-12-04), and any future provision amending it.

O.C. 1413-98, s. 23.

3.14.7. A portable welding machine supplied by an internal combustion engine shall be grounded if it is provided with auxiliary outlets of 120 V or 240 V and if those outlets are used simultaneously with the welding process.

O.C. 1413-98, s. 23.

3.14.8. It is prohibited to use electrical conductors or a pipeline containing flammable gases or liquids as a welding or cutting current neutral circuit.

O.C. 1413-98, s. 23.

§3.15. Diggings, excavations and trenches

3.15.1. Public services:

(1) Before digging, the employer shall verify whether there is underground piping within the perimeter of the works to be carried out and, where applicable, situate its exact location on the ground.

(2) The piping may be temporarily disconnected, relocated or left in place, on condition that:

(a) the excavation procedure has been approved in advance by the road office or the utility company;

(b) a work method is adopted which prevents damage to conduits; and

(c) temporary supports are constructed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.1; O.C. 807-92, s. 9.

3.15.2. Adjoining structure: Where there is any danger that the proposed digging might affect the stability of the adjoining structures and thus threaten the safety of workers, an adequate method of shoring or a substructure shall be provided, and a copy of the plans and specifications of such works shall be kept on the site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.2; O.C. 807-92, s. 10.

3.15.3. (1) The employer shall ensure that the banks of an excavation or trench are shored solidly with quality materials in accordance with the plans and specifications of an engineer. Shoring is not required in the following cases:

(1) where the trench or excavation is dug out of solid rock or where no workers are required to descend into it;

(2) where there is no risk of the banks of the trench or excavation collapsing and where they slope less than 45° from a point less than 1.2 m from the bottom;

(3) where there is no risk of the banks of the trench or excavation collapsing and where an engineer attests that it is not necessary to shore up the banks, given the slope, nature and stability of the ground. A copy of the engineer's attestation shall be available on the construction site at all times.

Solid rock means rock that cannot be excavated otherwise than by blasting.

(2) The shoring shall extend 300 mm above the excavation, except where the trench dug in on a public road and must be covered over to allow for the passage of traffic during periods when works are not in progress.

When the shoring consists of unjoined stays, the 300 mm extension above the excavation shall be fitted with a plinth 300 mm wide.

(3) The banks shall be shored as the work progresses unless the shoring can be carried out before the digging.

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

(4) During the work the employer shall ensure that the walls are inspected and maintained so as to remove:

(a) rocks or materials liable to become detached therefrom;

(b) overhanging matter.

Where the banks have been subject to environmental or climatic stress, inspections shall be more frequent.

(5) It is prohibited:

(a) to place materials less than 1.2 m from the rim of the banks;

(b) to drive or park vehicles or machines less than 3 m from the rim of the banks, unless reinforced shoring has been used;

(c) to let banks deteriorate.

(6) Shoring shall be removed by an experienced person or under his supervision:

(a) from the bottom upwards; and

(b) only in places to which workers no longer have access.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.3; O.C. 807-92, s. 11.

3.15.4. Means of access and supervision: The employer shall ensure that ladders are placed in trenches at intervals of 15 linear metres or fractions thereof and in such manner as to follow the progress of the works.

Ladders shall rest on the bottom of the trench and extend at least 1 m above ground level.

When workers are in a trench, the employer shall post an experienced person at ground level so that he may detect faults, earth breaking away or any other source of danger.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.4; O.C. 807-92, s. 12.

3.15.5. Barriers, barricades or warning line:

(1) Continuous barriers or barricades of a minimum height of 0.7 m or a warning line as provided for in section 2.9.4.1, must be set up on the edge of any escarpment or digging:

(a) whose depth is more than 3 m; or

(b) which might constitute a hazard to workers or the public in general.

(2) (Subsection revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.5; O.C. 995-91, s. 9; O.C. 606-2014, s.19.

3.15.6. Drainage: Any excavation or trench shall be kept reasonably clear of water.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.6.

3.15.7. Caissons:

(1) No worker shall enter a caisson unless:

(a) he is protected by a steel shoring:

i. that is sufficiently resistant;

ii. covering all the depth above the worker who is working therein; and

iii. extending at least 300 mm above ground level;

(b) he uses a boatswain's chair in accordance with section 3.9.17;

(c) (paragraph revoked);

(d) he is under the constant supervision of another worker for as long as he is in the trench;

(e) he accompanies each load of materials excavated if he cannot take shelter outside the caisson;

(f) any accumulation of water or mud inside the caisson is controlled;

(g) the provisions of Subdivision 3.21 have been complied with.

(2) No worker shall enter the belled area of a caisson unless such section is solidly shored or unless an engineer attests that such shoring is not necessary, given the nature and stability of the ground. A copy of that attestation shall be available on the construction site at all times.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.7; O.C. 807-92, s. 14; O.C. 329-94, s. 62; O.C. 35-2001, s. 21.

3.15.8. Shafts:

(1) The banks of shafts shall be shored in compliance with section 3.15.3.

(2) During shaft-sinking works, workers shall not enter the shaft unless the provisions of Subdivision 3.21 have been complied with.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.8; O.C. 807-92, s. 15.

3.15.9. Boring: A drilling machine shall be equipped with:

(a) pipes supplying water in holes during drilling; or

(b) a mechanized exhaust system for removing the dust created by drilling machines.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.9; O.C. 807-92, s. 16; O.C. 48-2022, s. 3; O.C 820-2023, s. 3.

3.15.10. Minimum age: No worker under 18 years of age is allowed to work in excavations or trenches.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.15.10.
§3.16. Handling and storage of materials

3.16.1. Generalities:

(1) All construction site materials shall be used, displaced or transported onto the site or unloaded from a vehicle or from a pile so as not to compromise the safety of workers and the public.

(2) The load imposed by storing materials on a permanent or temporary structure shall not exceed the permissible load.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.1.

3.16.2. Piling of materials:

(1) Piling of materials shall be done so that piles do not disturb:

(a) the spread of natural or artificial light;

(b) the good operation of machines and other installations;

(c) movement in passages, aisles, stairways, elevators, or near doors; and

(d) the efficient operation of fire prevention equipment.

(2) Materials shall not be piled against walls or partitions of buildings without ensuring that they can resist the lateral pressure.

(3) Materials shall not be piled to a height which might jeopardize the stability of the pile.

(4) The employer shall supply the necessary equipment to reach the height of piles to the person assigned to warehousing.

(5) The distance between the side of an opening in an opening in a floor or roof and the bottom of a pile of materials shall be more than the height of the pile except if a device prevents the falling of materials.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.2.

3.16.3. Building lumber: Reserves of building lumber shall be neatly piled. These piles shall be:

(a) supported at a certain height off the ground;

(b) disposed in horizontal layers, criss-crossed and slightly inclined; and

(c) stabilized by means of transversal supports or wedges, if the piles are more than 1.2 m high.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.6; O.C. 329-94, s. 66.

3.16.4. Masonry materials: Masonry units shall be stacked:

(a) on planks, a platform or other level base;

(b) in layers covering the entire area of the stacks;

(c) in such a way that the height of the vertical face of the stacks does not exceed 1.8 m;

(d) in tiers, when the height of the stockpiling exceeds 1.8 m;

(e) with wooden transversal supports between the layers to prevent the pile from crumbling, if the height of the pile exceeds 1.8 mm;

(f) bound together, if necessary, so as to ensure the stability thereof.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.7; O.C. 329-94, s. 66.

3.16.5. Pipes: Pipes shall be stacked:

(a) on racks or solid shelves;

(b) on wooden blocks provided with tappets at both ends; or

(c) on metal bars of which both ends are bent upwards. R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.8; O.C. 329-94, s. 66.

3.16.6. Bagged materials: Bagged materials shall: (*a*) be stacked by cross-piling the bags to form layers piled not higher than 10 bags unless:

i. the bags are stored in reservoirs or enclosures; and

ii. the sides of the piles are supported by the walls of the reservoirs or enclosures; and

(b) be removed from the piles in such a way that the top of the pile remains horizontal.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.9; O.C. 329-94, s. 66.

3.16.7. Bound objects:

(1) When drums or barrels are stacked:

(a) full, standing upright, the height of the piles shall be limited and 2 planks shall be laid side by side on each row before proceeding on to the next row;

(b) empty, lying on their sides, the piles shall be symmetrical and stable, and all units in the bottom row shall be carefully wedged.

(2) Subsection 1 also applies to large-diameter pipes, to rolls of paper or to any other object having a circular cross-section.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.10; O.C. 329-94, ss. 65 and 66.

3.16.8. Hand lifting and carrying:

(1) Mechanical apparatus shall be provided and used for carrying material, when the safety of the worker is jeopardized.

(2) (Subsection revoked).

(3) Workers assigned to the handling of material shall be instructed as to the manner of lifting and carrying loads.

(4) Where an inclined plane is used for raising or lowering heavy objects, the following steps shall be followed:

(a) avoid standing on the lower end of the plane; and

(b) control the displacement of such object by means of cables, blocks, wedges or other devices.

(5) Where rollers are utilized for moving objects, tools designed for this type of work shall be used instead of hands or feet to change the position of the moving rollers.

(6) Suitable protective equipment in accordance with Subdivision 2.10 shall be used for the manual transportation of objects having sharp and dangerous edges, and for the handling of burning, caustic or corrosive substances.

(7) The handling of heavy objects by a crew shall be done with signals that are well understood and designed to ensure uniform manoeuvering.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.3; O.C. 329-94, ss. 63 and 66.

3.16.9. Conveyors:

(1) The resistance of a structural element of a conveyor shall be calculated so that it supports the loads and overloads to which it is exposed.

(2) A mobile element shall be protected in accordance with section 3.10.13 if it is located less than 2 m away from an accessible floor.

(3) A conveyor installed above a walkway or a work area shall be equipped with a protector to prevent objects from falling.

(4) An aerial conveyor in which workers are required to travel shall be equipped with a runway along its entire length and with an emergency stopping device installed at each work location.

(5) It is forbidden for a worker to climb on a conveyor or to stand on the structure supporting it.

(6) A vertical bucket conveyor shall be surrounded by a full protector:

(a) covering the entire length; and

(b) equipped with a door or with a detachable part for inspection, cleaning and repairs; the opening of this door or the removal of a detachable part shall produce automatic stopping of the conveyor.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.4; O.C. 329-94, ss. 64 and 66; O.C. 606-2014, s. 20.

3.16.10. Handling, storage and use of hazardous products

(1) The storage and handling of hazardous products within the meaning of the Hazardous Products Information Regulation (2015, chapter 13, section 14) must be carried

out in such a way as to prevent them from spilling or catching fire. The following measures must therefore be taken:

(a) any hazardous products which, in mixing with other substances, may cause fires or explosions or may give off flammable or toxic gases must be stored separately;

(b) receptacles, piping and apparatuses must be kept in good condition;

(c) floors or shelves onto which a product has been spilled must be cleaned immediately.

(2) Only the quantity of flammable liquid and combustibles contained in safety bottles and necessary for one work day may be stored in a building under construction, unless it can be stored in a ventilated room having a fireresistance of at least 2 hours.

(3) Where a hazardous product is transferred from one container to another, the receptacle used must be safe for use with the product transferred.

(4) In addition to conforming to section 3.13.5, any cylinder of gas under pressure must not be

(a) lifted with slings or magnets; or

(b) exposed to shocks, in particular, shock caused by a fall.

(5) Oxygen must not be used for the following purposes:

- (a) operating compressed-air tools;
- (b) ventilation;
- (c) obtaining pressure;
- (d) cleaning;
- (e) starting internal combustion engines.

(6) Corrosive materials must be handled with care and must be stored:

- (a) away from areas in which the risk of fire is high;
- (b) away from combustive substances;
- (c) away from direct sunlight;
- (d) in cool and well ventilated areas;
- (e) in receptacles that are kept closed.

(7) Toxic or dangerously reactive substances must be stored:

(a) away from areas in which the risk of fire is high and away from heat sources;

- (b) away from combustive substances;
- (c) in cool and well ventilated areas.

For the purposes of this section, "corrosive materials", "oxydizing material", "toxic substances" and "dangerously reactive substances" mean a hazardous product that belongs to the corresponding hazard classes in the table provided for in section 70 of the Regulation respecting occupational health and safety.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.16.5; O.C. 995-91, s. 10; O.C. 329-94, s. 66; 2015, c. 13, s. 15; O.C. 805-2020, s. 4.

§3.17 (Revoked).

3.17.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.17.1; O.C. 1959-86, s. 24; O.C. 425-2010, s. 5.

3.17.2.-3.17.9. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 425-2010, s. 5.

§3.18. Demolition

3.18.1. Requirements prior to demolition:

(1) The principal contractor must send to the Commission a demolition notice giving the method used, at least 7 days before the beginning of wrecking work on a building or structure. In emergency cases the notice must be sent as soon as possible before the beginning of wrecking work; such emergency must be proven by the principal contractor.

Wrecking of any prestressed or poststressed concrete slab or structure shall be made according to a method approved by an engineer qualified in the field.

(2) Where mechanical wrecking is used, the notice must include the power of the machine, weight of the wrecking ball, space reserved for debris, successive wrecking steps, restrictions from and agreements with public utilities.

(3) The consumer supply mains for water, gas and electricity and the other services shall be cut off at the place and in the way specified by the authorities involved. These service mains shall be relocated to avoid any damage and shall not constitute a hazard for workers and the public.

(4) Any building or part of building in demolition shall be solidly braced or supported to avoid any collapse which could be dangerous for the workers.

(5) Provision shall be made for the necessary supports with respect to adjoining properties in order to ensure their stability and to avoid any collapse.

(6) Any glass shall be removed from outside openings prior to any demolition operations.

(7) Any sidewalk or street running alongside a demolition site shall be isolated from the latter by a covered sidewalk closed on the demolition side as stipulated in section 2.7.2. However, this covered sidewalk may be replaced by a barrier if this sidewalk or street is located at a distance exceeding half the height of the building, but this building shall not measure more than 7.5 m in height.

(8) (Subsection revoked).

(9) The public shall be prevented from entering a building which is marked for demolition.

(10) Construction signs shall be installed wherever the public may be exposed to any danger.

(11) All demolition works shall be done under the continuous supervision of a competent foreman.

(12) No employer shall employ for demolition work any worker younger than 18 years of age.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.1; O.C. 1959-86, s. 25; O.C. 995-91, s. 11; O.C. 329-94, s. 67.

3.18.2. Requirements during demolition:

(1) During demolition, the following precautions shall be taken:

(a) dust shall be kept down;

(b) debris shall not be burnt on the site without authorization;

(c) floors shall not be overloaded beyond their capacity;

(d) solid supports shall be provided in areas where there are gins, derricks and other similar hoisting apparatus;

(e) any scaffolding or platform shall be erected in compliance with Subdivisions 3.7 and 3.9 of this Code;

(f) materials with protruding nails shall be stacked or removed or nails shall be removed or hammered in;

(g) entrances and exits which are protected against hazards shall be provided;

(*h*) in buildings more than 7.5 m high, exterior openings shall be boarded up within 6 m of any shaft or chute for debris;

(*i*) floor openings below the demolition level which are not being used for the removal of debris shall be boarded up;

(j) stairways and ramps shall be kept as long as possible; and

(k) Subdivision 3.1 applies even when demolition works are suspended.

(2) Demolition works shall be done with the following provisions:

(*a*) demolition work shall proceed systematically from the roof down to the ground, unless another demolition procedure has been approved by an engineer;

(b) demolition and cleaning of a storey shall be finished before its supports give way or are removed;

(c) the steel frame may be left in place during demolition of the masonry. However, any material shall be removed from the framework as the removal of the masonry progresses;

(d) no beam, column or other structural part may be cut or separated from the others without having been previously separated from all supported loads;

(e) solid shores for the remaining structure shall be provided during the removal of masonry or concrete floors. Plankings or walkways shall be provided for workers and the area below such work closed to workers;

(f) masonry shall be demolished in reasonably level layers and not in large masses, so as to avoid affecting the solidity of the structure and its supports;

(g) cornices and other projections shall be supported until removed;

(*h*) no worker shall stand on any wall, pier or chimney unless scaffolding is provided on any side at a distance not exceeding 3 m below the level at which he is working;

(i) it is forbidden to leave without protection any wall, chimney or other structural part which might collapse under the effect of wind or vibrations;

(*j*) the demolition site shall be constantly inspected during the course of the works to prevent any eventual accident; and

(k) sand layers shall be used to break the fall of materials that pose a danger to the health and safety of workers.

(3) Removal, and transportation of debris shall be carried out in the manner described in section 3.2.2.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.2; O.C. 329-94, s. 68.

3.18.3. Mechanical method of demolition: Demolition by smashing walls or floors with a swinging weight, mechanical shovel, or other similar equipment shall be carried out according to the following provisions:

(a) the section which is to be mechanically demolished shall not be higher than 25 m from the ground;

(b) whenever the equipment is in operation, access to the section to be demolished is prohibited to all workers, except the operators of that equipment;

(c) if the public has access to the part to be demolished, a protective wall shall be placed around the building at a distance at least equal to 1.5 times the height of the part to be demolished or, if the public does not have access thereto, a demolition zone shall be marked off around the building by hazard signs placed at a distance equivalent to that prescribed for the protective wall; (d) it shall be possible to manipulate the controls of the equipment used in such method of demolition at a safe distance from the points of impact; and

(e) where a swinging weight is used, the supporting cables shall be of such a length that it is not possible for the weight to swing against any structure other than the structure being demolished.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.3; O.C. 1959-86, s. 26; O.C. 329-94, s. 69; O.C. 1413-98, s. 25.

3.18.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.4; O.C. 1959-86, s. 27; O.C. 329-95, s. 70.

3.18.5. Requirements following demolition: The following measures shall be taken after demolition:

(a) cellars and excavations shall be rendered inaccessible to the public by protective walls, unless they have been completely backfilled to grade;

(b) streets, sidewalks or public roads shall be free from all temporary obstruction required by the work and shall be returned to their original state; and

(c) the demolition site shall be left in such condition that no accident, fire or health hazard has been created.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.5; O.C. 1413-98, s. 26.

3.18.6. Sections 3.18.1, 3.18.2 and 3.18.5 apply to all types of demolition.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.18.6.

§3.19. Work on stilts

3.19.1. Work carried out on stilts or other similar apparatus is forbidden at all times.

R.R.Q., 1981, c. S-2.1, r. 6, s. 3.19.1.

§3.20. Sandblasting

3.20.1. The wearing of an air-supplied hood, gloves and clothing designed to ensure protection from dust and abrasive or metal projections is compulsory for any worker using an abrasive air blaster unless the worker is isolated from the process.

O.C. 1959-86, s. 28; O.C. 885-2001, s. 368; O.C. 48-2022, s. 4.

3.20.2. (*Revoked*).

O.C. 1959-86, s. 28; O.C. 885-2001, s. 369; O.C. 48-2022, s. 5.

3.20.3. Air flow: The air flow supplying the hood must be sufficient to prevent the admission of dust and suspended particles in the air breathed by the workman and must not be less than 165 litres per minute or greater than 430 litres per minute.

O.C. 1959-86, s. 28.

3.20.4. Maintenance of equipment: The employer must maintain the equipment as recommended by the manufacturer and keep it clean.

O.C. 1959-86, s. 28.

3.20.5. Ventilation: Where work with sandblasting is performed indoors, the area must be isolated and ventilated by extraction.

O.C. 1959-86, s. 28.

3.20.6. Abrasive blasting - Changing room and shower: For abrasive blasting, the employer must provide workers with a changing room that complies with section 3.2.11 and a shower that complies with section 3.2.15.

O.C. 885-2001, s. 370; O.C. 393-2011, s. 15.

§3.21. Work in confined space

3.21.1. Before work begins in a confined space, the principal contractor, together with the employer, specifies in writing:

(a) the tools required for the work;

(b) the equipment installed or to be installed in the confined space and the measures to be taken for its installation, use, maintenance and protection or to move it;

(c) the pipe and conduits entering the confined space;

(d) the hazards, and corresponding safety measures to be taken according to the work;

(e) contaminants which may be found in or around the confined space;

(f) the life saving devices and equipment needed and the corresponding emergency measures.

That information must be available on the job-site.

O.C. 1959-86, s. 28.

3.21.2. In confined spaces, the employer must take the contaminant concentration readings mentioned in subparagraph *e* of the first paragraph of section 3.21.1 and make sure that no worker, even when wearing a respirator, enters a confined space containing or which may contain:

(a) oxygen in a proportion lower than 19.5% or higher than 23%;

(b) flammable gas or vapour with a concentration higher than 25% of the lower explosion limit;

(c) contaminants in the air with a concentration higher than the threshold limit values mentioned in Schedule I to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13);

(d) any other contaminant in a concentration which could be hazardous to a workman's health.

O.C. 1959-86, s. 28; O.C. 885-2001, s. 377; O.C. 48-2022, s. 6.

3.21.3. Concentration readings: The employer must ensure that the concentration readings prescribed in section 3.21.2 are:

(a) taken every 2 hours in the most hazardous spots, by a person trained for such purpose;

(b) taken so as to obtain a precision equal to that yielded by the methods described in section 44 of the Regulation respecting occupational health and safety (chapter S-2.1, r. 13);

(c) entered in the record on the job site, except if the recordings are made with continuous recorders equipped with automatic alarms setting off when the quality of air does not meet the requirements of section 3.21.2.

O.C. 1959-86, s. 28; O.C. 885-2001, s. 371.

3.21.4. The employer must install a ventilation system powerful enough to ensure compliance with the concentrations prescribed in paragraphs a, b, c, and d of section 3.21.2 in and around the confined space.

O.C. 1959-86, s. 28; O.C. 329-94, s. 71.

3.21.5. The employer must ensure that measures are taken to prevent workmen who are not assigned to work in a confined space from entering that space.

O.C. 1959-86, s. 28.

3.21.6. Emergency: Where the environment is unhealthy and the situation endangers the life or safety of a person, the employer:

(a) must ensure that the emergency measures prescribed in section 3.21.1 are taken; and

(b) may, despite section 3.21.2, allow a workman to enter the confined space if such workman has had special training for such circumstances and if he is informed of the hazards he will be exposed to.

O.C. 1959-86, s. 28.

§3.22. Work in isolated locations

3.22.1. Where a workman works alone in an isolated location where it is impossible for him to call for help, the employer must provide an efficient intermittent or continuous watch.

O.C. 1959-86, s. 28.

§3.23. Work liable to produce asbestos dust emissions

3.23.0.1. For the purposes of this subdivision, any material and product contains asbestos where the asbestos concentration is of at least 0.1%.

In that respect, the second paragraph of section 69.5 of the Regulation respecting occupational health and safety applies.

O.C. 476-2013, s. 5.

3.23.1. This subdivision applies to any construction site where work liable to produce asbestos dust emissions is carried out.

O.C. 54-90, s. 3.

3.23.1.1. For the purposes of this subdivision,

"protective clothing" means clothing that:

(a) resists the penetration of asbestos fibres;

(b) covers the worker's body, excluding his face, hands and feet;

(c) is closed at the neck, wrists and ankles;

"wetting agent" means a surfactant or liquid detergent added to water according to the manufacturer's instructions in order to increase its ability to penetrate materials containing asbestos;

"work carried out outside" means work entirely carried out elsewhere than in a building used, having been used or intended to be used to shelter or receive persons, animals or things.

O.C. 459-99, s. 1; O.C. 645-2022, s. 1.

3.23.2. For the purposes of this subdivision, the following categories of sites are established:

(1) sites where low-risk work is carried out:

(a) the installation, handling or removal of manufactured goods containing asbestos, provided they are and remain in a non-friable condition, such as:

i. vinyl tiles;

ii. acoustic tiles;

iii. gaskets;

iv. seals;

v. asbestos cement products;

(b) the sawing, cutting, shaping or drilling of a product mentioned in subparagraph a of this paragraph with a hand tool or a power tool fitted with a dust-collection device equipped with a high-efficiency filter that fully covers the work area;

(c) (paragraph revoked);

(2) sites where moderate-risk work is carried out:

(a) the total or partial removal of false ceilings for the purpose of gaining access to a work area where friable materials containing asbestos are found;

(b) the enclosure of friable material containing asbestos, subject to subparagraph c of paragraph 3;

(c) the removal of friable material containing asbestos where, in the removal process, the work area is sealed off from the worker's breathing area;

(d) any work that is liable to produce asbestos dust emissions and that is not classified as low or high-risk;

(e) the handling or removal, otherwise than in the case referred to in subparagraph f, of small quantities of friable material containing asbestos having a volume of debris not exceeding 0.03³ for each minor renovation or regular specific maintenance job;

f) the removal of drywall installed with asbestos joint-filling compounds;

(3) sites where high-risk work is carried out:

(a) subject to subparagraphs c and e of paragraph 2, the handling or the removal of friable material containing asbestos;

(b) the cleaning or removal of a ventilation system, including rigid ducts, in buildings where the insulation contains asbestos applied by spraying;

(c) the enclosure of friable material containing asbestos by the spray application of a sealant;

(d) the repair, alteration or demolition of kilns, boilers or similar devices made entirely or partly of refractory materials containing asbestos;

(e) the use of a power tool not fitted with a dustcollection device equipped with a high-efficiency filter that fully covers the work area to grind, cut, drill or abrade a product mentioned in subparagraph *a* of paragraph 1;

(f) subject to subparagraph e of paragraph 2, the handling or removal of friable material containing crocidolite or amosite;

(g) subject to subparagraph *e* of paragraph 2, the total or partial removal of false ceilings on which friable materials containing asbestos is found.

O.C. 54-90, s. 3; O.C. 459-99, s. 2; O.C. 645-2022, s. 2.

3.23.3. The employer shall determine the types of asbestos present in the materials before undertaking work liable to generate asbestos dust.

O.C. 54-90, s. 3; O.C. 459-99, s. 3.

3.23.3.1. The use of crocidolite or amosite or of a product containing either of those materials is prohibited, except where their replacement is not reasonable or feasible in practice.

O.C. 459-99, s. 3.

3.23.3.2. Before demolition work is undertaken, materials liable to generate asbestos dust shall be removed. $\overline{O.C. 459-99. s. 3.}$

3.23.4. The spray application of a mixture of friable materials containing asbestos and the installation of friable insulating materials containing asbestos are prohibited.

O.C. 54-90, s. 3; O.C. 459-99, s. 4.

3.23.5. The use of compressed air is prohibited in a work area covered by this subdivision, except compressed air necessary to operate a respirator.

O.C. 54-90, s. 3.

3.23.6. Smoking, eating, drinking or chewing any substance in a work area covered by this subdivision is prohibited.

O.C. 54-90, s. 3; O.C. 459-99, s. 5.

3.23.7. Before undertaking work liable to emit asbestos dust, the employer must train the worker and inform him of the risks, prevention methods and safe working methods. The training and information program must contain at least:

(1) the employer's general obligations;

(2) the effects of asbestos on health;

(3) the standards applicable and the sampling to be carried out;

(4) the worker's rights and obligations;

(5) individual and common protective devices and equipment;

(6) the tasks to be carried out and the equipment and tools to be used;

(7) safe working methods and procedures;

(8) prevention and verification methods.

The information and training provided for in the first paragraph must have previously been established in writing.

O.C. 54-90, s. 3; O.C. 459-99, s. 6.

3.23.8. Before undertaking in a building work covered by this subdivision:

(1) all furniture shall be removed from the work area or protected by airtight film;

(2) all friable materials that contain asbestos and that are spread in the work area shall be removed in accordance with either of the following methods:

(a) after having wet those materials thoroughly using a wetting agent;

(b) with a vacuum cleaner equipped with a high-efficiency filter.

O.C. 54-90, s. 3; O.C. 459-99, s. 7; O.C. 645-2022, s. 3.

3.23.9. For the duration of work carried out inside a building, friable materials that contain asbestos and that are likely to be spread shall be kept thoroughly wetted using a wetting agent.

For the duration of work carried out outside, the employer shall prevent the dispersal of the dust of friable materials containing asbestos by spraying them. The employer shall ensure that those materials are kept wet or covered in order to prevent their dispersal.

Wetting is to be used, except where the procedure may create a danger to the health, safety and physical integrity of the worker and where the danger cannot be eliminated by another means.

O.C. 54-90, s. 3; O.C. 459-99, s. 7; O.C. 645-2022, s. 4.

3.23.9.1. Before moving kilns, boilers or similar devices made entirely or partly of refractory materials containing asbestos, the employer shall cover them entirely with an airtight film.

O.C. 459-99, s. 7.

3.23.10. During work inside a building, debris of materials containing asbestos shall be placed in airtight containers appropriate to the type of debris, regularly during the work shift and at the end of the work shift. Debris shall be removed by means of a vacuum cleaner equipped with a high-efficiency filter or by first wetting the debris using a wetting agent.

Where work is being carried out outside, the employer shall also prevent the dispersal of the debris of materials containing asbestos by using airtight film or any other equivalent means. The employer shall dispose of the debris using airtight containers, film, or any other means making it possible to ensure airtightness during transportation, depending on its intended use.

O.C. 54-90, s. 3; O.C. 459-99, s. 7; O.C. 645-2022, s. 5.

3.23.11. Upon completion of work where airtight drop sheets were used to protect the work area, drop sheets intended for re-use must be cleaned with a vacuum cleaner equipped with a high-efficiency filter. Drop sheets intended for disposal must first be wetted, then folded so that they hold all the dust that they have collected and, finally, placed in an airtight container.

O.C. 54-90, s. 3.

3.23.12. Upon completion of work covered by this subdivision, the work area and the area around it must be cleaned with a vacuum cleaner equipped with a high-efficiency filter or by damp wiping the surfaces and then cleaning them.

O.C. 54-90, s. 3.

3.23.12.1. The outside of containers for the debris of materials containing asbestos, the tools and the equipment shall be cleaned by wetting or by means of a vacuum cleaner equipped with a high-efficiency filter immediately before they are removed from the work area.

O.C. 645-2022, s. 6.

3.23.13. A label must be affixed to any receptacle containing asbestos materials, whether new or used, unless the receptacle has already been labelled by a supplier in accordance with the Hazardous Products Regulations (SOR/2015-17).

The label must be permanently affixed and legible and must bear the following indications:

Material containing asbestos

Toxic by inhalation

Keep container tightly closed

Do not breathe the dust

O.C. 54-90, s. 3; O.C. 459-99, s. 8; 2015, c. 13, s. 16.

3.23.14. Any worker working in a work area covered by this subdivision must wear protective footwear that meets the requirements of section 2.10.6 and that is fitted with soles that do not slip on wet surfaces.

O.C. 54-90, s. 3; O.C. 459-99, s. 9.

3.23.14.1. On a construction site where low-risk work is being carried out, except for the work referred to in subparagraph *a* of paragraph 1 of section 3.23.2, an employer shall ensure that any worker present in the work area wears a respirator.

O.C. 459-99, s. 10; O.C. 885-2001, s. 372; O.C. 48-2022, s. 7.

3.23.15. On a construction site where moderate-risk work is being carried out, an employer, in addition to the obligations provided for in sections 3.23.3 to 3.23.14, shall comply with the following:

(1) the employer shall ensure that any worker present in the work area is wearing a reusable protective respiratory apparatus equipped with a 100 series or HEPA high efficiency filter certified by the NIOSH;

(2) notwithstanding paragraph 1, a respirator that complies with the provisions of paragraph 1 or 2 of section 3.23.16 shall be worn in the following cases: (a) for the handling or removal of friable material containing crocidolite or amosite;

(b) for any work covered by subparagraph d of paragraph 2 of section 3.23.2;

(3) protective clothing shall be worn by any person present in the work area and the clothing worn shall be used exclusively for carrying out such work;

(3.1) the employer must provide workers with a changing room that complies with section 3.2.11;

(4) the employer shall ensure that the protective clothing is clean and dry at the beginning of each day on which it is to be used;

(4.1) where a worker wearing disposable protective clothing leaves the work area, the employer shall ensure that the clothing is immediately placed in a hermetically sealed airtight container or in a receptacle filled with water or supplied by him until it is washed;

(5) the employer shall cause reusable protective clothing to be washed or shall cause it to be cleaned by means of a vacuum cleaner equipped with a high-efficiency filter before reuse;

(6) where a worker wears winter work clothes, the employer shall provide disposable protective clothing so that the worker can at all times wear 2 layers of disposable protective clothing over his winter work clothes;

(7) where a person wearing disposable protective clothing leaves the work area, the employer shall ensure that the clothing is immediately placed in a hermetically sealed airtight container supplied by him;

(8) the employer shall ensure that a worker does not wear or carry his work clothes and protective footwear outside the work area, unless they have been washed or cleaned by means of a vacuum cleaner fitted with a highefficiency filter;

(9) during work to recover friable materials containing asbestos, the employer shall isolate the work area with an enclosure made of materials impervious to asbestos fibres that has an exhaust ventilation system equipped with a high-efficiency filter providing at least 4 changes of air per hour;

(9.1) during work to handle or remove friable materials containing asbestos having a volume of debris not exceeding 0.03 m³, the employer shall isolate the work area with an enclosure made of materials impervious to asbestos fibres that has an exhaust ventilation system equipped with a high-efficiency filter providing at least 4 changes of air per hour;

(9.2) during work to remove friable materials containing asbestos in a work area sealed off from the worker's

breathing area, if the worker uses a glove bag, the employer shall ensure

(*a*) that it is used solely for the purposes and conditions for which it was designed, according to the manufacturer's instructions;

(b) that it is not reused after it has been filled;

(c) that it is not used if there is a risk that it will not remain hermetically sealed during the work, in particular due to the location of the tube, the deterioration of the insulation or the temperature of the tube, duct or structure; and

(d) that, before the glove bag is dismantled, any part of the tube where insulating materials that could release asbestos fibres is encapsulated, and that the glove bag is sealed above the debris of materials so that the debris is isolated from the upper compartment;

(10) during work to remove false ceilings for the purpose of gaining access to a work area where friable materials containing asbestos are found, the employer shall protect the building's ventilation system from any contamination and isolate the work area with an enclosure made of materials impervious to asbestos fibres that has an exhaust ventilation system equipped with a high-efficiency filter providing at least 4 changes of air per hour;

(11) the employer shall post a sign at the entrance to each work area. That sign shall be yellow, 500 mm high by 350 mm wide and shall indicate in black letters of the size specified below the following information in the following order:

Information	Size of letters
ASBESTOS	50 mm
DANGER	40 mm
Do not breathe dust	15 mm
Protective equipment must be worn	15 mm
No admittance	15 mm
Inhaling asbestos dust may be harmful to your health	10 mm;

(12) where there is no enclosure such as referred to in paragraphs 9, 9.1 and 10, the work area shall be marked off by hazard signs.

O.C. 54-90, s. 3; O.C. 459-99, s. 11; O.C. 885-2001, s. 373; O.C. 393-2011, s. 16; O.C. 48-2022, s. 8; O.C. 645-2022, s. 7.

3.23.15.1. On a construction site where moderaterisk work is being carried out, workers must, before removing their protective clothing and other individual protective equipment, decontaminate them using a wet cloth or a vacuum cleaner equipped with a high-efficiency filter.

O.C. 645-2022, s. 8.

3.23.16. On a construction site where high-risk work is being carried out, other than the work mentioned in section 3.23.16.1, an employer shall fulfil the obligations provided for in section 3.23.15, except those provided for in paragraphs 1, 2, 4.1 and 5 of that section, and the following obligations:

(1) the employer shall ensure that any worker present in the work area during the use of electric tools not fitted with a dust collector equipped with a high-efficiency filter or during the handling of thoroughly wetted friable materials containing asbestos is wearing a full-facepiece respirator; the respirator must comply with one of the following types:

(a) a powered air-purifying respirator with a HEPA particulate filter;

(b) a supplied-air respirator operated in continuousflow positive-pressure mode or in a pressure-demand mode with positive pressure;

(2) notwithstanding paragraph 1, a supplied-air and continuous-flow positive-pressure adjusted, or pressure demand and positive pressure, full-facepiece respirator must be worn by any worker who is in one of the following situations:

(a) in the presence of friable materials containing asbestos that are not thoroughly wetted;

(b) in the presence of crocidolite or amosite, where the readings made under paragraph 4 give concentrations equal to or greater than 10 fibres/cm³;

(3) before work begins, the employer and the principal contractor shall identify in writing the following information and make it available in the work area:

(a) the equipment and tools necessary to carry out the work and the measures to be taken in order to install, use, care for, protect and move them;

(b) the hazards and the health and safety measures to be taken depending on the work to be carried out;

(c) the types of asbestos and other contaminants that may be encountered during the work;

(d) the individual and group protective devices and equipment that must be used;

(e) the measures to be taken in case of emergency, which shall include, in particular, locating emergency exits in the work area and the exits through which the building can be evacuated;

(4) the employer shall take a sample of the concentration of airborne breathable asbestos fibres in the work area, in accordance with section 44 of the Regulation respecting occupational health and safety (chapter S-2.1, r. 13), at least once per shift during the work, send it immediately to a laboratory for analysis and take reasonable measures

to obtain the results of those analyses within 24 hours; the results shall be recorded in a register that is available on the work premises during all the work;

(5) the employer shall ensure that reusable protective clothing is washed before it is reused;

(6) the employer must provide workers working in the work area with a double changing room that complies with section 3.2.13;

(7) the employer shall ensure that any worker leaving the work area follows the decontamination procedure described below:

(a) workers shall remove their disposable protective clothing in the work clothes changing room and treat them as waste or shall remove their reusable protective clothing and put it immediately in a receptacle filled with water or, where clothes are washed in the work clothes changing room, in the tub of a washer filled with water;

(b) workers shall remove their work clothes and protective footwear in the work clothes changing room and those articles, before being put away, shall be washed or cleaned by means of a vacuum cleaner equipped with a high-efficiency filter;

(c) workers shall wash and remove their safety helmets and respirators under the shower; disposable cartridges shall be thrown into a garbage can and the other parts of the respirator shall be washed under the shower and then hung to dry in a clean area free of dust;

(d) workers shall shower immediately before entering the street clothes changing room;

(e) work clothes and protective footwear shall be washed before being transported outside the work premises referred to in this section; where the work clothes are winter clothes, they shall be cleaned by means of a vacuum cleaner equipped with a high-efficiency filter and placed in an airtight bag and the employer shall cause them to be dry cleaned and water-proofed;

(8) the work area and the work clothes changing room shall be sealed off from the rest of the building by an airtight enclosure that has an exhaust ventilation system. The ventilation system shall meet the following standards:

(a) it shall be equipped with a high-efficiency filter;

(b) it shall provide at least 4 changes of air per hour;

(c) it shall ensure negative pressure of between 1 and 4 Pa;

(9) notwithstanding paragraph 8, when work is done outdoors, an airtight enclosure is required only for the work clothes changing room; in such case, the travel area of workers that connects the work area and the work clothes changing room shall be marked off by hazard signs; (10) at the beginning and end of each shift, the employer shall ensure that the airtight enclosure is in good condition. If the enclosure is punctured or becomes defective, the work shall cease until the enclosure is repaired;

(11) the air vents of the building's ventilation system shall be sealed off from the work area before the work begins and shall be kept sealed during the work;

(12) upon completion of the work, it shall be prohibited to dismantle the airtight enclosure or to remove the airtight drop sheets before the concentration of airborne respirable asbestos fibres in the work area drops to less than 0.01 fibres/cm³. That reading shall be taken in accordance with section 44 of the Regulation respecting occupational health and safety.

O.C. 54-90, s. 3; O.C. 459-99, s. 11; O.C. 885-2001, s. 374; O.C. 393-2011, s. 17; O.C. 48-2022, s. 9; O.C. 645-2022, s. 9.

3.23.16.1. An employer handling or removing friable materials containing asbestos whose volume of debris exceeds 0.03 m^3 without exceeding 0.3 m^3 , for each minor renovation or specific regular maintenance work, shall fulfil the obligations provided for in paragraphs 3, 3.1, 4, 6 to 9 and 10 to 12 of section 3.23.15, those provided for in paragraphs 1 and 2, in subparagraph *e* of paragraph 7 and paragraphs 10 and 11 of section 3.23.16, as well as the following obligations:

(1) the employer shall ensure that disposable protective clothing is used;

(2) the work area shall be sealed off from the rest of the building by an airtight enclosure that has an exhaust ventilation system equipped with a high-efficiency filter providing at least 4 changes of air per hour; notwithstanding the foregoing, where work is done outdoors, the airtight enclosure is not required;

(3) the following decontamination procedure shall be applied before any worker leaves the work area:

(a) the enclosure shall be cleaned by wetting or by means of a vacuum cleaner equipped with a high-efficiency filter;

(b) after removal of protective clothing, the respirator and the safety helmet shall be cleaned by wetting;

(c) the worker shall wash the parts of his body that have been exposed to dust containing asbestos.

O.C. 459-99, s. 11; O.C. 393-2011, s. 18; O.C. 645-2022, s. 10.

§3.24. Steel structure erection or dismantling work

3.24.1. Scope: This subdivision applies to work for the erection or dismantling of a steel structure including any sheet-metal apron, but excluding work to erect or dismantle an electrical transformer station, a telecommunications

tower or a power line tower used for electric power transportation or distribution.

O.C. 391-2011, s. 2.

3.24.2. Telecommunications system: A telecommunications system used by a worker who guides a maneuver under subsection 2 of section 3.10.5 must be bidirectional, hands-free and transmit on a radio frequency dedicated exclusively to the work in progress.

The work in progress must be interrupted in case of telecommunications system failure.

O.C. 391-2011, s. 2; O.C. 1078-2015, s. 6.

3.24.3. Means of access: The means of access provided for in Subdivisions 3.5. and 3.6 must be installed so that horizontal movements of workers on trusses, beams and joists do not exceed 30 m.

O.C. 391-2011, s. 2.

3.24.4. (*Revoked*).

O.C. 391-2011, s. 2; O.C. 63-2025, s. 15.

3.24.5. Presence of a first-aider: Despite section 7 of the First-aid Minimum Standards Regulation (chapter A-3.001, r. 10), during the work, the principal contractor must ensure that one first-aider within the meaning of that Regulation is present at all times on the work premises.

O.C. 391-2011, s. 2.

3.24.6. Protruding parts: Where studs or other protruding parts constitute a danger for workers on the steel structure, a temporary sidewalk must be installed to ensure the safe movement of workers.

O.C. 391-2011, s. 2.

3.24.7. Plans and procedures: The plans and procedures provided for in this subdivision must be kept on the work premises and be accessible.

O.C. 391-2011, s. 2.

—Erection of a steel structure

3.24.8. Preparation of the work area: Before the beginning of the steel structure erection work, the principal contractor must verify

(1) that the concrete foundations are backfilled and the work area, including the unloading area, is levelled, drained and of a sufficient dimension to store materials; and

(2) that the resistance of the soil under the work area allows support of hoisting apparatus, trucks and loads during the work.

O.C. 391-2011, s. 2.

3.24.9. Erection plans: Work must be carried out according to the steel structure manufacturer's plan. The plan must include

(1) the location of the various components of the structure and their erection marks;

(2) the main dimensions and their respective levels;

(3) the type of bolts used and their diameter;

(4) the welds to be done on site;

(5) the temporary structural components; and

(6) the number of rows of bridging and their position, if applicable.

O.C. 391-2011, s. 2.

3.24.10. Erection procedure: In addition to being carried out in accordance with the erection plan provided for in section 3.24.9, the work must be carried out according to a procedure established by the employer. The procedure must contain

(1) the installation method and the erection stages of the structure;

(2) the measures to be taken to ensure that the structure components are stable; and

(3) the measures to protect workers from falling.

In addition to the requirements provided for in the first paragraph, the erection procedure must include the hoisting procedure provided for in section 3.24.15.

O.C. 391-2011, s. 2.

3.24.11. Anchor rods: Column anchor rods must be installed according to an anchoring plan provided by the steel structure manufacturer. The plan must

(1) contain the following information:

(a) the dimensions of the anchor rods and their position;

(b) the details necessary for securing the anchor rods;

(2) provide an erection procedure where the columns are anchored with fewer than 4 anchor rods or where the position of the anchor rods does not ensure stability of the columns in all their axes.

In addition, column anchor rods must resist the application of a vertical construction load of at least 1.33 kN located 45 cm from the column face in each of its axes and at its maximum height.

O.C. 391-2011, s. 2.

3.24.12. Prior certification: Before the beginning of the work, the principal contractor must obtain from an engineer a certification according to which the anchor rods were installed in accordance with the anchoring plan provided for in section 3.24.11 and the concrete

foundations reached the strength required to support the erection of the steel structure.

O.C. 391-2011, s. 2.

3.24.13. Anchor rod modification or repair: During the work, the principal contractor must obtain from an engineer a new certification following any modification to or repair of column anchor rods that is subsequent to their installation.

O.C. 391-2011, s. 2.

3.24.14. Stacks of shims: Each column that stands on concrete foundations must rest on at least 2 stacks of shims of at least 9 in2 and located near the anchor rods, unless another levelling device offering equivalent safety is provided for in the erection plan.

O.C. 391-2011, s. 2.

3.24.15. Hoisting procedure: A hoisting procedure must be developed when a load is handled

- (1) by more than one hoisting apparatus;
- (2) by a hoisting apparatus other than a crane;

(3) on a pallet by a hoisting apparatus other than a fork lift truck.

Where a hoisting apparatus is anchored to an existing structure, the anchor point and its working load limit must be specified in the hoisting procedure.

O.C. 391-2011, s. 2.

3.24.16. Component weight: Information about the weight of each component of a steel structure to be erected must be accessible on the work premises.

In addition, the weight must be indicated on each component exceeding 500 kg.

O.C. 391-2011, s. 2.

3.24.17. Hoist hooks: Every hook used to hoist a load must present one of the following characteristics:

(1) be equipped with a safety latch;

(2) close under the application of the load and be equipped with a self-locking latch requiring a positive action to unlock the hoist hook.

Where a load is hoisted using a hook referred to in subparagraph 1 of the first paragraph, the load must be hung using a shackle or a wrought alloy steel ring.

Where a load remote unhooking device is used, it must have the following features:

(1) the minimum and maximum loading capacities are conspicuously indicated on the device;

(2) where the device is engaged, it locks under the application of the load; and

(3) it opens only when it no longer withstands the weight of the load and a command to open it is issued. $\overline{O.C. 391-2011, s. 2}$.

3.24.18. Beam-column connection of a multi-span steel structure: During beam-column connection work, a beam must not be supported by a spanner. The type of connection must be designed so as to take that prohibition into account.

The type of beam-column connection may be designed in one of the following manners:

(1) the beam is attached to the column while being supported by a bracket previously attached to the column;

(2) the upper right corner of the plate or angle seat is notched to clear the first supporting bolt of the beam placed previously in the manner specified in Schedule 6.

O.C. 391-2011, s. 2.

3.24.19. Erection of an open web steel joist: Erection work of an open web steel joint must be carried out in accordance with the following standards:

(1) before a hoisting apparatus lands the joist and in order to ensure the lateral stability of the joist during its placement, erection bridging must be installed, if applicable, in accordance with the joist manufacturer's plan. Bridging must be of the bolted diagonal type and the number of rows indicated in the plan must at least comply with the following specifications:

Joist length	Minimum number of rows
(a) less than 12 m	None
(b) 12 m to 18 m	1 row
(c) 18 m to 30 m	2 rows
(d) 30 m or more	4 rows

(2) as soon as a joist is placed, each joist end must be bolted using at least one bolt; and

(3) no load may be placed on the joist as long as all the rows of bridging have not been secured and each end of row has not been anchored, except if the joist manufacturer specifies on the joist connection plan the measures to be taken to do so and those measures have been complied with.

Joists may be hoisted and landed in bundles on the steel structure if they do not require erection bridging and are put in place one after the other to prevent them from falling off their supports.

O.C. 391-2011, s. 2.

3.24.20. Space between girts: If the provisions set out in section 3.10.7 may not be applied to the installation of girts because of the work environment or the height of the steel structure, the vertical space between girts must not exceed 1.6 m and a means of access to allow a worker to go from one girt to another must be provided in the plan or in the hoisting procedure.

O.C. 391-2011, s. 2.

-Dismantling of a steel structure

3.24.21. Obligations of the principal contractor: Before the beginning of the dismantling work of a steel structure, the principal contractor must

(1) locate, if applicable, the electric network of the work area, turn off the power and apply a locking procedure;

(2) locate any gas, steam or liquid line and apply, if applicable, a method for purging the line and a locking procedure.

O.C. 391-2011, s. 2.

3.24.22. Dismantling plan: Work must be carried out according to a plan drawn up by an engineer. The plan must contain

(1) the dismantling method and sequence, including the bolt, rivet and weld removal sequence;

(2) measures to ensure the stability of the hoisting apparatus and structure components. The measures must particularly take into account the following elements:

(a) the load must not exceed 70% of the hoisting apparatus capacity, including hoisting accessories, specified in the applicable load rating chart;

(b) anchor rods must be considered as having no resistance, unless pull-out tests are performed to establish their resistance;

(3) the weight and the centre of gravity of the structure components;

(4) measures to protect workers against falls; and

(5) any other relevant measure to ensure the safe dismantling of the structure.

O.C. 391-2011, s. 2.

§3.25. Work liable to produce crystalline silica dust emissions

3.25.1. Scope: This subdivision applies to any construction site where work involving materials presumed to contain or containing crystalline silica is carried out.

Only sections 3.25.7, 3.25.10 and 3.25.11 apply to the sandblasting work referred to in subdivision 3.20.

O.C. 820-2023, s. 4.

3.25.2. Materials presumed to contain crystalline silica: For the application of this subdivision, the following materials are presumed to contain crystalline silica:

- (a) slate;
- (b) asphalt;
- (c) concrete;
- (d) brick;
- (e) ceramic;
- (f) cement;
- (g) fibrocement;
- (h) granite;
- (i) granulate;
- (j) sandstone;
- (k) mortar.

O.C. 820-2023, s. 4.

3.25.3. Absence of crystalline silica: This subdivision does not apply where the employer has a safety data sheet, a technical description or an analysis carried out according to a recognized method showing that crystalline silica is not present in the material.

A copy of the data sheet or technical description or the results of the analysis must be available at all times during the work on the construction site.

O.C. 820-2023, s. 4.

3.25.4. Measures to control exposure to crystalline silica: Where work involving a material presumed to contain or containing crystalline silica is liable to produce dust emissions, the employer must implement at least one of the following control measures:

(a) the use of a local exhaust ventilation system equipped with a high-efficiency filter;

(b) the use of a procedure to wet the dust emissions;

(c) the isolation of workers from the source of the dust emissions;

(d) the confinement of the source of the dust emissions so that workers are not exposed.

The equipment used to control dust must be used and cared for in accordance with the manufacturer's instructions or a standard offering equivalent safety.

O.C. 820-2023, s. 4.

3.25.5. Closed operation cabin: Where the worker is isolated from the source of the emission of dust from the materials presumed to contain or containing crystalline silica by the use of a closed operation cabin of a mobile machine, the cabin must have the following characteristics:

(a) the air intake in the cabin must be filtered by a high-efficiency filter;

(b) positive pressure must be maintained;

(c) a heating and air-conditioning system must be included;

(d) the door and window joints must be kept in good condition to ensure they are sealed.

O.C. 820-2023, s. 4.

3.25.6. Respiratory protection: When work involving a material presumed to contain or containing crystalline silica is carried out, in addition to one of the control measures listed in section 3.25.4, except in the case of those set out in subparagraph c or d, the wearing of a protective respiratory apparatus is mandatory for all workers present in the work area where one of the following types of work is carried out:

- (a) sawing;
- (b) grinding, sanding or bush hammering;
- (c) jackhammering;
- (d) boring in a confined location;
- (e) drilling.

The protective respiratory apparatus supplied by the employer must provide at minimum an assigned protection factor of 10 and be equipped with a 100 series or HEPA high-efficiency filter.

The obligations set out in section 45.1 of the Regulation respecting occupational health and safety apply where one of the types of work provided for in this section is carried out. In addition, the protective respiratory apparatus must be selected, used and cared for in accordance with CAN/CSA Standard Z94.4-11 Selection, use, and care of respirators.

The wearing of a protective respiratory apparatus is not mandatory where the employer demonstrates that the level of exposure of workers to silica dust is less than the permissible exposure values indicated in Schedule I to the Regulation respecting occupational health and safety.

O.C. 820-2023, ss. 4 and 5.

3.25.7. Training: Before undertaking work referred to in this subdivision, the employer must train and inform workers of the risks, prevention methods and safe working methods. The training and information program must contain at least the following elements:

(a) the materials presumed to contain crystalline silica;

(b) the work that exposes workers to crystalline silica dust;

(c) the effects of exposure to crystalline silica dust on health;

(d) safe working methods and procedures;

(e) the use and care of control equipment and tools for crystalline silica dust;

(f) the wearing and care of individual and common protective equipment.

The information and training provided for in the first paragraph must have previously been established in writing.

O.C. 820-2023, s. 4.

3.25.8. Delimitation of the work area: When the work provided for in section 3.25.6 is carried out, the work area must be delimited by hazard signs. The delimitation must allow workers outside the work area to remain at a safe distance from the area where the work is carried out.

Only workers wearing a protective respiratory apparatus in compliance with section 3.25.6 may access that work area.

O.C. 820-2023, s. 4.

3.25.9. Cleaning of work clothes: Before leaving the work area referred to in section 3.25.8, the worker must remove the work clothes worn and place them in a closed bag supplied by the employer or clean them using either a wet cloth or a vacuum cleaner equipped with a highefficiency filter.

O.C. 820-2023, s. 4.

3.25.10. Cleaning: When the work area and equipment are being cleaned, it is prohibited to use work methods that may cause dust from materials presumed to contain or containing crystalline silica to become suspended in the air, such as dry sweeping or the use of compressed air.

Cleaning must be carried out by means of wetting or using a vacuum cleaner equipped with a high-efficiency filter.

O.C. 820-2023, s. 4.

3.25.11. Debris of materials: Where work is carried out in a building, the debris of materials presumed to contain or containing crystalline silica that is liable to become dispersed in the air must be wetted or placed in closed containers and clearly identified.

Where work is carried out outside, as defined in section 3.23.1.1 of this Code, the debris of materials presumed to contain or containing crystalline silica that is liable to become dispersed in the air must be wetted or an equivalent means that prevents the dispersion of crystalline silica dust in the air must be used.

O.C. 820-2023, s. 4.

DIVISION IV

HANDLING AND USE OF EXPLOSIVES

§4.1. General provisions

4.1.1. No explosive shall be used on a construction site if the following information is not clearly printed or indicated on the explosive box or container:

(a) the manufacturer's name;

(b) the name under which the explosive is known;

(c) the date of manufacture; and

(d) the instructions in French for the safe storage, handling, use and destruction of the explosive.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.1; O.C. 1959-86, s. 29; O.C. 57-2015, s. 2.

4.1.2. Explosives shall be:

(a) protected against impacts, frictions, fire, flames and sparks;

(b) sheltered from rain and snow; and

(c) kept in a ventilated place.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.2; O.C. 57-2015, s. 3.

4.1.3. Explosives bearing the oldest manufacture date shall be used first.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.3; O.C. 57-2015, s. 4.

4.1.4. Deteriorated explosives shall be handled with utmost care and shall be destroyed without delay according to the manufacturer's instructions.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.4; O.C. 57-2015, s. 5.

4.1.5. It is prohibited to use an explosive that has reached its freezing point, unless the manufacturer's instructions allow such use.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.5; O.C. 1959-86, s. 30; O.C. 57-2015, s. 6.

4.1.6. No explosive shall be left lying about.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.6; O.C. 57-2015, s. 7.

4.1.7. An employer must ensure that no one smokes, brings a flame, a substance or material likely to increase the risk of explosion or fire within 8 m from any place where explosives are present.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.7; O.C. 1959-86, s. 31; O.C. 57-2015, s. 8.

4.1.8. An employer must ensure that

(a) explosives are handled and used in accordance with the manufacturer's instructions;

(b) explosives that are brought on the site correspond to the quantities required to carry out the blasting for one workday;

(c) explosives not used to carry out blasting are stored in a depot designed for that purpose; (d) explosives are not transported by hand at the same time as detonators or other blasting accessories.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.1.8; O.C. 1959-86, s. 32; O.C. 57-2015, s. 9.

4.1.9. When there is a risk of accidental blasting by electric induction, caused notably by a radio frequency transmitter or a power line, the employer must favour a non-electrical priming method.

If the employer nonetheless carries out blasting using an electrical priming method, the employer must take all the safety measures required, including the following:

(a) inform the Commission, before the work begins, of the safety measures agreed upon with public utilities where the blasting is near a power line of 125,000 V or more;

(b) place, 300 m around the loading area, signs requiring drivers to turn off the radio transmitter of their vehicle;

(c) isolate electrical circuits and make sure that the detonator leg wires are twisted together when the detonator is inserted in the firing point;

(d) make sure that all equipment emitting radio, electric or magnetic waves

i. are turned off within 15 m around the loading area before the electric detonator is assembled with the lead wires

ii. comply with the distances recommended in the Safety Guide for the Prevention of Radio Frequency Radiation Hazards in the Use of Commercial Electric Detonators, published by the Institute of Makers of Explosives (Safety Library).

O.C. 1959-86, s. 32; O.C. 57-2015, s. 10.

4.1.10. Fireworks, igniter cords, military devices and safety fuses may not be used on a construction site.

Despite section 295 of the Regulation respecting occupational health and safety, this section does not apply to an establishment as defined in section 1 of the Act respecting occupational health and safety.

O.C. 57-2015, s. 11.

4.1.11. Nothing in this Regulation exempts a person from the obligation to comply with the requirements of any applicable act or regulation, particularly with regard to the acquisition, possession, storage, transportation or delivery, handling, use and sale of explosives.

In the event of inconsistency between a provision of this Regulation and a provision of another act or regulation, the stricter standard is to apply.

O.C. 57-2015, s. 11.

§4.2. Shot-firer's certificate

4.2.1. Shot-firer: A person who carries out blasting must hold a shot-firer's certificate issued by the Commission or a body recognized by it.

The certificate is issued until the date of expiry of the general permit held under the Act respecting explosives (chapter E-22) by the shot-firer. The certificate is renewed upon request by its holder as long as renewal of the general permit is granted.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.1; O.C. 1959-86, s. 33; O.C. 1279-98, s. 1; O.C. 57-2015, s. 12.

4.2.1.1. A shot-firer must be in possession of the original of his or her certificate during blasting operations. $\overline{O.C. 57-2015, s. 12}$.

4.2.2. A shot-firer may not receive assistance from more than 2 workers who are not certificate holders.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.2; O.C. 1959-86, s. 33; O.C. 57-2015, s. 13.

4.2.3. In addition to holding a general permit, a candidate for a shot-firer's certificate must

(a) be 18 years of age and older;

(b) provide a document certifying that the candidate's character, knowledge and experience make the candidate competent to handle explosives; and

(c) pass the written examination prepared by the Commission with a mark of at least 80%.

Unless the Commission has suspended or revoked the certificate it issued to the shot-firer, a shot-firer holding a certificate issued by a competent authority in another province or a territory of Canada and recognized by the Commission as equivalent to the certificate issued under this Division is not required to undergo the examination provided for in subparagraph c of the first paragraph.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.3; O.C. 1959-86, s. 34; O.C. 1279-98, s. 2; O.C. 57-2015, s. 14.

4.2.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.4; O.C. 1959-86, s. 35; O.C. 57-2015, s. 15.

4.2.5. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.5; O.C. 1959-86, s. 36; O.C. 57-2015, s. 15.

4.2.6. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.6; O.C. 1959-86, s. 37; O.C. 57-2015, s. 15.

4.2.7.-4.2.8. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 38.

4.2.9. The Commission may suspend or revoke a certificate if the shot-firer

(a) has been the subject, in his or her work, of a remedial order under section 182 of the Act respecting occupational health and safety or of an order under section 186 of that Act, by reason of his or her refusal to comply with the Act or this Regulation;

(b) is found guilty of an offence under section 236 of the Act respecting occupational health and safety in relation to this Division;

(c) no longer holds a general permit issued under the Act respecting explosives (chapter E-22).

The Commission must notify the shot-firer in writing of the suspension or revocation of the certificate.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.2.9; O.C. 1959-86, s. 39; O.C. 57-2015, s. 16.

4.2.10. The Commission must revoke a certificate if the shot-firer is found guilty of an offence under section 237 of the Act respecting occupational health and safety in relation to this Division.

The Commission must notify the shot-firer in writing of the revocation of the certificate.

O.C. 57-2015, s. 16.

4.2.11. An employer must ensure that a worker who acts as shot-firer holds a certificate.

O.C. 57-2015, s. 16.

§4.3. Transportation of explosives

4.3.1. An employer must make sure that a vehicle transporting explosives is in good working order and allows for the safe transportation of explosives, in particular in compliance with the following standards:

(a) the display of safety marks in accordance with the Transportation of Dangerous Goods Regulations (SOR/2001-286);

(b) the part of the vehicle containing explosives must be isolated, fireproof, in compliance with section 45 of the Regulation under the Act respecting explosives (chapter E-22, r. 1) and locked at all times except during the loading or unloading of explosives;

(c) the metal parts likely to come into contact with explosives or their wrapping during transportation must be covered with a material to prevent such contact;

(d) the installation of a tracking and communication system, for a vehicle carrying 2,000 kg or more of explosives, that makes it possible at all times to locate the vehicle and to communicate with its driver. The employer must make sure that a person is in charge of locating and communicating with the driver at all times during the transportation of explosives, as well as alerting police services in case of emergency.

The tracking and communication system provided for in subparagraph d must be installed not later than 26 February, 2018.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.1; O.C. 1959-86, s. 40; O.C. 57-2015, s. 17.

4.3.1.1. An employer must subject the vehicle referred to in section 4.3.1 to a mechanical inspection once a year and remedy without delay the failures observed during such inspection.

Inspection of a vehicle carried out by a holder of a certificate of competency issued under the Highway Safety Code, in the context provided for in the Code or its regulations or under another Act or regulation, stands in lieu of the annual inspection referred to in the first paragraph. Otherwise, the inspection must be carried out by a mechanic whose competency is equivalent to that of the holder of a certificate of competency issued under the Highway Safety Code.

The employer must keep proof that the vehicle has been inspected.

O.C. 57-2015, s. 17.

4.3.1.2. An employer must ensure that objects other than explosives are not transported with the explosives, unless they are stored, or separated from the explosives, in a way that reduces the risk of ignition to a minimum.

Despite the first paragraph, it is prohibited to transport diesel fuel, gasoline or other flammable products with explosives.

O.C. 57-2015, s. 17.

4.3.2. During loading and unloading, the employer must ensure that the driver is accompanied by a person who is responsible for watching over the explosives.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.2; O.C. 1959-86, s. 40; O.C. 57-2015, s. 18.

4.3.3. Where detonators are transported with explosives, the employer must make sure that they are stored separately in a compartment of the vehicle that is completely closed and does not communicate with the part of the vehicle that contains the explosives.

The compartment wall that separates the detonators from the explosives must be as high as the roof and be made of solid wood 150 mm thick or of a material that prevents the explosion of the detonators for at least 1 hour in case of fire.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.3; O.C. 1959-86, s. 41; O.C. 57-2015, s. 19.

4.3.4. (Revoked.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.4; O.C. 1959-86, s. 42; O.C. 57-2015, s. 20.

4.3.5. Every vehicle transporting explosives must be equipped with 2 portable fire extinguishers graded and classified: 4-A:40-B:C and complying with the standards in section 3.4.4.

The employer must ensure that the driver is capable of using the fire extinguishers.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.5; O.C. 1959-86, s. 42; O.C. 329-94, s. 72; O.C. 57-2015, s. 21

4.3.6. (Revoked.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.6; O.C. 1959-86, s. 42; O.C. 57-2015, s. 22.

4.3.7. Loading and unloading: During the loading or unloading of explosives in a vehicle, the employer must ensure that all the safety measures required to eliminate the risk of accidental blasting are taken. The employer must make sure, in particular, that

(a) the vehicle's engine is not running;

(b) the loading or unloading takes place without interruption and with care, except in the case of bulk explosives.

Once the explosives are unloaded, the employer must make sure that every explosive is stored in a depot, as soon as possible, unless the vehicle constitutes a depot covered by a magazine permit within the meaning of section 38 of the Regulation under the Act respecting explosives (chapter E-22, r. 1).

4.3.8. The driver of a vehicle transporting explosives must no make unnecessary stops. When the vehicle must be parked, the driver must turn the ignition off, set the parking brake and ensure a continuous watch over the vehicle. A vehicle carrying explosives shall not be fueled except where the distance to cover is greater than the range allowed by the fuel tank capacity of the vehicle. In that case, fueling shall nevertheless be done before loading explosives.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.8; O.C. 1959-86, s. 44.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.7; O.C. 1959-86, s. 43; O.C. 57-2015, s. 23.

4.3.9. Only the driver and the persons engaged in the handling of explosives may ride in a vehicle transporting explosives.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.9.

4.3.10. Vehicle equipped with radio transmitter: When detonators are not in their original packing, the employer must make sure that the radio frequency transmitter is not in use unless the detonators are not electrical or are contained in a closed metal case lined with a material not likely to produce sparks.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.10; O.C. 57-2015, s. 24.

4.3.11. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.11; O.C. 57-2015, s. 25.

4.3.12. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.3.12; O.C. 1959-86, s. 45; O.C. 57-2015, s. 25.

4.3.13. Where explosives are transported outside motor roads using a means other than a vehicle referred to in this subdivision, the employer must ensure that

(a) the quantity of explosives transported does not exceed the quantity required for the blasting;

(b) the explosives are contained in a chest that has no material inside that could produce sparks;

(c) detonators are transported separately from the explosives, either in another chest, or in the same chest if it is equipped with a safe partition;

(d) if applicable, the additional quantity of fuel necessary for the transportation is properly separated from the chests containing the detonators and explosives.

O.C. 57-2015, s. 26.

§4.4. Storage of explosives

4.4.1. On a construction site, the employer must make sure that an explosive depot meets the following safety standards:

(a) conform to the standards of the Regulation under the Act respecting explosives (chapter E-22, r. 1);

(b) be laid out so as to comply with the distances established in the standard *Explosives – Quantity Distances*, BNQ 2910-510, or in the table in Schedule 2.3;*

(c) be used exclusively for the storage of explosives or blasting accessories;

(d) be locked;

(e) be under the employer's supervision and responsibility;

(f) be kept clean inside, be coated or covered in such a way that no iron or steel is left uncovered and no particle of rough iron, steel or any similar substance may become loose or come into contact with the explosives contained in the depot;

(g) be of the colour white, aluminum or red with the word EXPLOSIVES written on all visible sides, in contrasting colours at least 150 mm high.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.1; O.C. 1959-86, s. 46; O.C. 57-2015, s. 27.

* NOTE: Until the standard Explosives – Quantity Distances, BNQ 2910-510, is made and published by the Bureau de normalisation du Québec, the distances for the disposition of depots, provided for by the standard in paragraph b of section 4.4.1 of the Safety Code for the construction industry, are those provided for in the Quantity Distance Principles User's Manual published in 1995 by the Explosives Regulatory Division of Natural Resources Canada. (O.C. 57-2015, s. 63.)

4.4.1.1. A construction site chest used as an explosive depot may not contain more than 227 kg of explosives.

The chest may be kept without dike. However, it must be kept at least 15 m from any building, meeting place or road.

If there is more than one chest, each chest must be protected from the others by dikes and comply with the standards prescribed in column 3 of Schedule 2.3.

O.C. 57-2015, s. 27.

4.4.1.2. Where a truck is used to temporarily store the explosives required for a workday, the employer must ensure that the following standards are met:

(a) the quantity of explosives stored may not exceed 800 kg;

(b) the vehicle is covered by a magazine permit within the meaning of section 38 of the Regulation under the Act respecting explosives (chapter E-22, r. 1);

(c) the vehicle is equipped with an automatic fire suppression system, with dry chemicals, complying with AS 5062-2006 standard Fire Protection for Mobile and Transportable Equipment, published by Standards Australia;

(d) during blasting, the truck must be kept in a safe place, outside the blasting area, under the constant supervision of a person holding a general permit issued under the Act respecting explosives (chapter E-22).

O.C. 57-2015, s. 27.
4.4.1.3. During working hours, if it is impossible to conform to the distances provided for in paragraph b of section 4.4.1, the employer may store the explosives required for a workday in a construction site chest or an explosive transportation truck in compliance with the quantity and distance standards provided for in section 4.4.1.1 or 4.4.1.2, as the case may be.

O.C. 57-2015, s. 27.

4.4.2. In off-duty hours, the explosives, detonators and other accessories must be sent back to the supplier or stored in accordance with the Regulation under the Act respecting explosives (chapter E-22, r. 1).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.2; O.C. 1959-86, s. 46.

4.4.3. Dangerous substances: Any flammable substance and any product likely to cause a fire or explosion must be handled and stored in compliance with the measures prescribed in section 3.16.10, away from any explosive depot.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.3; O.C. 57-2015, s. 28.

4.4.4. Every detonator must be stored in a depot separate from the explosive depot. No dike is required around that depot, which must be at least 8 m from any other explosive depot.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.4; O.C. 1959-86, s. 47; O.C. 57-2015, s. 29.

4.4.5. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.5; O.C. 1959-86, s. 48.

4.4.6. *Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.6; O.C. 57-2015, s. 30.

4.4.7. In a depot, explosives and their packages must be stored safely, in particular by

(a) limiting the height of piles so as to prevent explosives from tipping over;

(b) keeping sufficient space between the explosive piles, walls, roof and ventilation openings, so as to maintain proper air circulation;

(c) not opening packages or wooden containers closed with metal ties or strips. The other types of packaging or containers may be opened, one at a time, for inspection purposes or to remove the explosives;

(d) storing only explosive packages or containers that are clean, dry and free of small abrasive materials or any contamination.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.7; O.C. 57-2015, s. 31.

4.4.8. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.8; O.C. 1959-86, s. 48.

4.4.9. At the end of a workday, empty containers that were used to pack explosives must be destroyed according to the manufacturer's instructions or be returned to the supplier so that they may not be used for other purposes.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.4.9; O.C. 1959-86, s. 49; O.C. 57-2015, s. 32.

4.4.10. An explosive depot must be farther from a 44 kV or more overhead electric power line than the distance between the supports of the line located near the depot. Where the distance between the supports of the line is greater than 15 m, the depot must be placed from that line at the greater of the following distances:

(a) 15 m;

(b) the result of the following formula: P / 2 - H (in meters), where "P" is the distance between the supports of the transmission line and "H" is the height of the supports of the transmission line; or

(c) the distance provided for in column 2 of the table in Schedule 2.3.

O.C. 57-2015, s. 32.

§4.5. Drilling

4.5.1. No drilling may be done closer than

(a) 1.5 m from a misfire or blowout. If necessary, holes may be drilled closer but not less than 600 mm on condition that drilling is performed by remote control under supervision and that all precautions are taken to ensure the safety of workers should an explosion occur at the blasting face;

(b) 8 m from any loaded blasthole or explosive loading site.

Despite the foregoing, a blasthole may be drilled closer than 8 m if adaptation to specific job-sites conditions is required, particularly for trench work or work in permafrost zones in unstable conditions. The employer must then ensure that

i. loading and drilling are performed alternately;

ii. the shot-firer supervises and controls the drilling operations;

iii. only cartridge explosives are used. However, if soil degradation does not allow for the insertion of cartridge explosives in the boring hole, the shot-firer may use a blasting agent to load the hole. That method may not be used for more than 3 holes per blasting;

iv. a carpenter's level is used to make sure that blastholes are vertical; v. the minimum distance from any loaded hole is 1.5 m or 20% of the hole depth to a maximum depth of 12 m, using the greater distance between the two;

vi. if the holes have a depth of 6 m or more, the first drilling rod must be replaced by a guide tube or another means providing equivalent precision to avoid the blasting of another loaded hole in the vicinity;

vii. loaded holes must be marked by stakes of a distinct color or carrying a distinct ribbon.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.5.1; O.C. 1959-86, s. 50; O.C 57-2015, s. 33.

4.5.2. The diameter of the explosive used must be smaller than the diameter of the borehole throughout the depth of the latter.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.5.2; O.C. 1959-86, s. 50.

4.5.3. Before being loaded, all holes must be examined and corrected where necessary.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.5.3; O.C. 1959-86, s. 50; O.C. 57-2015, s. 34.

4.5.4. Drilling and loading of explosives may not be carried out simultaneously less than 8 m from one another, or one on top of the other.

O.C. 57-2015, s. 35.

4.5.5. Before drilling any surface of a digging where blasting was performed, all bottoms of blastholes must bemarked in either of the following manners:

(a) by a circle in paint or crayon of a colour contrasting with the soil;

(b) by inserting a stick into the hole.

O.C. 57-2015, s. 35.

4.5.6. It is prohibited to deepen holes remaining intact after blasting.

O.C. 57-2015, s. 35.

§4.6. Loading of blastholes

4.6.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.1; O.C. 57-2015, s. 37.

4.6.1.1. A loading area must be delimited by means of ribbons, trestles or a warning line provided for in section 2.9.4.1. Only persons holding a valid general permit, issued under the Act respecting explosives (chapter E-22), may access the area.

O.C. 57-2015, s. 38.

4.6.2. The primer cartridge of a blasthole shall be prepared only when priming the hole.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.2; O.C. 1959-86, s. 51.

4.6.3. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.3.; O.C. 57-2015, s. 39.

4.6.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.4; O.C. 1959-86, s. 52; O.C. 57-2015, s. 39

4.6.5. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.5; O.C. 1959-86, s. 52 O.C. 57-2015, s. 39.

4.6.6.-4.6.7. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 53.

4.6.8. Tamping rod and punch: For loading, only a tamping rod and punch made of non-ferrous material may be used.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.8; O.C. 1959-86, s. 54.

4.6.9. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.9; O.C. 1959-86, s. 55.

4.6.9.1. Conditions for untamping and repriming a blasthole or misfire: Prior to the untamping or repriming of a blasthole or misfire, the employer must prepare a written procedure that takes into account the types of explosives and the manufacturer's instructions in that regard, as well as environmental conditions.

The employer must also make sure that

(a) the procedure is available on the construction site;

(b) the untamping of the collar is done by the shot-firer who loaded and fired the blasthole, unless it is impossible for him to do so;

(c) during all untamping, repriming and firing operations, all persons except the shot-firer are outside the blasting area;

(d) the constituents of the material used for untamping the blasthole and inserted in it is made of non-ferrous materials.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.9.1; O.C. 1959-86, s. 56; O.C. 57-2015, s. 40.

4.6.10. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.10; O.C. 1959-86, s. 57.

4.6.11. At the first signs of a thunderstorm, the employer must stop all loading and detonator connecting operations. The employer must evacuate the blasting area, prohibit access thereto and supervise the situation from a distance.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.11. O.C. 57-2015, s. 41.

4.6.12. Electric blasting operations:

(a) For electric blasting operations, the entire length of the lead wires must be made of 2 insulated conductors that do not come in contact with the ground or other conductor and must reach the immediate vicinity of the firing point. These conductors must not be intertwined with conductors intended for any other purpose whatsoever, nor run in the same tube with them, nor have the possibility of coming in contact with them;

(b) To avoid stray electric currents or current leaks to ground, bare connections between the lead wires and the cap leg wires or between the cap leg wires must not come in contact with the ground or equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.12; O.C. 1959-86, s. 58.

4.6.13. During the final connection of lead wires and the various electric blasting caps, the entire firing circuit must be checked using a blasting ohmmeter.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.13; O.C. 1959-86, s. 58; O.C. 57-2015, s. 42.

4.6.14. Blasting machine:

(1) The blasting machine shall be stored in a cool, dry place.

(2) It shall be kept in good operating condition and shall be tested regularly.

(3) The capacity of the blasting machine shall be clearly marked on each machine and such capacity shall never be exceeded. Only the shot-firer shall have access to the operating components of such apparatus.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.14.

4.6.15. Detonating fuse: When the lead wires are composed of detonating fuses, the employer must ensure that the following safety measures are complied with:

(a) spliced pieces are not used in a single blasthole;

(b) after priming, the down line is cut from the reel and a sufficient length, approximately 200 mm, protrudes from the blasthole to prevent a possible settlement of the load prior to make final connections;

(c) main fuses are connected to detonating fuses at right angles;

(d) when priming a detonating fuse with a detonator, the end with the explosive charge is set in the same direction as the expected shock wave;

(e) no detonating relay is placed in a blasthole;

(f) the shot-firer has visually checked all the connections;

(g) the firing point of the detonating fuse must be located outside the surface covered by the blasting mats;

(*h*) the detonator used for starting the detonating fuse must be set in place only once the covering operations have been completed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.15; O.C. 1959-86, s. 59 O.C. 57-2015, s. 43.

4.6.16. Blending of ANFO: The blending of ammonium nitrate and fuel oil or of other nitrocarbonitrates must not be carried out unless a licence has been granted or written permission given under the Explosives Act (R.S.C. 1985, c. E-17).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.16; O.C. 1959-86, s. 60.

4.6.17. When explosives in bulk are loaded, a semiconductor loading hose must be used and the loading equipment must be grounded according to the manufacturer's instructions.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.17; O.C. 1959-86, s. 60, O.C. 57-2015, s. 44.

4.6.18. Unloading and refiring: It is prohibited to unload or clean out a blasthole or a misfire.

If the starting operation or refiring is impracticable, the explosives must be pulled out in accordance with a procedure prepared in writing by an engineer, taking into account the types of explosives and the manufacturer's instructions in that regard, as well as environmental conditions.

The procedure must be available on the work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.18; O.C. 1959-86, s. 60 O.C. 57-2015, s. 45.

4.6.19. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.6.19; O.C. 1959-86, s. 60 O.C. 57-2015, s. 46.

§4.7. Firing

4.7.1. All loaded holes must be primed and fired in the same round. When firing cannot be done at the end of the loading, the blasting area must remain under supervision, be evacuated, and no access shall be permitted until after the firing.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.1; O.C. 1959-86, s. 61 O.C. 57-2015, s. 47.

4.7.2. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.2.; O.C. 57-2015, s. 48.

4.7.3. Lead wires must be connected to the blasting machine after the signal indicating that blasting is imminent. Lead wires must always be disconnected from the blasting machine immediately after the firing or after the attempted blast. Both ends of the lead wires must be short-circuited and insulated to guard against stray currents.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.3; O.C. 1959-86, s. 62.

4.7.4. Firing may be done from a power line or from a portable generator, provided that:

(a) voltage does not exceed 220 V;

(b) the blasting switch be designed so that:

i. the weighted handle of the blasting switch automatically gravitates to the "off" position and short-circuits the lead wires; and

ii. the door of the box housing the blasting switch is provided with a device that prevents it from being locked or closed unless the blasting switch is in the "off" position. The door must be kept locked except when firing and only the shot-firer shall have the key;

(c) a fused switch is installed between the power source and the blasting switch;

(d) the fused switch and the blasting switch are situated in the proper place for blasting and at a distance of 1.5 to 1.8 m from each other as a precaution against lightning. Underground they will be placed on either side of a tunnel or gallery.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.4; O.C. 1959-86, s. 63.

4.7.5. When blasting is done in the vicinity of a structure such as a building, railway or road, the employer must limit the quantity of explosives so that the vibrations caused by the blasting do not damage those structures.

To that end, the employer must comply with the most stringent standards between those provided for in specifications designed for that purpose by a public authority and those provided for in blasting specifications signed and sealed by an engineer. Failing such specifications, the employer must comply with one of the standards provided for in Schedule 2.6.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.5; O.C. 1959-86, s. 64; O.C. 57-2015, s. 50.

4.7.5.1. During blasting, projections must stay with in the blasting area. To that end, the employer must take appropriate measures to reduce and control projections, in particular by using blasting mats.

When blasting mats are used, they must be deposited, not slid, onto blastholes loaded with explosives.

O.C. 57-2015, s. 50.

4.7.6. The firing procedures are as follows:

(*a*) before proceeding with the firing, the shot-firer must ensure with the employer that all persons have taken shelter;

(b) sound signals must be emitted with a siren of at least 120 dB:

i. immediately before blasting, 12 short horn signals at one-second intervals;

ii. 30 seconds must elapse between the last warning signal and the time of firing;

iii. after blasting, once the blasting area is safe, one continuous 15-second horn signal must announce that work may be resumed in the area;

(c) the employer must make sure that workers take shelter outside the blasting area before the first signal and that they remain there until the 15-second signal is sounded;

(d) a code of sound signals reserved for blasting operations must be written in coloured letters 150 mm high, against a contrasting background, on a board 1.2 m high by 2.4 m wide, placed at all points of access to the work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.6; O.C. 1959-86, s. 64 O.C. 57-2015, s. 51.

4.7.7.-4.7.8. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 65.

4.7.9. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.9; O.C. 57-2015, s. 52.

4.7.10. Blasting logbook: The blasting logbook must at least contain the information provided for in Schedule 2.2 and it must be maintained and signed by the shot-firer. The employer must keep it for 3 years and make it available at all times on the work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.7.10; O.C. 1959-86, s. 66; O.C. 57-2015, s. 53.

§4.8. Work after blasting

4.8.1. After blasting, the shot-firer must be the first to enter the blasting area to make sure it is safe. For that purpose, the shot-firer must

(a) wait for the smoke to dissipate;

(b) make sure, using a device to measure the concentration of carbon monoxide, that the carbon monoxide concentration is below the exposure limit values indicated in Schedule I to the Regulation respecting occupational health and safety;

(c) examine the work site;

(d) look for possible misfires, blowouts and bootlegs;

(e) mark those found.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.8.1; O.C. 1959-86, s. 67 O.C. 57-2015, s. 55.

4.8.2. When the shot-firer considers that the blasting area is safe, the shot-firer informs the employer that the employer may

(a) sound the 15-second signal;

(b) remove the blasting mats as soon as possible after the end of blasting;

(c) excavate blasting debris.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.8.2; O.C. 1959-86, s. 68; O.C. 57-2015, s. 56.

4.8.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.8.3; O.C. 1959-86, s. 69; O.C. 57-2015, s. 57.

4.8.4. When it is found that a defect in the electrical circuit has prevented the charges from detonating at time of firing, the shot-firer may immediately check over the circuit after having made sure that:

(a) the ends of the lead wires are disconnected from the power supply and short-circuited;

(b) the blasting machine is locked or under a worker's surveillance; and

(c) if used, the blasting switch is locked and in open circuit.

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.8.4.

§4.9. (Revoked).

4.9.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.9.1; O.C. 1959-86, s. 70; O.C. 57-2015, s. 58.

4.9.2. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.9.2; O.C. 1959-86, s. 71; O.C. 57-2015, s. 58.

4.9.3. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 4.9.3; O.C. 1959-86, s. 71; O.C. 57-2015, s. 58.

DIVISION V WORK IN THE PROXIMITY OF ELECTRICAL LINES

§5.1. Scope

5.1.1. This Division applies to any construction work carried out near an aerial electrical line.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.1.1.

5.1.2. However, this Division does not apply to:

(a) a neutral conductor;

(b) an insulated cable less than 750 V of the duplex, triplex or quadruplex assembly type;

(c) a consumer or distributor branching less than 750 V;

(d) the electrical installation of the consumer;

(e) the construction, repair or maintenance of an electrical line carried out by a worker of an electrical power company or by an employer authorized by it;

(f) the construction, repair or maintenance of a communications network performed on a structure supporting an electrical line by an employer authorized by an electrical power company; or

(g) work performed near an electrical line 750 V or less, provided that there is insulation between the worker and the non-insulated live parts.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.1.2.

§5.2. Interdictions

5.2.1. The employer shall ensure that no one performs work liable to bring any part, load, scaffolding, machine component or person closer than the minimum approach distance specified in the following table:

Tension between phases (volts)	Minimum approach distance (metres)
Less than 125 000	3
125 000 to 250 000	5
250 000 to 550 000	8
More than 550 000	12

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.2.1; O.C. 35-2001, s. 22.

5.2.2. The employer who wishes to carry out work liable to bring any part, load, scaffolding, machine component or person closer to a power line than the minimum approach distance specified in section 5.2.1 may proceed with such work provided that

(a) that power line has been turned off. The employer shall ensure that no one runs any risk of being electrocuted before turning the power on again;

(b) the employer has come to an agreement with the electrical power company as to safety measures to be adopted. Before the work begins, the employer shall transmit a copy of such agreement as well as the work plan to the Commission. Such measures shall be carried out before the work begins and shall be maintained throughout the course of the work; or

(c) the employer uses extensible construction equipment, such as a backhoe, a power shovel, a crane or a dump truck, and complies with the following conditions:

i. the extensible construction equipment is equipped with a device having a first function of warning the operator or stopping the equipment from operating so that the minimum approach distance specified in section 5.2.1 is respected, and a second function of stopping the equipment from operating should the device fail to perform its first function. A written declaration signed by an engineer, certifying that the extensible equipment performs those functions and that it neither damages the equipment from operating, shall be obtained by the employer. If the device fails to operate partially or completely, or is inoperative, the employer shall cease to use the extensible construction equipment and shall obtain a new written declaration signed by an engineer before re-using the equipment;

ii. the operator of the extensible construction equipment equipped with the device referred to in subparagraph *i* must have received the manufacturer's training on the proper use of the device.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.2.2; O.C. 1959-86, s. 72; O.C. 53-90, s. 10; O.C. 35-2001, s. 23; O.C. 483-2021, s. 8.

5.2.3. The employer who carries out work less than 30 m away from an electrical line with a voltage higher than 250,000 V shall ensure that the following requirements are respected:

(a) refueling shall be done outside the area;

(b) construction equipment on tires shall be equipped with an electrostatic link between the metallic part and the ground; and

(c) during the installation or manipulation of a metal duct, fence or above ground structure, it shall be grounded every 30 m.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.2.3.

5.2.4. Before carrying out work near an electrical line held at each supporting point otherwise than by one wooden post, the employer shall obtain in writing the line voltage from the electrical power company.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.2.4.

§5.3. Warning sign

5.3.1. The employer shall ensure that the owner or lessee of any piece of machinery used to lift a load and capable of vertical, lateral or rotational motion, posts on such machinery, in a conspicuous place so as to be seen by the operator, a warning sign reading: DANGER —

N'APPROCHEZ PAS DES LIGNES ELECTRIQUES, in letters at least 12 mm in height.

R.R.Q., 1981, c. S-2.1, r. 6, s. 5.3.1.

DIVISION VI SHORING OF CONCRETE FORMWORK

§6.1. Shoring drawing

6.1.1. A copy of the drawing mentioned in paragraph b of subsection 2 of section 2.4.1 shall be kept on the construction site for the entire duration of the work.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.1.1.

6.1.2. If, for a construction, there is repetition of the form of the structure and it is not necessary to change the shoring drawing, the same drawing may be used for the following stages on the same site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.1.2.

6.1.3. The shoring drawing shall specifically mention all the information respecting the installation of the shoring; the drawing shall contain the following information in particular:

- (1) spacing of load bearing elements;
- (2) bracing;
- (3) size of the parts;
- (4) design loads;
- (5) resistance of the materials;
- (6) bearing surface;
- (7) method for pouring the concrete; and

(8) any other information taken into account by the engineer who signed the drawing.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.1.3; O.C. 1413-98, s. 26.

§6.2. Concreting work

6.2.1. Before commencing concreting work, the employer shall:

(a) obtain from an engineer:

i. a signed and sealed declaration attesting that the work complies with all the points in the shoring drawing filed; and

ii. the authorization to proceed with the concreting work;

(b) transmit immediately a copy of the declaration to the Commission; and

(c) keep a copy of such declaration at the work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.2.1; O.C. 1959-86, s. 73.

§6.3. Exception

6.3.1. In the case of the construction of slabs and beams that are uniform and horizontal, Subdivisions 6.1 and 6.2 do not apply when the 4 following conditions are respected:

(a) the thickness of the concrete slab does not exceed 150 mm and the concrete beams are not higher than 300 mm including the thickness of the slab;

(b) the distance between the underside of the poured concrete and the bearing surface of the posts does not exceed 4.9 m;

(c) Subdivisions 6.4 and 6.9 are complied with;

(d) prior notice is given to the Commission.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.3.1; O.C. 1959-86, s. 74.

§6.4. Materials

6.4.1. All formwork and shoring parts shall be designed to support:

(a) the dead load of the formwork;

(b) the dead load of the concrete and the embedded materials;

(c) a minimum vertical live load evenly distributed of $2,400 \text{ N/m}^2$ or more to the satisfaction of the engineer; and

(d) a minimum horizontal live load evenly distributed of $1,500 \text{ N/m}^2$ of vertical surface or more, to the satisfaction of the engineer.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.4.1.

6.4.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.4.2; O.C. 393-2011, s. 19.

§6.5. Bearing surface

6.5.1. The bearing surface shall be level and firm.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.1.

6.5.2. The size of the sills shall be determined by taking into account the bearing capacity of the soil and the imposed loads.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.2.

6.5.3. A sill shall be installed between the posts and the soil surface. To ensure continuity of the sills, a piece of the same section as the sill and of sufficient length to support at least 2 posts shall be used over the joint. If it is impossible to ensure such continuity, the sill shall support at least 3 posts and shall be at least equal in length to the number of posts multiplied by their spacing.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.3.

6.5.4. If the shores rest on concrete slabs, the loads due to the shoring of the upper floors shall not exceed the initial design loads. For slabs:

(a) 100 mm thick or less, the sill shall support a minimum of 3 posts and shall be at least equal in length to the number of posts multiplied by their spacing;

(b) more than 100 mm thick, wooden pieces covering more than the entire telescoping jack flange area shall be used between the slab and the telescopic jacks.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.4.

6.5.5. The bearing surface of shores shall be so designed as not to damage the concrete.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.5.

6.5.6. During the freezing and thawing period of the soil, the ice and snow shall be removed and the frozen soil shall be covered with a layer of gravel or sand dust before the sills are installed. The soil shall be:

(a) protected against a possible thaw; or

(b) thawed completely before undertaking the works.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.6.

6.5.7. The construction of embankments or earth fills on which sills rest shall be carried out by successive layers and each layer shall be compacted mechanically. A granulated cohesive material shall be used to form a compact, resistant and waterproof base.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.5.7.

§6.6. Support parts

6.6.1. Telescopic steel jacks, timber shores and scaffold shoring shall comply with the requirements of Subdivision 2.13.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.6.1.

6.6.2. Posts, joists and stringers:

(1) the minimum length of the protrusion of a joist supported on a stringer shall not be less than 300 mm, unless the joist is affixed by ties, in which case the joist shall be supported at least on the full width of the stringer.

(1.1) on a same linear section of the perimeter of the shoring of the framework of a slab, the length of the protrusion of the joists and stringers shall be equal.

(2) Stringers shall be long enough to be supported by at least 3 posts.

(3) The continuity of stringers shall be ensured in one of the following ways:

(a) by a piece of wood 50 mm in nominal thickness and as wide as that of the stringer or the head of the telescopic jack, whichever is the smallest, and long enough to ensure solid attachment to the 2 stringers when the posts are placed immediately below the joint of the stringer and when:

i. telescopic jacks with flanges of 100 \times 100 mm or 100 \times 150 mm are used; or

ii. stringers of 100 mm wide are used;

(b) by a part of the same section as that of the stringers nailed to the latter and of sufficient length to be supported by at least 2 posts when the posts are placed on each side of the joint and not at the extremities of the stringers;

(c) by the flange of the telescopic jack if the dimensions are sufficient to do so, that is when:

i. telescopic jacks with flanges of 100 \times 200 mm are used;

ii. stringers of 100×100 mm are used; and

iii. the posts are placed directly beneath the joints of the stringers.

(4) Posts shall be firmly fixed, supported, and secured at each extremity.

(5) Formworks for reinforced concrete beams shall be supported by at least 2 rows of posts.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.6.2; O.C. 1413-98, s. 27.

§6.7. Bracing

6.7.1. Horizontal bracing:

(1) A horizontal bracing shall be placed horizontally in 2 directions at right angles to each post of any shoring when:

(a) this post is 1.8 m high or more and is placed on the soil;

(b) this post is 3.5 m high or more and rests on a concrete slab.

(2) The horizontal bracing shall be placed:

(a) as close as possible to the mid-height of the posts unless another point of buckling is indicated by the calculations; and

(b) when posts are 5.5 m long or more, at heights not exceeding 2.7 m between this bracing and:

i. the bottom or the top of the post; or

ii. another bracing on the same post.

(3) The bracing required in subsection 1 may be omitted if the engineer who drew up the calculations of the concrete shoring plan indicates on the plan he submits that it is not necessary.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.7.1.

6.7.2. Diagonal bracing: At every 4 rows of telescopic jacks, a diagonal bracing placed at a 45° angle shall be installed on both vertical planes perpendicular to each other. This bracing shall alternate from top to bottom and from bottom to top.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.7.2.

6.7.3. Especially solid bracing shall be provided for scaffold shoring and for structures when the imposed loads are not axial.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.7.3.

6.7.4. Only one telescopic jack shall be used for each level of shoring.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.7.4.

§6.8. Shoring of multistory structures

6.8.1. Concrete slabs resting on the ground or on foundation piles and which cannot be shored shall be designed to support the loads due to the shoring of the upper floors.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.8.1.

§6.9. Shoring, dismantling and deforming

6.9.1. The following provisions shall be respected during deforming:

(a) during construction, shoring shall remain in place for 21 days, unless an engineer delivers an attestation establishing that the cement has attained sufficient strength to support its own mass and the loads that it may have to support;

(b) forms shall be removed progressively by section:

i. by taking care to place one's feet on a steady support;

ii. by taking care not to damage the concrete elements; and

iii. by proceeding continuously ahead and foreseeing a free exit path in case of falling objects;

(c) in the case of building framework, the columns shall be deformed first; and

(d) the forms under the slabs and those on the side of the beams and arches shall be removed before the shoring of the beams and arches.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.9.1.

6.9.2. Telescopic jacks shall be removed by means other than a sledgehammer or other heavy objects.

R.R.Q., 1981, c. S-2.1, r. 6, s. 6.9.2.

DIVISION VII PORTABLE TOOLS

R.R.Q., 1981, c. S-2.1, r. 6, sec. VII; O.C. 329-94, s. 7; O.C. 483-2021, s. 9.

§7.0. General

7.0.1. For the purposes of this Division, "portable tool" means a tool whose weight is supported by a person during its use.

O.C. 483-2021, s. 9.

7.0.2. A portable tool shall not compromise the safety of workers. To that end, it shall

(1) be maintained in good working condition;

(2) be verified by a qualified person, where it is powered by a source of energy other than manual, before its initial use on the site and daily thereafter when it is used; and

(3) be maintained in accordance with the manufacturer's instructions.

O.C. 483-2021, s. 9.

7.0.3. A portable tool shall be used in accordance with the manufacturer's instructions.

It shall not be used if weather conditions may make its use dangerous.

O.C. 483-2021, s. 9.

7.0.4. Unless the manufacturer's instructions so allow, a portable tool shall not be in operation while it is being recharged, repaired or adjusted, maintained or cleaned.

In addition, the engine shall be cooled before refuelling and the portable tool shall not be started less than 3 m from the place where it was refuelled.

O.C. 483-2021, s. 9.

7.0.5. Subject to section 7.1.1.3, a portable tool shall not be modified unless the manufacturer or an engineer certifies in writing that the modification does not compromise its safety or offers the same safety as the original tool.

O.C. 483-2021, s. 9.

7.0.6. A portable tool powered by an internal combustion engine shall be used in accordance with section 3.10.17.

O.C. 483-2021, s. 9.

7.0.7. A portable tool powered by an electrical source shall be used in accordance with subdivision 2.11.

O.C. 483-2021, s. 9.

7.0.8. Subject to a special provision in this Division, the personal protective equipment provided for in subdivision 2.10 for the protection of workers against the risks of injury caused by a portable tool shall be worn during the use of the tool.

O.C. 483-2021, s. 9.

§7.1. Special provisions relating to certain portable tools

R.R.Q., 1981, c. S-2.1, r. 6, sec. VII, ss. 7.1; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

§7.1.1. Explosive actuated tool

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.1.1; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.1.1.1. Only a low velocity explosive actuated tool may be used.

O.C. 483-2021, s. 9.

7.1.1.2. A low velocity explosive actuated tool shall (1) be unloaded when not in use; and

(2) never be left without supervision when it is loaded. $\overline{O.C. 483-2021, s. 9}$.

7.1.1.3. Only the manufacturer may modify a low velocity explosive actuated tool.

O.C. 483-2021, s. 9.

7.1.1.4. A low velocity explosive actuated tool may only be operated by a worker having received training and holding the certificate of low velocity explosive actuated tool operator, as provided for in Schedule 8.

O.C. 483-2021, s. 9.

7.1.1.5. No work may be performed by a low velocity explosive actuated tool operator who is less than 18 years of age.

O.C. 483-2021, s. 9.

7.1.1.6. An operator may not use a low velocity explosive actuated tool to drive

(1) fasteners into

(a) curved or rounded objects, except if the tool is equipped with a protective device designed for such work;

(b) plaster tiles, hollow bricks or slates;

(c) cast iron, marble, granite, glazed linings and other hard and brittle materials;

(d) steel or alloys that are harder than the fastener used;

(e) hard materials in which holes have already been made, except if the tool is equipped with a device that is capable of holding back the fasteners;

(f) corner bricks or vertical mortar joints; and

(g) steel where

i. the steel is less than 4.83 mm thick;

ii. the point of entry of the fasteners is less than 50 mm from a weld;

iii. the point of entry of the fasteners is less than 13 mm from an edge;

(2) fasteners with a shaft diameter equal to or less than 4.83 mm into concrete where

(*a*) the concrete is less than 65 mm thick or equal to 3 times the depth of penetration of the shaft of the fasteners;

(b) the point of entry of the fasteners is less that 50 mm from an unsupported edge; and

(c) the point of entry of the fasteners is less that 75 mm from another fastener that is broken.

O.C. 483-2021, s. 9.

7.1.1.7. Before firing, the operator shall ensure

(1) that the low velocity explosive actuated tool

(a) is placed in a stable firing position; and

(b) is held so that the barrel of the tool is perpendicular to the firing surface; and

(2) that there is no other person within firing range. $\overline{O.C. 483-2021, s. 9}$.

7.1.1.8. Firing incident: Where a firing incident or a misfire occurs, the low velocity explosive actuated tool shall be held in its firing position for at least 15 seconds; the tool shall then be unloaded. In such a case, the barrel of the tool shall

(1) not be pointed toward the operator or any other person;

(2) be held pointing obliquely toward the ground; and

(3) be held as far as possible from the body of the operator.

O.C. 483-2021, s. 9.

7.1.1.9. The employer shall prohibit the use of a low velocity explosive actuated tool in shops or any other area where the concentration of inflammable vapours, gases or dust has reached the lower explosive limit.

O.C. 483-2021, s. 9.

7.1.1.10. The employer shall ensure

(1) that a low velocity explosive actuated tool is

(a) checked before its first use each day; and

(b) regularly inspected to detect worn or damaged parts, in accordance with the manufacturer's recommendations;

(2) that all parts of the low velocity explosive actuated tool have been cleaned after its use; and

(3) that the safety devices on a low velocity explosive actuated tool are in proper working order.

O.C. 483-2021, s. 9.

7.1.1.11. Only spare parts recommended by the manufacturer shall be used.

O.C. 483-2021, s. 9.

7.1.1.12. No low velocity explosive actuated tool may be used where any of its parts or accessories is defective.

O.C. 483-2021, s. 9.

7.1.1.13. When it is not in use, a low velocity explosive actuated tool shall be placed in a case designed for that purpose. The case shall contain

(1) a copy of the manufacturer's instructions for the use and maintenance of the tool;

(2) all the accessories and implements necessary for the maintenance of the tool at the work site; and

(3) a logbook recording the date of each inspection provided for in subparagraph *b* of paragraph 1 of section 7.1.1.10 as well as the date and type of each repair made. $\overline{0.C.483-2021.s.9}$.

7.1.1.14. The case provided for in section 7.1.1.13 and the boxes containing the fasteners and cartridges shall be put in a place that

(1) is kept locked; and

(2) is inaccessible to unauthorized persons.

O.C. 483-2021, s. 9.

7.1.1.15. The operator shall

(1) pick up, as work progresses, the cartridge cases that exploded;

(2) store unused cartridges in accordance with section 7.1.1.14; and

(3) dispose of used cartridges that did not explode in accordance with the manufacturer's instructions.

O.C. 483-2021, s. 9.

7.1.1.16. The following notices shall be affixed permanently and shall be clearly legible:

(1) on each low velocity explosive actuated tool:

(a) the manufacturer's name or trademark;

(b) the type and model of the tool;

(c) the strength of the maximum charge permitted by the manufacturer's specifications;

(2) on the accessories, the manufacturer's name or trademark;

(3) on each box containing fasteners:

(a) the manufacturer's name or trademark;

(b) the nominal dimensions of the fasteners;

(4) on each box containing explosive charges:

(a) the manufacturer's name or trademark;

(b) the place where it was manufactured;

(c) the strength of the explosive charge of the cartridges.

O.C. 483-2021, s. 9.

§7.1.2 Nailing gun

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.1.2; O.C. 329-94, s. 73.

7.1.2.1. Definitions: For the purposes of this subdivision,

"nailing gun" means a device held by hand by a single operator and in which energy is transmitted in a linear manner to a steel nail charged into the device in order to drive the nail. The energy required for driving comes in particular from compressed air, combustion gas or an electrical load, but not from propellant powder;

"dual-action contact-trip command" means a command method in which the trigger and the nose contact element must be interlocked so that only one drive operation is carried out by pressing the trigger while the nose contact element is pressed on the material. To repeat the operation, the trigger and the nose contact element must first return to their idle position;

"trigger" means a finger-actuated part that controls the arrival of energy to the driving mechanism of a nailing gun;

"nose contact element" means a mechanism at the end of a nailing gun that, for as long as it is not touching a material, prevents the firing of a nail;

"framing work" means construction work related to the structure of walls, floors and roofs. Finishing work and work for covering roofs with shingles are excluded. $\overline{\text{O.C.} 483-2021, \text{s.} 9}$.

7.1.2.2. A nailing gun used for framing work shall

(1) be equipped with a trigger and a nose contact element; and

 $\underbrace{(2) \text{ operate by dual-action contact-trip command.}}_{O.C. 483-2021, \text{ s. 9.}}$

7.1.2.3. A nailing gun shall be used

(1) in a stable position;

(2) while wearing the protective glasses described in section 2.10.5; and

(3) without pointing the nailing gun at the operator or any other person.

O.C. 483-2021, s. 9.

7.1.2.4. A nailing gun shall be disconnected from its energy source before its maintenance or unblocking.

§7.1.3 Saws

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.1.3; O.C. 1959-86, s. 75; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

§7.1.3.1. Circular saws

7.1.3.1.1. A circular saw, except a cut-off machine, shall comply with paragraph 2 of section 3.10.15.

O.C. 483-2021, s. 9.

§7.1.3.2. Chainsaws

7.1.3.2.1. A chainsaw shall comply with CSA Standard Z62.1-15, Chainsaws, for Classes 1A and 2A. O.C. 483-2021, s. 9.

7.1.3.2.2. A chainsaw shall not be used to cut materials other than wood, unless such use is specified by the manufacturer and the required recommended modifications have been made, if applicable.

It shall not be used inside a closed building if it has an internal combustion engine.

O.C. 483-2021, s. 9.

7.1.3.2.3. The user of a chainsaw shall wear protective footwear from among the following:

(1) footwear referred to in section 2.10.6

(a) that complies with the recommendations for the user of a chainsaw; or

(b) that has protective gaiters and that complies with Part 9 of EN Standard 381-9, Personal protective equipment for users of hand chainsaw;

(2) footwear that complies with ISO Standard 17249, Safety footwear with resistance to chainsaw cutting;

(3) footwear that complies with ISO Standard 20345, Personal protective equipment — Safety footwear, that has protective gaiters and that complies with Part 9 of EN Standard 3819, Personal protective equipment for users of hand chainsaw.

O.C. 483-2021, s. 9.

7.1.3.2.4. The user of a chainsaw shall wear pants or chaps complying with Class A, C or D of ASTM Standard F 3325-18, Standard Specification for Leg Protective Devices for Chainsaw Users.

O.C. 483-2021, s. 9.

7.1.3.2.5. The user of a chainsaw shall wear gloves that allow a grip on the chainsaw's handles.

O.C. 483-2021, s. 9.

7.1.3.2.6. The user of a chainsaw shall

(1) start the chainsaw by firmly maintaining the front handle with the left hand and the rear handle between the knees or on the ground by standing with the right foot in the rear handle;

(2) hold the chainsaw with both hands and with feet solidly set during use; and

(3) apply the chain brake during a displacement when the engine is on.

During its use, a chainsaw shall not be held higher than the shoulders.

O.C. 483-2021, s. 9.

7.1.4.-7.1.6.

(Replaced).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

§7.2. Operator of a low velocity explosive actuated tool

(Replaced).

R.R.Q., 1981, c. S-2.1, r. 6, sec. VII, ss. 7.2; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.2.1. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.2.1; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.2.2. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.2.2; O.C. 1959-86, s. 76; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.2.3.-7.2.5. (Replaced).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 329-94, s. 73.

§7.3. Precautions

(Replaced).

R.R.Q., 1981, c. S-2.1, r. 6, sec. VII, ss. 7.3; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.3.1.-7.3.3. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.3.4. (*Replaced*).

O.C. 329-94, s. 73; O.C. 483-2021,s. 9.

§7.4. Maintenance and repair

(Replaced).

R.R.Q., 1981, c. S-2.1, r. 6, sec. VII, ss. 7.4; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.4.1. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.4.1; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.4.2. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 7.4.2; O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.4.3.

(Replaced).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

§7.5. Storage

(Replaced).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.5.1. (*Replaced*).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.5.2. (Replaced).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

§7.6. Notices

(Replaced).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

7.6.1. (*Replaced*).

O.C. 329-94, s. 73; O.C. 483-2021, s. 9.

DIVISION VIII UNDERGROUND WORK SITES

§8.1. Fire prevention and protection

8.1.1. A temporary building built on the surface less than 12 m away from an opening giving access to an underground work site or erected in the underground work site shall:

(1) be of non-combustible construction; or

(2) be protected against fire in the following way:

(a) the building shall be equipped, inside and outside, with portable fire extinguishers complying with section 3.4.4 of this Code;

(b) the construction site shall be equipped with:

i. a water supply system not less than 300 litres per minute during at least 30 minutes under a minimum residual pressure of 80 kPa; and

ii. flexible rubber or plastic hoses not more than 15 m in length and not less than 19 mm inside diameter, equipped with nozzles. Moreover, hoses shall be so located that every part of the building can be sprayed with a stream of water and sprayed areas must be within 6 m from the nozzle of the hose supplying the spray.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.1; O.C. 329-94, s. 74.

8.1.2. In addition to complying with the provisions of NFPA 30 Flammable and Combustible Liquids Code, the storage of flammable liquids in an underground work site must

(a) be in closed reservoirs of 200 litres with a tap or in safety tanks holding not more than 20 litres equipped with a neck which closes by means of a spring cover and so designed as to allow the safe release of interior pressure where they are subjected to fire; and

(b) be limited to the amount needed for 1 day's work. R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.2; O.C. 393-2011, s. 20.

8.1.3. Waste timber, scrap paper, rags and other combustible waste materials shall daily:

(a) be taken from the underground work site; and

(b) be cleared away from around the temporary surface buildings.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.3.

8.1.4. No gasoline or liquid petroleum gases shall be lowered into, stored or used in an underground site. However, the propane kept in cylinders may be used for welding.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.4.

8.1.5. Oil and grease shall be stored underground in hermetically sealed containers:

(a) placed at safe distances from explosive depots, electric stations and shaft stations; and

(b) in quantities not exceeding those necessary for 1 week.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.5.

8.1.6. Every self-propelled vehicle shall be equipped with a portable fire estinguisher complying with section 3.4.4 of this Code.

The driver of the vehicle shall have the necessary knowledge concerning the type of fire extinguisher with which the vehicle is equipped and how it is operated.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.6; O.C. 329-94, s. 75.

8.1.7. Temporary building: A temporary building for an underground work site shall not be installed in traffic areas.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.1.7.

§8.2. Ground stability

8.2.1. The employer shall ensure that experienced workers scale or shore up the earth liable of coming away from a wall.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.1.

8.2.2. The employer shall ensure that experienced workers continually check and examine the vault, heading-face, partition walls of each work area and travelways for workers or equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.2.

8.2.3. Overhanging masses and materials liable to fall off from the banks during the work shall be cut down or shored up immediately.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.3.

8.2.4. Scaling bars shall be:

(a) of an appropriate length so as to work safely and efficiently;

(b) pointed at one end; and

(c) equipped with a protector for hands.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.4.

8.2.5. Workers who check or scale earth liable of falling shall:

(a) proceed by starting from solid ground;

(b) place their feet in a firm stance; and

(c) have enough free space to allow get away.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.5.

8.2.6. Equipment placed at the disposal of workers for scaling shall be designed so as not to endanger them.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.6.

8.2.7. Pieces of timber or steel, shotcrete or other supports necessary for shoring shall be put into place quickly and solidly.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.7.

8.2.8. Damaged or dislocated supports shall be repaired or replaced without delay and new supports shall be put into place if possible, before the damaged supports are removed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.8.

8.2.9. Shoring, if necessary, shall follow shaft-sinking in order to prevent the fall of rocks from the partition walls. R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.9.

8.2.10. During rock bolting, the required bolts shall be:

(a) put into place as soon as a zone has been exposed; and

(b) equipped with a stress distribution plate on the rock. R.R.Q., 1981, c. S-2.1, r. 6, s. 8.2.10.

§8.3. Ventilation and air quality

8.3.1. The concentration of impurities in the air shall be kept at less than the permissible values indicated in Schedule I to the Regulation respecting occupational health and safety (chapter S-2.1, r. 13). The underground site shall be supplied with fresh air. The minimum volume of air must be equivalent to the greatest of the following requirements:

(a) 5.5 m^3 per minute of fresh air for each worker underground;

(b) 15 m³ of fresh air per minute for each square metre of section for tunnels; or

(c) where there is mobile equipment driven by a diesel engine:

i. certified by the National Institute for Occupational Safety and Health (NIOSH), the supply of fresh air shall be based on the values given in Schedules 24 and 31 of that Institute;

ii. certified by the Department of Energy, Mines and Resources of Canada, the supply of fresh air shall be that specified at the time of certification of the equipment; or

iii. not certified, the fresh air supply shall correspond to a minimum rate of 5.5 m^3 per minute per kilowatt measured at the motor shaft and the total quantity of fresh air necessary when several of these machines are in operation at the same time shall be:

(a) 100% of the given supply for the most stringent unit from the point of view of ventilation;

(b) 75% of the given supply for the second unit; and

(c) 50% of the given supply for each additional unit;

(d) the fresh air supply, as specified in paragraphs a, b and c shall be increased where applicable until the concentration of impurities in the air is reduced to the permissible value indicated in Schedule I to the Regulation respecting occupational health and safety.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.1; O.C. 885-2001, s. 377.

8.3.2. The test samples of the fresh air supply for concentration of carbon monoxide, nitrogen dioxide or other noxious gases in an underground work site shall be performed as often as necessary, but at least 2 times each day, and one of these, one hour after the beginning of operations. All test results shall be compiled and made available for the inspector's examination.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.2.

8.3.3. Ventilators shall:

(a) operate at all times when workers are in the underground work site and if ventilators are stopped while the underground work site is unoccupied, they shall be put into operation for a sufficient period of time prior to the beginning of operations in order to ensure proper ventilation;

(b) be placed in constructions of fireproof materials; and

(c) be equipped with a surface remote control device. R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.3.

8.3.4. The employer shall ensure that the fresh air supply for the ventilation system is without impurities, and for this purpose:

(a) the ventilation ducts shall be placed so as to prevent the return of foul air into the air supply system through the shaft or tunnel entrance; and

(b) vehicles not directly involved in the underground work site shall be kept at a distance of at least 15 m from the access or opening connected to the underground work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.4.

8.3.5. The access to abandoned zones and any non-ventilated sector of the work site shall be declared prohibited for workers.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.5; O.C. 885-2001, s. 375.

8.3.5.1. Traffic lanes used by vehicles shall be free from any obstruction.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.5.1.

8.3.6. Self-propelled vehicles used for performing work in underground work sites shall:

(*a*) if they are powered by an internal combustion diesel type engine, be equipped with an exhaust-gas cooling system making it possible to keep the gas at 83 °C, whatever the motor's operating conditions may be;

(b) be equipped with position lights indicating their maximum width;

(c) not discharge into the air non-diluted exhaust gas containing over 0.25% carbon monoxide.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.6; O.C. 885-2001, s. 376.

8.3.7. On an underground construction site, any motor vehicle with a gasoline-powered engine and used for supervising operations shall:

(a) have an engine with a displacement of less than 6 litres;

(b) be equipped with emission control devices, in accordance with the standards prescribed in the Motor Vehicle Safety Regulations (C.R.C., chapter 1038) under the Motor Vehicle Safety Act (Statutes of Canada, 1993, chapter 16), with the same efficiency of performance as initially; and (c) have a maximum gross mass of 2,720 kg.

When such a motor vehicle stops in an underground construction site, its driver shall turn off the engine.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.7; O.C. 606-2014, s. 21.

8.3.8. When a vehicle used for the surveillance of underground work is immobilized, the engine shall be turned off.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.8.

8.3.9. The standard grades of distilled diesel fuel shall have:

(a) a flash point of a minimum of 65 °C; and

(b) a sulphur content less than 0.25%.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.9.

8.3.10. A warning system shall be set up to warn operators to stop the motor of the self-propelled vehicle and to prohibit the firing of explosives in case the ventilation system should break down.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.10.

8.3.11. When the concentration of vapours or flammable gases is greater than 25% of the lowest explosion limit, the work shall be stopped immediately in the affected zone and workers evacuated.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.11; O.C. 805-2020, s. 5.

8.3.12. The scrubber system or the oxy-catalyst exhaust purifier system shall be:

(a) checked at frequent intervals; and

(b) kept in good working order.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.3.12.

§8.4. Explosives

8.4.1. The provisions of Division IV of the Code that are consistent with this subdivision apply with the necessary modifications.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.1; O.C. 1959-86, s. 77.

8.4.2. Only explosives or a combination of explosives developing Class I blasting fumes as established by the Department of Energy, Mines and Resources of Canada may be used in an underground job-site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.2; O.C. 1959-86, s. 77.

8.4.3. Explosives and detonators must be transported directly to the site where they will be used in the underground job-site only at loading time, once all other operations have been completed and the equipment not required for the loading of explosives has been moved away.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.3; O.C. 1959-86, s. 77.

8.4.4. If explosives are transported in a shaft:

(a) explosives and detonators shall not be transported together;

(b) only the operator of the cage may travel with the explosives;

(c) the operator of the winch and the cage operator's helpers must be notified; and

(d) the transportation of other materials in the shaft is prohibited.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.4; O.C. 1959-86, s. 77.

8.4.5. When explosives or blasting accessories are hauled by a locomotive:

(a) explosives and blasting accessories must not be transported on the locomotive; and

(b) explosives carriers must be pulled. They may, however, be pushed by hand only.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.5; O.C. 1959-86, s. 77.

8.4.6. If explosives are stored in an underground jobsite, the employer must establish the supply of explosives for a 24-hour period and not store underground a greater amount of explosives.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.6; O.C. 1959-86, s. 77.

8.4.7. Storage:

(1) Explosives and detonators may be stored in an underground job site only if they are kept in a depot or powder house. This depot must be situated at least:

- (a) 750 m from a blasting area;
- (b) 450 m from the firing switch;
- (c) 90 m from the shaft or work zone; and
- (d) 7.5 m from a travelway.

(2) There must be at least one right angle turn in the pathway which connects the storage area with any work zone or travelway.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.4.7; O.C. 1959-86, s. 77.

8.4.8.-8.4.9. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 1959-86, s. 77.

§8.5. Drilling

8.5.1. Workers who use or work near drilling machines shall be positioned so as not to be hit or to lose their balance in case a drill-bit breaks, slips or is jammed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.1.

8.5.2. Workers shall not hold the steel drill-bit with their hands.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.2.

8.5.3. The supply of compressed air in the hose shall be cut off and drained before the displacement of the portable drills from one face to the other.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.3.

8.5.4. The platforms of jumbos shall be equipped:

(a) on their sides with fixed or removable guardrails if their height is greater than 1.2 m; and

(b) with an access ladder if their height is greater than 1.8 m.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.4.

8.5.5. Jumbos shall be solidly fixed in positions to avoid accidental displacement.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.5.

8.5.6. Drill-bits and their accessories shall be placed on the jumbos in boxes, and placed on supports or their equivalent when not used on the platform.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.6.

8.5.7. Unless equipped with a threaded coupling, compressed air hoses of 50 mm or more in interior diameter, mounted on a jumbo, shall be equipped at each end with a steel cable 5 mm in diameter or with an equivalent safety chain to prevent a whipping action.

Compressed air hoses smaller in diameter shall be equipped with a self-tightening device.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.7.

8.5.8. Any air line installed underground shall be:

(a) if metallic, placed so that it is protected from any shock capable of being caused by equipment or self-propelled vehicles; or

(b) if flexible, equipped with collars attached by a chain on each side of the coupling.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.5.8.

§8.6. Transportation of personnel and materials in a shaft

8.6.1. If the depth of a shaft is less than 60 m, hoisting apparatus or equipment governed by Subdivision 3.10 may be used for the transportation of personnel in a shaft.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.6.1.

8.6.2. If the depth of a shaft is to exceed or exceeds 60 m the transportation of personnel and materials shall be done according to the Regulation respecting occupational health and safety in mines (chapter S-2.1, r. 14).

A copy of the Regulation respecting occupational health and safety in mines shall be available on any construction site where underground work is being performed.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.6.2; O.C. 213-93, s. 540.

§8.7. Traffic

8.7.1. The employer shall prepare a compartment reserved exclusively for the circulation of workers by means of ladders or stairs in:

(a) any shaft over 30 m in depth;

(b) a raise inclined more than 55° with respect to the horizontal and dug on a length of more than 18 m except when the equipment used allows safe access.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.1.

8.7.2. The compartment for ladders or stairs in a shaft shall be separated from the compartment or the section of the shaft in which material, hoisting engines, or a counterweight are moved about, by a wire mesh made of No. 9 AWG galvanized steel wire and forming links of not more than 40 mm on a side so that the persons travelling in the compartment will not be struck by the conveyance or the counterweight or be hit by objects that may fall into the shaft.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.2; O.C. 393-2011, s. 21.

8.7.3. Shafts and raises inclined more than 30° with respect to the horizontal shall be equipped with ladders or stairs.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.3.

8.7.4. The minimum open space required for movement on a ladder shall be $600 \text{ mm} \times 600 \text{ mm}$.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.4.

8.7.5. A clearance space shall be ensured along the length of travelways.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.5.

8.7.6. Hazards which may result due to an abrupt change in the clearance space shall be clearly indicated in a travelway.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.6.

8.7.7. A flexible ladder may be used if the use of a rigid ladder is impossible and on condition that it is:

(a) solidly anchored at both ends; and

(b) suspended so that movement of hands and feet is not restricted.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.7.

8.7.8. Any travelway in an underground work site shall be provided with shelter spaces:

(a) if the travelway does not have, when a piece of equipment travels on tracks or by any other guided system, a minimum free space of 500 mm between the widest part of the machinery and the wall of each side of the equipment, or a free space of 600 mm on one side of the equipment;

(b) if the width of the travelway does not exceed the width of the equipment not travelling on tracks by at least 1.5 m.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.8.

8.7.9. The shelter spaces shall:

(a) have a free space of 750 mm and have a height of 1.8 m or be the height of the travelway if it is less than 1.8 m;

(b) be clearly indicated;

(c) be at maximum intervals of 30 m; and

(d) be free of any obstruction.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.7.9.

§8.8. Loading and transportation

8.8.1. Pneumatic equipment shall be equipped with a valve which cuts the air supply to the machine and this valve shall be kept closed when the equipment is not in operation.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.1.

8.8.2. If shelter spaces are required, self-propelled vehicles shall reduce speed to walking speed, and they shall sound a warning when they approach:

(a) workers or a work zone; or

(b) a zone where visibility is limited.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.2.

8.8.3. Any locomotive shall be equipped with:

(a) alarm signals;

(b) lights which can light up the travelway in 2 directions;

(c) marker lights indicating its maximum width; and

(d) an amber flashing light visible from any direction when in movement.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.3.

8.8.4. Railroads shall be:

- (a) kept in good condition;
- (b) constructed so as to prevent derailment;

(c) reasonably level; and

(d) free of bumps and obstructions.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.4.

8.8.5. A stop block shall be provided for a convoy or a car stationed on a plane with an incline greater than 1 1/2%.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.5.

8.8.6. Passenger wagons:

(1) If wagons are used to transport workers to their work areas, they shall:

(a) be passenger type wagons;

(b) be pulled at speeds appropriate to the condition of the travelway and the equipment used;

(c) be in sufficient number; and

(d) not carry materials and tools other than hand tools.

(2) The convoy shall be under the responsibility of a qualified person.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.6.

8.8.7. Workers who use passenger wagons shall:

(a) remain seated; and

(b) not board or leave the convoy while it is moving.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.7.

8.8.8. Only those self-propelled vehicles directly involved in an underground work site may be stationed during the periods when the underground work site is not in operation.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.8.8.

§8.9. Hygiene

8.9.1. Underground work site - Changing room and shower: In every underground work site, the employer must provide workers with a changing room located on the surface that complies with section 3.2.11 and a shower that complies with section 3.2.15.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.9.1; O.C. 393-2011, s. 22.

8.9.2.-8.9.3. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 393-2011, s. 23.

8.9.4. The employer shall put waterproof clothing at the disposal of any worker who works in an excessively damp zone.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.9.4.

§8.10. Lighting

8.10.1. Any worker shall carry a miner's lamp unless a fixed system of lighting is available in the underground work site.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.10.1.

8.10.2. If a fixed lighting system exists, the lighting intensity shall be appropriate to the nature of the areas or to the work done in any place where persons work or circulate.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.10.2.

8.10.3. Stairs and ladders in shafts shall be lighted from top to bottom and landings shall also be lighted.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.10.3.

8.10.4. If workers do not have miner's lamps or portable lamps, an emergency lighting system with an automatic relay shall be available in the underground work site. This system shall be kept in good working condition for a takeover in case of a break in the main supply of electrical current.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.10.4.

§8.11. Communication and standard signals

8.11.1. During the sinking of a shaft, a temporary communications system must be provided.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.1.

8.11.2. At the end of all shaft sinking operations and before tunnelling operations, each shaft used for hoisting whose depth is more than 15 m shall be equipped between the bottom of the shaft and the surface with 2 distinct signalling systems which may be electric, pneumatic or mechanical. One of these systems shall be connected to a telephone or to a speaking-tube.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.2.

8.11.3. Hoisting signals shall be adopted and used at each hoisting shaft in conformity with Schedule 4, if visual signals cannot be understood.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.3.

8.11.4. Standardized signals may be established which correspond with local conditions, provided that they are easily discernible and do not conflict with established standard signals.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.4.

8.11.5. A legible copy of standard signals and of standardized signals if need be, shall be posted in full view of the hoisting operator and in all locations where signals may be given or received.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.5.

8.11.6. No worker shall be taken into a shaft as long as the appropriate signals have not been previously given.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.6.

8.11.7. The devices for transmitting signals shall be kept within easy reach at the bottom of the shaft during sinking operations.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.7.

8.11.8. When tunnels cover a distance of more than 150 m from the base of a shaft, a telephone system shall be installed with extensions:

- (a) at the surface;
- (b) at the base of the shaft; and

(c) at a maximum distance of 75 m from the face.

Telephone outlets shall be installed at intervals of 150 m along the tunnel.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.8.

8.11.9. The beginning of any fire, explosion or other emergency shall be signalled by 9 intermittent signals originating from the lighting system.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.11.9.

§8.12. Emergency measures

8.12.1. An evacuation and rescue procedure shall be set up and kept up-to-date. All means of egress shall be indicated and all workers shall be familiar with this procedure and the means of egress.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.12.1.

8.12.2. A rescue team shall be available at all times. This team shall be made up of at least 3 persons who shall:

(a) be in excellent physical condition;

(b) have received pertinent training:

i. in rescue procedures;

ii. in the use, maintenance and the ultimate capacity of oxygen respirators; and

iii. in the use and maintenance of fire fighting equipment; and

(c) carry out rescue drills.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.12.2.

8.12.3. A control system for entrances and exits shall be enforced in an underground work site and register of workers involved in the underground works shall be available at the surface.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.12.3.

§8.13. Minimum age

8.13.1. No person below 18 years of age shall be employed underground, at the face of an open pit site or at the controls of hoisting or moving equipment.

R.R.Q., 1981, c. S-2.1, r. 6, s. 8.13.1.
DIVISION IX *Revoked, O.C. 1005-2015, s. 1.*

§9.1. Generalities Revoked, O.C. 1005-2015, s. 1.

9.1.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.1; O.C. 1005-2015, s. 1.

9.1.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.2; O.C. 1005-2015, s. 1.

9.1.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.3; O.C. 1005-2015, s. 1.

9.1.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.4; O.C. 1005-2015, s. 1.

9.1.5. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.5; O.C. 1959-86, s. 78; O.C. 1005-2015, s. 1.

9.1.6. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.6; O.C. 1005-2015, s. 1.

9.1.7. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.7; O.C. 1005-2015, s. 1.

9.1.8. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.8; O.C. 1005-2015, s. 1.

9.1.9. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.1.9; O.C. 1005-2015, s. 1.

§9.2. Hours of work and rest periods Revoked, O.C. 1005-2015, s. 1.

9.2.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.1; O.C. 1005-2015, s. 1.

9.2.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.2; O.C. 1005-2015, s. 1.

9.2.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.3; O.C. 1005-2015, s. 1.

9.2.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.4; O.C. 1005-2015, s. 1.

9.2.5. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.5; O.C. 1005-2015, s. 1.

9.2.6. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.6: O.C. 1005-2015, s. 1.

9.2.7. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.7; O.C. 1005-2015, s. 1.

9.2.8. (Revoked). R.R.O., 1981, c. S-2.1, r. 6, s. 9.2.8; O.C. 1005-2015, s. 1. 9.2.9. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.9; O.C. 1005-2015, s. 1. **9.2.10.** (*Revoked*). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.2.10; O.C. 1005-2015, s. 1. §9.3. Air supply Revoked. O.C. 1005-2015. s. 1. (Revoked). 9.3.1. R.R.Q., 1981, c. S-2.1, r. 6, s. 9.3.1; O.C. 1005-2015, s. 1. **9.3.2.** (*Revoked*). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.3.2; O.C. 1005-2015, s. 1. 9.3.3. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.3.3; O.C. 1005-2015, s. 1. 9.3.4. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.3.4; O.C. 1005-2015, s. 1. §9.4. Air locks and working chambers Revoked, O.C. 1005-2015, s. 1. **9.4.1.** (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.1; O.C. 1005-2015, s. 1. 9.4.2. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.2; O.C. 1005-2015, s. 1. 9.4.3. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.3; O.C. 1005-2015, s. 1. 9.4.4. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.4; O.C. 1005-2015, s. 1. 9.4.5. (Revoked). R.R.O., 1981, c. S-2.1, r. 6, s. 9.4.5; O.C. 1005-2015, s. 1. 9.4.6. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.6; O.C. 1005-2015, s. 1. 9.4.7. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.7; O.C. 1005-2015, s. 1. 9.4.8. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.8; O.C. 1005-2015, s. 1. 9.4.9. (Revoked). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.9; O.C. 1005-2015, s. 1. **9.4.10.** (*Revoked*). R.R.Q., 1981, c. S-2.1, r. 6, s. 9.4.10; O.C. 1005-2015, s. 1.

§9.5. *Pressure gauges Revoked*, *O.C.* 1005-2015, *s.* 1.

9.5.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.5.1; O.C. 1005-2015, s. 1.

9.5.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.5.2l; O.C. 1005-2015, s. 1.

9.5.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.5.3; O.C. 1005-2015, s. 1.

9.5.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.5.4; O.C. 1005-2015, s. 1.

§9.6. Electric power supply and lighting Revoked, O.C. 1005-2015, s. 1.

9.6.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.6.1; O.C. 1005-2015, s. 1.

9.6.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.6.2; O.C. 1005-2015, s. 1.

9.6.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.6.3; O.C. 1005-2015, s. 1.

§9.7. *Hygiene and welfare Revoked*, *O.C.* 1005-2015, *s.* 1.

9.7.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.7.1; O.C. 393-2011, s. 24; O.C. 1005-2015, s. 1.

9.7.2. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.7.2; O.C. 1005-2015, s. 1.

§9.8. Medical examinations and services Revoked, O.C. 1005-2015, s. 1.

9.8.1. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.1; O.C. 1005-2015, s. 1.

9.8.2. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.2; O.C. 1005-2015, s. 1.

9.8.3. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.3; O.C. 1005-2015, s. 1.

9.8.4. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.4; O.C. 1005-2015, s. 1.

9.8.5. (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.5; O.C. 1005-2015, s. 1.

9.8.6. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.6; O.C. 1005-2015, s. 1.

9.8.7. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.7; O.C. 1005-2015, s. 1.

9.8.8. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.8; O.C. 1005-2015, s. 1.

9.8.9. (*Revoked*).

R.R.Q., 1981, c. S-2.1, r. 6, s. 9.8.9; O.C. 1005-2015, s. 1.

DIVISION X WORK ON ROADS OPEN TO TRAFFIC *§10.1. Definitions*

10.1.1. In this subdivision:

"public highway" means a public highway within the meaning of the Highway Safety Code (chapter C-24.2); (chemin public)

"road vehicle" means a road vehicle within the meaning of the Highway Safety Code. (véhicule routier)

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.1.1; O.C. 995-91, s. 12.

§10.2. Scope

10.2.1. This Division applies to any construction site or part thereof located on or bordering on a public highway or private road open to public vehicular traffic.

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.2.1; O.C. 995-91, s. 12.

10.2.2.-10.2.3. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, O.C. 995-91, s. 12.

§10.3. Traffic signs

10.3.1. The principal contractor must ensure that any construction site or part of a construction site situated on or near a public highway or a private road open to public vehicular traffic has traffic signs that comply with the standards of Chapters 1, 4 and 6 of Volume V of the manual entitled "Traffic Control Devices", determined and set out by the Minister of Transport under the second paragraph of section 289 of the Highway Safety Code (chapter C-24.2).

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.3.1; O.C. 995-91, s. 12; O.C. 873-2003, s. 1.

10.3.2. Where traffic must be directed by a flag person for road users, the employer must ensure that the flag person

(1) is aware of all the responsibilities inherent in his or her work;

(2) has undergone training relating to his or her responsibilities recognized by the Joint Sector-Based Construction Association on Occupational Health and Safety; (3) wears high-visibility safety apparel and is equipped with other accessories in compliance with the standards determined by the Minister of Transport and recorded in Volume V of the manual entitled "Traffic Control Devices", determined and set out by the Minister of Transport under the second paragraph of section 289 of the Highway Safety Code (chapter C-24.2).

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.3.2; O.C. 995-91, s. 12; O.C. 1078-2015, s. 7.

§10.4. Other precautions

10.4.1. High-visibility safety apparel: Subject to paragraph 1 of section 2.8.4 and to paragraph 3 of section 10.3.2, the wearing of fluorescent orange high-visibility safety apparel of Class 2 or 3 and of Level 2 that complies with CSA Standard Z96, High-Visibility Safety Apparel, is mandatory for every worker who performs tasks on or near a road where self-propelled vehicles are likely to hit a worker.

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.4.1; O.C. 995-91, s. 12; O.C. 1078-2015, s. 8.

10.4.2. Electric lighting: Electric lighting used in work areas must be positioned in such a manner as not to dazzle users of the road.

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.4.2; O.C. 995-91, s. 12.

10.4.3. Protection of manholes:

(1) A portable guardrail made of steel or wood must be used to prevent falls into manholes.

(2) Where the guardrail is made of steel, it must:

(a) be made of metal tubes at least 22 mm in diameter enclosing the manhole on 3 sides, the fourth side being closed off with 2 chains;

(b) be bright in colour, have a middle rail and bear flags with reflective bands; and

(c) conform to figure 1 in Schedule 5.

(3) Where the guardrail is made of wood, it must:

(a) be made of boards 100 mm in width and must enclose the manhole on 3 sides, the fourth being closed off with 2 chains;

(b) bear vertical black bands 100 mm in width on an orange background;

(c) be fitted with flags with reflective bands; and

(d) conform to figure 2 of Schedule 5.

R.R.Q., 1981, c. S-2.1, r. 6, s. 10.4.3; O.C. 995-91, s. 12.

10.4.4.-10.7.4. (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6; O.C. 995-91, s. 12.

DIVISION XI

WORK ABOVE OR NEAR WATER

11.1. On a site where work will be carried out above or near water, the principal contractor must, before the work begins,

(1) prepare a description of the body of water or watercourse indicating, in particular,

(a) the type of body of water or watercourse and its characteristics during the time of the work;

(b) the means used to know the water temperature and weather conditions;

(2) prepare a description of the work indicating, in particular,

(a) the work sites;

(b) the nature of the work;

(c) the number of workers on the work sites;

(d) the work schedules;

(e) the dates on which the work begins and ends;

(f) the location of the work, lunch and rest areas;

(g) the platforms, barges and other boats, specifying their dimensions, capacity and respective use;

(3) prepare a water transportation plan adapted to the specific conditions of the work and the characteristics of the body of water or watercourse indicating, in particular,

(a) the name of each person in charge of water transportation operations;

(b) the location of the boarding and landing areas;

(c) the boats used to transport workers, specifying the name of each driver, the routes and the direction of the boats;

(d) the safety rules to be complied with during water transportation;

(4) prepare a rescue plan adapted to the specific conditions of the work and the characteristics of the body of water or watercourse indicating, in particular,

(a) the name of each person in charge of rescue operations;

(b) the name of each person in charge of the maintenance or inspection of rescue equipment;

(c) the rescue procedures in case of a worker falling into water, wreck, fire or other accident, and the instructions regarding those matters;

(d) the name of each rescue attendant and each first-aider;

(e) the emergency call code used to start rescue operations;

(f) the location of rescue equipment;

(g) the location of first-aid stations;

(h) if applicable, the type and number of boats intended for rescue.

The descriptions of the work and body of water or watercourse, as well as transportation and rescue plans, must be posted at work sites. They must also be integrated into the prevention program, specifying the means used to inform workers of their content, in particular with regard to safety rules and instructions intended for workers.

O.C. 513-2015, s. 5.

11.2. The principal contractor or the person designated by the principal contractor to prepare the descriptions and plans prescribed by section 11.1 and each person in charge of transportation or rescue operations must hold a certificate from the Association paritaire pour la santé et la sécurité du travail du secteur de la construction or the Lifesaving Society issued following training of at least 7 hours pertaining in particular to the following:

(*a*) the hazards associated with working above or near water and the prevention measures to counter those hazards;

(b) the hazards associated with cold water immersion and the prevention measures to counter those hazards;

(c) the identification of the various pieces of safety equipment required to work above or near water and their use;

(d) the identification of the various pieces of rescue equipment required to recover a person in the water and their use;

(e) the requirements of the federal and provincial legislation regarding work above or near water and the use of a boat on a body of water or watercourse;

(f) the preparation and application of transportation and rescue plans.

O.C. 513-2015, s. 5.

11.3. The equipment required by a rescue plan referred to in subparagraph 4 of the first paragraph of section 11.1, as well as any accessories, must be

(a) adapted to the intended use, to the specific conditions of the work and the characteristics of the body of water or watercourse;

(b) inspected and kept in good working order;

(c) present and visible on the premises during working hours;

(d) accessible so that a quick intervention is possible. $\overline{O.C. 513-2015, s. 5}$.

11.4. Where a rescue plan includes the use of a boat, the boat must, in addition to the requirements provided for in section 11.3,

(1) in particular, be

(a) adapted and equipped for the research and recovery of persons;

(b) equipped with a propulsion system adapted to the boat;

(c) equipped with the following rescue equipment:

i. 2 rope bags, each containing 1 single-length buoyant heaving line that remains flexible, with a minimum diameter of 9.5 mm and a minimum length of 15 m;

ii. a life buoy with an outside diameter of 762 mm, approved by Transport Canada or by a body recognized by Transport Canada, as evidenced by the tag or approval stamp affixed to it;

iii. a boat hook;

(2) be used by a team of at least 2 rescue attendants trained in the approach and recovery of a person in conditions and according to the characteristics of the body of water or watercourse where their assistance is needed, and who meet either of the following conditions:

(a) hold a Pleasure Craft Operator Card issued by Transport Canada and a certifi cate from the Lifesaving Society attesting that training was received in the approach and recovery of persons; or

(b) hold a certificate of competency issued by Transport Canada, other than a Pleasure Craft Operator Card, and a certifi cate attesting that training was received in Marine Emergency Duties (MED) issued by Transport Canada. $\overline{O.C. 513-2015, s. 5}$.

11.5. During work above or near water, life buoys with an outside diameter of 762 mm must be placed and spread out over the entire length of the site where work is performed, at a maximum linear distance of 60 m between life buoys. The life buoys must be approved by Transport Canada or by a body recognized by Transport Canada, as evidenced by the tag or approval stamp affixed to each of them.

O.C. 513-2015, s. 5.

11.6. A sound alarm system intended to start rescue operations must be present on the work sites.

O.C. 513-2015, s. 5.

SCHEDULE

SCHEDULE 0.1 (s. 3.5.6)

INCLINE OF A LADDER NOT PERMANENTLY FASTENED





Indicates the zone within which the foot of the ladder must be placed.

Horizontal distance between the foot of the ladder and the vertical plane against which it is propped, in relation to the length of the ladder

Length of the	Horizontal distance											
(metres)	¹ / ₄ × L (metres)	¹ / ₃ × L (metres)										
8	2	2.7										
10	2.5	3.3										
12	3	4										
14	3.5	4.7										
15	3.75	5										

O.C. 329-94, s. 76.

SCHEDULE 0.2

(s. 3.9.10)

ANCHORS OF A SCAFFOLDING LESS THAN 18 METRES WHEN A TARPAULIN OR NET IS USED

Table 1 - Types of anchors necessary for holding a scaffolding covered with a tarpaulin¹ according to its dimensions and the region

Region	Surface 3m x 3m	Surface 3m x 6m
Gaspésie-Îles-de-la-Madeleine/ Côte-Nord / Nord-du-Québec	Tube ²	n.a.
Bas-Saint-Laurent	Pin #9 ³	n.a.
Chaudière-Appalaches / Estrie / Laurentides / Laval / Mauricie / Montérégie / Montréal / Outaouais / Capitale-Nationale / Saint-Jean-sur- Richelieu / Valleyfield / Yamaska	Pin #9 ³	Tube ²
Abitibi-Témiscamingue/ Lanaudière / Saguenay–Lac-Saint-Jean	Pin #9 ³	Tube ²



Table 2 - Types of anchors necessary for holding a scaffolding covered with a net according to its dimensions and the region

Region	Surface 3 m x 6 m	Surface 3 m x 9 m
Gaspésie-Îles-de-la-Madeleine/ Côte-Nord / Nord-du-Québec	Tube ²	n.a.
Bas-Saint-Laurent	Pin #9 ³	Tube ²
Chaudière-Appalaches / Estrie / Laurentides / Laval / Mauricie / Montérégie / Montréal / Outaouais / Capitale-Nationale / Saint-Jean-sur- Richelieu / Valleyfield / Yamaska	Pin #9 ³	Tube ²
Abitibi-Témiscamingue/ Lanaudière / Saguenay-Lac-Saint-Jean	Pin #9 ³	Tube ²



NOTES:

(1) The tarpaulin or net shall be able to resist the loads and wind gusts to which it is exposed.

(2) Tube: metal tube with a welded bracket, and a hole at one of its ends The tube is linked to the front by a concrete mechanical anchor or the equivalent, and to the scaffolding by a metal fastener The minimum capacity of the mechanical anchor shall be 9.0 kN with a safety factor of 2.

(3) Pin #9: metal wire caliber #9 with a double loop, having a diameter of 3.8 mm, used as tie rod, fastened to the scaffolding at one end and at the other end to a mechanical anchor (expansion shield, eye bolt, etc.), in accordance with CSA standards S269.2-M87 and Z797-09. The minimum capacity of the anchor shall be 5.4 kN with a safety factor of 2.

O.C. 640-2019, s. 5.

SCHEDULE 1

(Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 1; O.C. 425-2010, s. 5.

SCHEDULE 2 HANDLING AND USE OF EXPLOSIVES

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 2.

SCHEDULE 2.1

(Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 2.1; O.C. 57-2015, s. 59.

"SCHEDULE 2.2 Blasting logbook (s. 4.7.10.)

Name of enterprise:	Address
	(optional):
Location of site:	Client:
Information on blasting Location: 	Weather conditions Temperature: °C
Date:	Sunny:
Hour:	Cloudy:
Chaining (option):	Rain/snow:
Number of holes and drilling diameter: Burden and spacing: Height of drilling in metres: Height of collar: Height of overburden: Nature of tamping (clean, crushed stone):	
Blasting mats (type):	Distance from closest structures (building/bridge/road):
Explosives	
Type: Number of detenator:	
Quantity of explosives used (primers, explosives used (primers, explosives used (primers, explosives))	plosives) in units, bags, cases or
Remarks:	
Name of shot-firer:	Signature:

Blasting pattern (Required information)

- Departmentation of the mining sequence (including delays)
 Description of explosives per hole (dimensions, number and weight)
 Disposition of blastholes
- Number and orientation of free faces
 Biasting direction
 Lidentification of the firing sequence (including
 delays)
 Description of explosives per hole (dimensions,
 Description of explosives per hole (dim



R.R.Q., 1981, c. S-2.1, r. 6, sch. 2.2; O.C. 1959-86, s. 79; O.C. 57-2015, s. 60.

2 500	$3\ 000$	4 000	$5\ 000$	6 000	7 000	$10\ 000$	$15\ 000$	$20\ 000$	25 000	$30\ 000$	$40\ 000$	$50\ 000$	75 000	$100\ 000$	$125\ 000$	150 000
32	34	38	41	44	46	52	59	66	70	74	82	88	100	110	120	128
270	300	350	380	400	420	480	540	600	640	680	760	820	940	1 050	1 115	1 175
135	150	175	190	200	210	240	270	300	320	340	380	410	470	525	558	588
2 500	$3\ 000$	4 000	$5\ 000$	6 000	7 000	$10\ 000$	$15\ 000$	$20\ 000$	25 000	$30\ 000$	40 000	$50\ 000$	75 000	$100\ 000$	125 000	150 000

References respecting the Table of Distances in Schedule 2.3.

(1) any outdoor meeting place, road, railroad or waterway, except a road leading to the depot.

(2) any building or any other depot.

(3) any other depot surrounded by dikes.

Remark: For computing the distance,

(*a*) 1,300 detonators or 150 detonating relays are equal to 1 kilogram of explosives;

(b) except for the depot for detonators and detonating relays, 2 depots placed side by side may be considered as a single depot having a capacity equal to the total capacity of both depots.

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 2.3; O.C. 1959-86, s. 80; O.C. 57-2015, s. 61.

SCHEDULE 2.4

(Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 2.4; O.C. 1959-86, s. 81.

SCHEDULE 2.5 (Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 2.5; O.C. 1959-86, s. 81.

SCHEDULE 2.6

Evaluation of the maximum authorized particle speed, of the distance between blasting and buildings or of the acceptable frequency of vibrations (s. 4.7.5)

The employer must comply with the prescribed limits, according to one of the following 3 methods:

1. in the table below:

TABLE 2.6.1 - MAXIMUM AUTHORIZED PARTICLE SPEED ACCORDING TO THE DISTANCE OF STRUCTURES												
Distance from blasting site	Maximum speed authorized											
0 to 90 m (300 ft)	31.75 mm/s (1.25 in/s)											
91 to 1,524 m (301 to 5,000 ft)	25.4 mm/s (1 in/s)											
1.525 m and more (5,000 ft)	19 mm/s (0.75 in/s)											

2. in one of the graphs below:

GRAPHS 2.6.2 - ESTIMATE OF THE MAXIMUM AUTHORIZED PARTICLE SPEED ACCORDING TO THE FREQUENCY OF VIBRATION (1N/S)



The employer must use, according to the manufacturer's instructions, a seismograph to monitor the velocity of particles to ensure the compliance of the results with table 2.6.1 or graphs 2.6.2 as provided above. The method for monitoring vibrations and the calculation of frequency must be approved by an engineer.

3. in the proportionate distance equation shown in the table below:

TABLE 2.6.2 - COMPUTATION OF THE MINIMUM DISTANCE TO BE KEPT BETWEEN A STRUCTURE AND A BLASTING BASED ON EXPLOSIVE LOAD													
Maximum quantity of explosives fired in less than 8 milliseconds Dictance from blacting cite Metric Units English Units													
Distance from blasting site	Metric Units (W in kg and D in m) Impériale	English Units (W in lb and D in ft)											
Less than 92 m (300 ft) 92 to 1,524 m (301 to 5000 ft) More than 1,524 m (5000 ft)	$W = (D/22.6)^2$ $W = (D/24.9)^2$ $W = (D/29.4)^2$	$W = (D/50)^2$ $W = (D/55)^2$ $W = (D/65)^2$											

W = Maximum quantity of explosives (in kilograms or pounds) that may detonate in less than 8 milliseconds.

 $D=\mbox{Distance}$ to be kept between the blasting area and the closest structure to be protected.

O.C. 57-2015, s. 62.

SCHEDULE 3

(Revoked). R.R.Q., 1981, c. S-2.1, r. 6, Sch. 3; O.C. 1005-2015, s. 1.

SCHEDULE 3.1

(Revoked). R.R.Q., 1981, c. S-2.1, r. 6, Sch. 3.1; O.C. 1005-2015, s. 1.

SCHEDULE 3.2

(Revoked).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 3.2; O.C. 1005-2015, s. 1.

SCHEDULE 4 (s. 8.11.3) CODE OF SIGNALS

- (1) The following code of signals shall be used in underground work sites :
- **1 bell** Stop immediately if in motion (executive signal).
- 1 bell Hoist (executive signal).

3-3-1 bells — Hoist slowly.

2 bells — Lower (executive signal).

3-3-2 bells — Lower slowly.

- **1-2 bells** Chairing and unchairing.
- 3 bells (Cautionary signal). Men about to ascend or descend. This signal shall be given before men are permitted to enter or leave the hoisting conveyance. It shall also be given in case a stop has been made at a level and men in the conveyance are to be hoisted or lowered to another level. In all cases, after the 3-bell signal has been given, the conveyance shall not be moved until the signals to hoist or lower have been given.
- **4 bells** (Blasting signal). Hoistman shall answer by raising the bucket, cage or skip a few feet and letting it back slowly. Following a 4-bell signal, only a 1-bell signal shall be required to signal for hoisting of men away from a blast, and the hoistman shall remain at the controls until the act of hoisting has been completed.
- **5 bells** (Release signal). The hoistman may, after receiving this signal, act at his own discretion to perform any movement or series of movements involving the hoisting conveyance.
- 9 bells (Danger signal, special cautionary). To be given only in case of accident, fire or other danger. The signal for the level at which the emergency exists shall be given following the giving of the danger signal. These signals should be given on the voice communication system or the call system except in shaft sinking or at shaft stations which are not provided with such communication systems.
- (2) The following method and order shall be observed in giving signals:
 - (a) strokes on the bell shall be made at regular intervals;
 - (b) when men are being hoisted or lowered, signals shall be given in the order designated :

1st — cautionary signal, 2nd — destination signal, 3rd — executive signal.

- (3) The hoistman shall delay briefly before moving the hoisting conveyance after receiving a signal designating a movement at any time that men are carried. In case he is unable to act within one minute of the time he has received any complete signal, he shall not move the hoisting conveyance until he has again received another complete signal.
- (4) When a hoistman has received a 3-bell signal, he shall remain at the hoist controls until he has received the signal designating the movement required and has completed that movement. After he has commenced the movement, he shall complete it without interruption, unless he receives a stop signal or in case of emergency.

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 4.

SCHEDULE 5 (S. 10.4.3)

PORTABLE GUARDRAILS





Wooden guardrail. Fig. 2

R.R.Q., 1981, c. S-2.1, r. 6, sch. 5; O.C. 995-91, s. 13.

SCHEDULES 5.1 TO 5.3 (*Replaced*).

R.R.Q., 1981, c. S-2.1, r. 6, Sch. 5.1; R.R.Q., 1981, c. S-2.1, r. 6, Sch. 5.2; R.R.Q., 1981, c. S-2.1, r. 6, Sch. 5.3; O.C. 995-91, s. 13.

SCHEDULE 6

(s. 3.24.18)



R.R.Q., 1981, c. S-2.1, r. 6, Sch. 6; O.C. 391-2011, s. 3.

SCHEDULE 7

(Revoked).

O.C. 53-90, s. 11; O.C. 483-2021, s. 10.

SCHEDULE 8

(s. 7.2.1) TRAINING AND CERTIFICATE OF A LOW VELOCITY EXPLOSIVE ACTUATED TOOL OPERATOR

8.1 Only persons so authorized by the manufacturer of a low velocity explosive actuated tool may be instructors and, in that capacity, supervise the training and qualification of workers wishing to use a low velocity explosive actuated tool.

8.2 Any worker wishing to use a low velocity explosive actuated tool shall have received training covering the following points :

(1) identification of the tool's parts and accessories;

(2) how to operate the tool and its accessories, such as a spall stop;

(3) maintenance of the tool;

(4) identification of damaged or worn parts;

(5) identification of the strength of explosive charges.

8.3 A low velocity explosive actuated tool operator's certificate shall be issued to any worker who:

(1) has received the training provided for in section 8.2;

(2) has demonstrated to his instructor that he is able to perform the operations provided for in section 8.2.

The certificate is issued by the instructor who dispenses the training.

8.4 A low velocity explosive actuated tool operator's certificate shall contain the following information :

(1) the worker's name;

(2) the attestation that the worker has received the training provided for in section 8.2 and has demonstrated that he is able to perform the operations provided for in that section;

(3) the date of issue of the certificate;

(4) the name or trademark of the manufacturer of the tool;

(5) the model of the tool;

(6) the name and signature of the instructor who dispensed the training provided for in section 8.2.

O.C. 329-94, s. 77.



-						_														1				
3																								
30																								
29																								
28																								
27																								
26																								
25																				1				
24																								
23																				1				
22																				1				
51																				1				
5																				1				
6																								
8																								-
5							-							-	-							\vdash	\vdash	
19							-							-	-							\vdash	\vdash	
5						-	-	-			-			-	-	-			-			\vdash	\vdash	-
4							-	-			-			-	-	-			-			\vdash	\vdash	-
31		-					-	-	\vdash	\vdash	-		\vdash	-	-	-		\vdash	-			\vdash	\vdash	-
1																								-
=								_			-					_								
5		_								_			_	_				_						
-								_			-					_								
5		_											_											
-																								
9																								
ŝ																								
4																								
З																								
2																								
-																								
×																								
D																								
]																							
			>							ts														
			lov							bel														
			inc							Ы				e										
			γp							r a.				vic					ces		Ê			
			an	E		e				ate				de			LS		evi		tio	su		Ê
			ers	ala		vic				Υ.				ing			ISI	ols	g d	m	pec	Ē		00
	Ļ		vip	đ		de				oil		4		niti	u			ntr	Ŀ.	15	ins	lge		c p
	RA	ual	γþ	lck	H	ng	or	ato1		ks,		litc		lir	utio		Ā	CO	ocl	ĮŻ	al	į		uli
	NE	an	niel	l b	atc	Ľ.	cat	lic		VOL		SW		ach	lice		Z	and	lpt	1 A	isu	pu		dra
	B	s n	dsh	anc	dic	Wa	ndi	Ë.		erv	nrd:	mit	3er)LO	ıqn		0	es :	s ar	6	2	1S 8		(hy
	-	er'	vin	ILS	E.	ng	h ii	ou	s	ddi	gug	E	list	app	l D		ĮŻ.	tch	lets	lĝ	nar	pit		es (
		tur	s, v	rro	gle	cki.	ngt	tati	n gu		1 D	lg lg	lgu	ne	ar	ing	M	slut	atch	[H	ter	d	105
		fac	ow: ster	Ē	an	loc	lei	lo	ga	e l	ine	an	xtiı	r lii	ing) ck		s, c	, r8		Su	co	ste	d b
		nu	ros	Ë	ы	<u>-</u> -	- mo	E H	als,	gin	[chi	E I	e e	wei	eas	P I		ake	wls		<u> </u>	IS,	E E	an
		Μŝ	W ₁	H	Bo	Τ	Bo	D	Di:	En	Ϊ	Bo	Fir	Po	ģ	Ca		Br	Pa			Pir	Bo	Ö
		-1	~i	<i>…</i>		<u>.</u> .	<u>ج</u>	~	~				~;	~	÷.			<u>.</u>	7.			~		
		_	. 4	6,	7.	a.)	9	10		5	lΞ	=	12	13	14	13		16	1			3	1.	20

ORIES ction)	es					ety latches	sses							
ROPES AND ACCESSC (Summary visual inspec	21. Main and auxiliary hoist rope	22. Boom hoist rope	23. Rope spooling on drums	24. Boom and jib pendants	25. Rope attachments	26. Pulley blocks, hooks and safe	27. Sheaves, bearings and harnes	CARRIER	28. Brakes	29. Tires and wheel lugs	30. Engine oil and water	31. Lights and flashers	32. Heater and window defroster	33. Outriggers



NOTE: All the monthly tests prescribed by the manufacturer's manual shall be carried out.

tail (ref. CSA Z-150).	COMMENTS											
ed in detail (ref.												
irs shall be recorde	INITIALS											
damage or repai	DATE											

Enter below the complete details with respect to any maintenance, inspections, tests and the revisions and modifications carried out. All incidents, mishaps, dar

O.C. 1413-98, s. 29.



-															_		_								
03																									
) 3(
825	_	_	_								_	_									_	_			
728	_	_	_	_						_	_	_									_	_	_		
5																									
5 26																									
125																									
3 5																									
5																									
1 5	_	_									_	_									_	_			
02																									
9 2(_																			
8 1																									
1	_	_	_	_						_	_	_									_	_	_		
61																									
51		_					-	-		_	_	_					-	_		_	_				
4		_	_	_								_									_	_	_	\vdash	
31							-	_								_	-	_							
21		_				-	-	-								-	-	-	_	_					
11		_	_	_			-	-			_	_					-	-			_	_	_		
01																				_					\square
9 1																									
~																									
5				_																			_		\square
9																									
Ś																									
4																									
З																									
2																									
-																									
ž	S																								
DA	Z															ch									
_	R															lat									
	R															ety					c.)				
	ΥE															saf					, et				
	õ		_												s	pu					nts				
	E	s	tion												ster	k a					joi				
	ž	per	icat			e									sys	0 Q					ns,				
	Е	wi	unu			vic			ste						lge	ts,					atio				
	ΥE	eld	nm		ice	de		ator	fro	tch					lor	len			er		rm				
	RA	shi	3		dev	nan		li:	ď	wi	les				lc l	퇹			ord		efo				
	E	ind	dio		ы ц	adr		Ē.	low	nit s	itcl				da	tta	gu		ng		s, d	ų			H
	z	мр	r ra		ani	de		10	/in	lin	SW		6	e	tan	e,	50		nn		eld	vit			she
	ΥA	an	e 0	etei	erv	and		osit	n pi	ght	mit	ke	rak	oral	por	rop	d ri	50	l ru	ng	N)	it s	Ħ		ingu
	^k	SWC	nor	- mo	ath	slc.		۲ م	r an	hei	y li	bra	y bi	1g	dn	l agu	an	din	anc	cki	ure	im	be	Ъ	xtin
	N	ndc	epł	em	we a	ntr	E	lle	ater	ok	olle	ist	lle	wir	y. s	isti	ngs	luno	ise	b lc	ucti	ad l	ety	nya	e e
	Ĕ	Wi	Te	An	Jib	ပီ	Ho	Ĕ	He	Hо	Tr	Ho	1 _r	Sle	ß	Ho	SI	Ū	No	Ca	Str	Lo	Sat	Lai	Fir
	LA		~	<i></i>	÷	<u>.</u>	<u>.</u>		÷.		Ö.		~i	e.	4	<u>~</u>	<u>ن</u>		œ.	9.	Ö.		2	m.	4
	ŝ			1.6.1	7	, °,		Ľ.	~	<i>.</i>	ĭ	Ξ	1	1	-	12	Ĕ	-	1	1	2(5	5	3	ň


ed and repairs carried out shall be recorded in detail (ref. CSA standard Z-248).												
	COMMENTS											
	INITIALS											
all damage sustaine	DATE											

Enter below the complete details in respect of any maintenance, inspections, testings and revisions and modifications carried out. All incidents, mishaps and all damage sustained and repairs carried out shall be recorded in detail (ref. CSA standard Z-248).

Г

O.C. 1413-98, s. 29.

© Éditeur officiel du Québec, 2025 Toute reproduction interdite

2015 (O.C. 428-2015)

Section 9

Sections 1 to 5 of this Regulation apply, from the following dates, to open sites that the number of workers indicated will occupy simultaneously at a particular stage of the work or for the duration of the work:

(1) 18 June 2015 if there are 100 or more workers;

(2) 18 December 2015 if there are between 50 and 99 workers;

(3) 18 June 2016 if there are 50 workers or fewer.

Despite section 8, sections 162 to 165 of the Regulation respecting occupational health and safety continue to apply to construction sites or, where applicable, to the categories of sites specified in that Regulation, until the rules provided for in sections 1 to 5 apply to them in accordance with the first paragraph.

R.R.Q., 1981, c. S-2.1, r. 6 Decision 83-11-17, 1984 G.O. 2, 849 O.C. 21-85, 1985 G.O. 2, 331 O.C. 1959-86, 1987 G.O. 2, 188 S.Q. 1986, c. 89, s. 50 O.C. 1960-86, 1987 G.O. 2, 202 S.Q. 1988, c. 23, s. 73 O.C. 53-90, 1990 G.O. 2, 333 O.C. 54-90, 1990 G.O. 2, 335 O.C. 995-91, 1991 G.O. 2, 2913 O.C. 807-92, 1992 G.O. 2, 2869 O.C. 213-93, 1993 G.O. 2, 1757 O.C. 329-94, 1994 G.O. 2, 1335 O.C. 1279-98, 1998 G.O. 2, 4259 O.C. 1413-98, 1998 G.O. 2, 4441 O.C. 459-99, 1999 G.O. 2, 1115 O.C. 35-2001, 2001 G.O. 2, 1035 O.C. 885-2001, 2001 G.O. 2, 3888 O.C. 873-2003, 2003 G.O. 2, 2729 O.C. 119-2008, 2008 G.O. 2, 682 O.C. 425-2010, 2010 G.O. 2, 1313 O.C. 391-2011, 2011 G.O. 2, 970 O.C. 393-2011, 2011 G.O. 2, 975 O.C. 476-2013, 2013 G.O. 2, 1255 O.C. 606-2014, 2014 G.O. 2, 1413 O.C. 57-2015, 2015 G.O. 2, 94 S.Q. 2015, c. 13 O.C. 428-2015, 2015 G.O. 2, 1001 O.C. 513-2015, 2015 G.O. 2, 1086 O.C. 1005-2015, 2015 G.O. 2, 3023 O.C. 1078-2015, 2015 G.O. 2, 3277 O.C. 1186-2015, 2015 G.O. 2, 3477 S.Q. 2015, c. 15, s. 237 S.Q. 2016, c. 25, s. 45 S.O. 2018, c. 19, s. 19 O.C. 640-2019, 2019 G.O. 2, 1094 S.Q. 2021, c. 27, s. 267 O.C. 805-2020, 2020 G.O. 2, 2007 O.C. 820-2023, 2023 G.O. 2 923 O.C. 890-2020, 2020 G.O. 2, 2329 O.C. 781-2021, 2021 G.O. 2, 1676 O.C. 483-2021, 2021 G.O. 2, 1207 O.C. 1112-2023, 2023 G.O. 2, 1776 O.C. 48-2022, 2022 G.O. 2, 187 O.C. 1393-2024, 2024 G.O. 2, 3562 O.C. 645-2022, 2022 G.O. 2, 1057 O.C. 63-2025, 2025 G.O. 2, 555. O.C. 646-2022, 2022 G.O. 2, 1059

Analytical Index

INTRODUCTION

The analytical index enables the user to find, by means of a descriptor or keyword, a concept developed in matters of legislation or regulation.

The analytical index consists essentially of two elements: a) the descriptor and b) the reference.

THE DESCRIPTOR

Descriptors are classified in alphabetical order and are organized by levels.

The main descriptor is complemented by secondary descriptors of 1st and 2nd level; each level clarifies the preceding one.

Exemple:

SUPERIOR COURT: Main descriptor

Appeal : Secondary descriptor 1st

Delay : Secondary descriptor 2nd

The descriptor's function is not to inform on the content of the concept but to indicate its localization. Therefore, the descriptor has a synthetized form so as to make the index easier to consult.

THE REFERENCE

There are two types of reference:

- the reference SEE indicates that a descriptor is rejected to the advantage of another descriptor;

- the reference SEE ALSO acts as a link between descriptors so as to enable the user to carry out a more thorough research in the analytical index.

INDEX

chapter S-2.1, r. 4

SAFETY CODE FOR THE CONSTRUCTION INDUSTRY

ABRASIVE WHEEL Standard 3.10.14

ACCESS TO CONSTRUCTION SITE Authorized personnel 3.1.1

ACT RESPECTING EXPLOSIVES (chapter E-22) 4.2.1; 4.2.9; 4.4.1.2; 4.6.1.1

ACT RESPECTING LABOUR RELATIONS, VOCATIONAL TRAINING AND MANPOWER MANAGEMENT IN THE CONSTRUCTION INDUSTRY (chapter R-20) 2.5.2(2c)

ACT RESPECTING OCCUPATIONAL HEALTH AND SAFETY (chapter S-2.1) 2.1.1; 2.5.4(3d); 4.1.10; 4.2.9; 4.2.10

ACT RESPECTING PRESSURE VESSELS (chapter A-20.01) 3.13.5

AERIAL PLATFORM APPARATUS Hazard signs 3.2.5

AIR LINE Retake 3.11.9 Underground 8.5.8

AIR LOCK See Work in compressed air

AIR SUPPLY Pneumatic equipment 8.8.1 Sandblasting 3.20.3 Underground work site 8.3.1; 8.3.4; 8.5.3

AMERICAN SOCIETY FOR TESTING AND MATERIALS 1.1(3)

AMMONIUM NITRATE AND FUEL OIL See ANFO (AMMONIUM NITRATE AND FUEL OIL)

ANCHORING DEVICE Plan 2.4.1(2j)

ANCHORAGE SYSTEM 2.10.15

ANFO (AMMONIUM NITRATE AND FUEL OIL) 4.6.1.7; 4.6.9.1

ANTI-SLIP SUBSTANCE Scaffolding 3.9.14(1c) Traffic lane 3.2.4(c)

APPLICATION OF THE CODE 2.1.1

ASBESTOS Definition 1.1(1.2); 3.23.0.1

ASBESTOS DUST Definition 1.1(29.1) See also Work liable to produce asbestos dust emissions

ASSISTANT SHOT-FIRER See Shot-firer

ASTM Definition 1.1(3)

ATTESTATION Ability to drive 3.10.4(2b) Digging and excavation 3.15.3(2c) Modification to construction equipment 3.10.1(h) Transmission by engineer 2.4.1(4)

AUTOMATIC RESET BACKUP ALARM 3.10.12

AUTOMATIC SPRINKLER VALVE 3.4.1

BACKHOE LOADER 3.10.3.3

BACKHOE LOADER FOR THE INSTALLATION OF POSTS 3.10.3.4

BACKUP AREA Backup maneuver 2.8.5 Definition 1.1(1.3) Responsabilities of the principal contractor 2.8.1

BARRIER Demolition 3.18.1(7)Excavation 3.15.5 Hoisting apparatus 3.10.9(4) BENGAL FLARE Vehicle 2.18.4(2d) BLASTHOLE Definition 1.1(34.1)Firing 4.7.1 Loading of blastholes 4.6.1.1; 4.6.18 Untamping and repriming a blasthole or misfire 4.6.18; 4.6.9.1 See also Explosive BLASTING AREA Definition 1.1(38)BLASTING ACCESSORY Definition 1.1(1)BLASTING MACHINE 4.6.14 BLASTING MAT Definition 1.1(28)BLEND Explosive 4.6.16 BOATSWAIN'S CHAIR Definition 1.1(31.1)Hazard signal 3.2.5 Minimum age 3.9.17.(1a) Scaffolding 3.9.17(1) Standard 3.9.17(2) BOLTING Definition 1.1(7)Underground work site 8.2.10 BOX Electricity 2.11.8 Explosive 4.1.1; 4.4.7; 4.4.9 BRACING Concrete formwork 6.7.1-6.7.4 BRAKE Hoisting apparatus 2.15.1(d); 3.10.9(1) BRANCH CIRCUIT Switch 2.11.8

BRICK Piling 3.2.2(5) Scaffolding 3.9.21 BUILDING Damage Safety measure 2.14.1 Underground work site 8.1.1; 8.1.7 BUILDING ACT (chapter B-1.1) 2.4.2 CABIN Self-propelled equipment 3.10.10 CAISSON Protection of worker 3.15.7 CAN/CSA (CANADIAN STANDARDS ASSOCIATION) Definition 1.1(7.0)CANADIAN STANDARDS ASSOCIATION See CSA (CANADIAN STANDARDS ASSOCIATION) CANADIAN WELDING BUREAU 3.7.1(g) CARBON MONOXIDE Underground work site 8.3.2; 8.3.12 CARRIER Explosive 8.4.5 Underground work site 8.8.5 CATERPILLAR Bulldozer 3.10.3(6) CLAMP Scaffolding assemblies 3.9.9(c) Use 2.10.2(b) CLOTHING Adjustment 2.10.2(a) Change of 8.9.1; Quarter to change 8.9.1 COMMISSION DES NORMES, DE L'ÉQUITÉ, DE LA SANTÉ ET DE LA SÉCURITÉ DU TRAVAIL Power line 5.2.2 Shot-firer's examination 4.2.3(c); Suspension of shot-firer's certificate 4.2.9 COMPETITION On job site 2.4.2(d)

COMPRESSED AIR See Work in compressed air CONCRETE PUMP Hazard sign 3.2.5 CONCRETING WORK 6.2.1 CONDUIT Digging and excavation 3.15.1 Fire protection 3.4.2 CONFINED SPACE Definition 1.1(17.1)CONNECTION Demolition 3.18.1(3)CONSTRUCTION EOUIPMENT Abrasive wheel 3.10.14 Access 3.10.6 Automatic reset backup alarm 3.10.12 Cabin 3.10.10 Control of 3.10.4(5)Heavy 3.10.3 Internal combustion motor 3.10.17 Lighting 3.10.11 Long load 3.10.19 Material hoisting apparatus 3.10.9 Repair 3.10.18 Safety device 3.10.13 Saw 3.10.15 Utilization 3.10.1; 3.10.4 Vehicle 3.10.2 Warning device 3.10.12 Workers acting as signal persons 3.10.5 CONSTRUCTION SITE Access 3.1.1 Changing room 3.2.10-3.2.13; 3.20.6; 3.23.15; 3.23.16; 8.9.1 Clothing 3.23.15(4.1)(7) Digging and excavation 3.15.1-3.15.10 Employer's obligation 2.4.1(6); 2.4.2; 3.21.1-3.22.1 Fire protection 3.4.1-3.4.6 Guardrail 3.8.2-3.8.5 Glove bag 3.23.15(9.2) Handling and storage of material Bagged material 3.16.4 Bound object 3.16.5 Building lumber 3.16.9

Hand lifting and carrying 3.16.3-3.16.7 Hazardous chemical substance 3.16.8 Masonry 3.16.10 Piling 3.16.2 Pipe 3.16.3 Safety 3.16.1 Hazard sign 3.2.5 High risk 1.1(8) Housekeeping of site Drinking water 3.2.6 Waste material 3.2.2 Inspector Communication of findings 2.2.4 Ladder 3.5.1-3.5.9 Lunch room 3.2.9 Maintenance and arrangement of the work 3.2.4 Major 1.1(8.1) Opening 2.4.1 Platform 3.7.1; 3.7.3 Protruding nail 3.2.3 Ramp 3.7.1-3.7.2 Runway 3.7.1; 3.7.3 Safety officer 2.5.3-2.5.4 Showers 3.2.13-3.2.15; 3.20.6; 8.9.1 Sink 3.2.8 Site committee 2.5.1-2.5.2 Stairs 3.6.1-3.6.4 Steam supply Insulation 3.12.5 Lines 3.12.2-3.12.3 Pile driving engine 3.12.4 Safety 3.12.1 Stepladder 3.5.1-3.5.9 Storage 3.2.1 Supply of gas under pressure 3.13.1-3.13.5 Temporary heating Compliance of device 3.11.7(a) Discharge gases 3.11.7(b) Electric heaters 3.11.5 Gasoline or naptha 3.11.2 3.11.3; 3.11.6 Oil Retake ducts 3.11.9 Use 3.11.1; 3.11.4 Temporary structure 3.3.1 Tie rods for concrete formwork protuding 3.2.3 Toilet facility 3.2.7 Traffic control Backup maneuver 2.8.5 Responsabilities of the principal contractor 2.8.1Signal persons 2.8.3; 2.8.4 Transportation of workers 2.18.1-2.18.7

Welding and cutting 3.14.1-3.14.4 Protection screen 3.14.4 Workers acting as signal persons 3.10.5 See also Supply of gas under pressure; Construction equipment; Demolition; Excavation and trench; Personal protective equipment; Scaffolding; Underground work site; Work in compressed air; Work in confined air; Work in isolated location; Work on road CONTAINING ASBESTOS Definition 3.23.0.1 CONTAMINANT See Dangerous substance CONTRACTOR See Employer CONTROLLED PRODUCT Storage, handling and use 3.16.8 CONVEYOR Safety measures 3.16.3-3.16.7 COPY Compliance certificate 2.4.1(4)Notice of correction 2.2.4 COVERED SIDEWALK Demolition 3.18.1(7) Public safety 2.7.1-2.7.2 CRANE Load-rating plate 2.15.5 Mobile 2.15.7.2; 2.15.7.2.2 Hazard sign 3.2.5 Log book 2.15.7.2; Ann. 9 With a luffing boom 2.15.7.3 With a telescopic boom 2.15.7.2.1 Overhead travelling 2.15.7.5 Plan 2.4.1(3) Tower 2.15.7.4 Log book 2.15.7.4; Ann. 10 CSA (CANADIAN STANDARDS ASSOCIATION) Definition 1.1(1.1)CUTTING See Welding and cutting DANGER Hazard sign 3.2.5

DANGEROUS SUBSTANCE Explosive 4.4.3 Safety measure 2.4.1(6d) Simultaneous transportation of persons 2.18.7(1)Storage and handling 3.16.8 DEBRIS Demolition 3.18.2 DEFINITION 1.1; 2.15.0.1 DEMOLITION Approval 3.18.1(1); 3.18.1(2) Bracing 3.18.1(4); 3.18.1(5) Cut off of supply mains 3.18.1(3) Mechanical method 3.18.3 Minimum age 3.18.1(2) Notice to chief inspector 3.18.1(1)Notice to the Commission 2.4.1(1.3)Protective equipment 3.18.1(8) Protective wall 3.18.3 Public 3.18.1(8) Removal of glass 3.18.1(6) Requirements during 3.18.2 Safety area 3.18.3 Sign 3.18.1(10) Site returned to original state 3.18.5 DEPARTMENT OF ENERGY, MINES AND **RESOURCES (CANADA)** 8.3.1(c); 8.4.2 DEPOT Definition 1.1(13) Explosive 4.4.1; 8.4.9 DERRICK Hazard sign 3.2.5 DETONATING FUSE Storage 4.4.5 Utilization 4.6.15 DETONATING RELAY Storage 4.4.4 DETONATOR Compartment wall 4.3.3 Explosive 4.1.8 Storage 4.4.4 DIGGING 1.1(12.1)

DIGGING AND EXCAVATION Adjoining structure 3.15.2 Banks 3.15.3 Barrier 3.15.5 Caisson 3.15.7 Conduit 3.15.1 Definition 1.1(18); 1.1(34) Drainage 3.15.6 Drilling machine 3.15.9 Means of access and supervision 3.15.4 Minimum age 3.15.10 Pile 3.15.7 Shaft 3.15.8 Underground service 3.15.1 DIKE Definition 1.1(25.01)DISMANTLING Concrete form 6.9.1-6.9.2 Scaffolding 3.9.4 DRAINAGE Excavation or trench 3.15.6 DRILL-BIT 8.5.2; 8.5.6 DRILLING Bit 8.5.2; 8.5.6 Explosive 4.5.1-4.5.6 Machine 3.15.9 DRIVING PERMIT Mandatory 3.10.4(2a) DRUM Piling 3.16.5 EARTH-MOVING MACHINERY USED FOR INSTALLATION POSTS 3.10.3.4 EARTH-MOVING MACHINERY USED FOR LIFTING PURPOSES 3.10.3.3 EATING Application of code 2.1.1 ELECTRIC BLASTING 4.6.12-4.6.13; ELECTRIC HEATER Standard 3.11.5 Compliance with standard 3.11.7(a)

ELECTRIC LIGHTING Work on roads open to traffic 10.4.2 ELECTRIC TOOL 2.11.1; 2.11.2 ELECTRICAL APPLIANCE 2.11.1; 2.11.2 ELECTRICAL LINE Approach distance 5.2.1-5.2.3 Emergency repair 2.4.1(1.2)Exception 5.1.2 Voltage 5.2.4 Warning sign 5.3.1 ELECTRICAL POWER COMPANY 1.1(7); 5.2.2 ELECTRICAL WIRE Extension 2.11.3-2.11.6 ELECTRICITY 2.11-2.11.11 ELEVATOR Plan 2.4.1(3) ELEVATOR SHAFT Ladder 3.5.6.1 EMERGENCY Evacuation 8.12.1 Lighting 8.10.4 Repair 2.4.1(1.2) Rescue 8.12.1-8.12.2 Signal 8.11.9 EMISSION CONTROL DEVICE Underground work site 8.3.7(b) EMPLOYER Obligation 2.4.2 Demolition area 3.18.3 Electrical line 5.3.1 Explosive 4.1.7; 4.3.7; 4.7.6 Explosive actuated tool 7.4.1 General 2.4.1(6) Plan 2.4.1(2)-2.4.1(5) Power line 5.2.2 Scaffolding 3.9.14(1)Surveillance of vehicle 3.10.6(2) Underground work site 8.2.1-8.2.2; 8.3.4 Work in confined air 3.21.1-3.21.5 Work in isolated location 3.22.1

EMPLOYER'S OBLIGATION 7.4.1 ENGINEER Bracing 6.7.1(3)Concrete formwork 6.1.3(8); 6.2.1(a); 6.9.1Demolition 3.18.1(1)Digging and excavation 3.15.3(2c)Modification to equipment 3.10.1(h) Plan 2.4.1(2)Work in the proximity of electrical lines 5.2.2 EQUIVALENCE OF MATERIALS 2.3.1 **EVACUATION** Area of great depth 2.4.1(6d) Underground site 8.12.1 **EXAMINATION** Scaffolding 3.9.12 **EXPLOSION** Warning signal 8.11.9 EXPLOSIVE Abandonment 4.1.6 Bulk 4.6.17 Compartment wall 4.3.3 Container 4.1.1 Definition 1.1(18.1) Deteriorated 4.1.4 Drilling 4.5.1-4.5.3 Electrical defect 4.8.4 Firing Blasthole 4.7.1 Blasting logbook 4.7.10 Blasting machine 4.7.3 Blasting 4.7.5; 4.7.5.1 Electrical supply 4.7.4 Fireworks 4.1.10 Guard 4.7.8 Ignition cords 4.1.10 Military devices 4.1.10 Portable generator 4.7.4 Shelter 4.7.6 Signal 4.7.6 Loading Blasting machine 4.6.14 Blends 4.6.16 Detonation fuse 4.6.15 Electric blasting 4.6.12-4.6.13 Employer's obligations 4.3.7 Priming 4.6.2 Tamping rod and punch 4.6.8

Thunderstorm 4.6.11 Untamping 4.6.18; 4.6.9.1 Low temperature 4.1.5 Misfire 1.1(34.2); 4.9.1-4.9.3 Oldest used first 4.1.3 Power line 4.1.9 Protection 4.1.2 Shelters 4.7.6 Smoking 4.1.7 Storage Case 4.4.7; 4.4.9 Depot 4.4.1; 4.4.1.1; 4.4.7; 4.4.10 Empty container 4.4.9 Other substance 4.4.3; 4.4.4 Sign 4.4.1 Truck 4.4.1.2; 4.4.1.3 Transportation Authorized passenger 4.3.2; 4.3.9 Fueling 4.3.8 Loading 4.3.7 Radio-transmitter 4.3.10-4.3.12 Stops 4.3.8 Underground work site 8.4.3-8.4.6 Vehicle 4.3.1; 4.3.1.1; 4.3.5; 4.3.7-4.3.10; 4.3.13 Underground work sites Authorized class 8.4.2 Storage 8.4.7 Transportation 8.4.3-8.4.6 Unloading of 4.3.7 Waiting period 4.8.1-4.8.4 See also Shot-firer EXPLOSIVES ACT (R.S.C. 1985, c. E-17) 4.6.16 EXPLOSIVE ACTUATED TOOL Definition 1.1(29) EXTENSION CORD Electricity 2.11.3-2.11.6 EXTINGUISHER All-terrain vehicle 3.10.2.1 Explosive transportation 4.3.5 Portable 3.4.1; 3.4.3-3.4.6 Standard 3.4.4 Training required Driver of a self-propelled vehicle 8.1.6 Underground work site 8.1.1(2a); 8.1.6 Vehicle 2.18.4(2e) EYE

Protection 2.10.5

FACE Definition 1.1(22)Protection 2.10.5 FALL Installation of guard-rails 2.9.2 Safety measures 2.9.1 Safety net 2.9.3 Scaffolding 3.9.4(5) FALL ARREST CONNECTING DEVICE Definition 1.1(24.0)FALL ARRESTING DEVICE Standard 1.1.(7.1); 2.10.12 See also Safety harness FALL PROTECTION 2.9-2.9.3 FEET Protection 2.10.6 FIRE Protection 3.4.1-3.4.6 Signal 8.11.9 Underground work site Building 8.1.1; 8.1.7 Extinguisher 8.1.1(2a); 8.1.6 Flammable substance 8.1.2-8.1.5 FIRING INCIDENT Explosive actuated tool Precaution 7.3.3 FIRST AID Kit 2.18.4(2f) FLAG PERSON Work on roads open to traffic 10.3.2 FLAMMABLE LIQUID Reservoir 8.1.2(a) FLAMMABLE SUBSTANCE Underground work site 8.1.2-8.1.5 FLARE Vehicle 2.18.4(2d) FLOOR Demolition 3.18.3 Scaffolding 3.9.8

FLOORS Overload Demolition 3.18.2 FRAMEWORK Strength 2.12.1 FREE FALL DISTANCE Definition 1.1(23.0)FRIABLE MATERIAL Definition 1.1(24.2)FROST Concrete formwork 6.5.6 Explosive 4.1.5 FUEL Underground work site 8.3.9 GAME On construction site 2.4.2(d)GAS Concentration of 8.3.1-8.3.2 Discharge 3.11.7(b) Heating 3.11.3 Purifier 8.3.12 Underground work site 8.1.4; 8.3.2; 8.3.11-8.3.12 Water pipe Using a liquified gas 2.19.1-2.19.3 See also Supply of gas under pressure; Dangerous substance; Gas cylinder GAS CYLINDER Handling 3.16.10 Using 3.13.5-3.13.9 See also Supply of gas under pressure; Gas GAS PIPELINE Emergency repair 2.4.1(1.2)GASOLINE Heating 3.11.2 Underground work site 8.1.4 GLASS Demolition 3.18.1(6) GLOVE Utilization 2.10.10

GREASE

Roadway 3.2.4(b) Underground work site 8.1.5 GUARDRAIL 2.9.2 Construction 3.8.3 Manhole 10.4.3 Metal 3.8.4-3.8.5 Platform, temporary ramp, runway 3.7.1(i) Platform of jumbo 8.5.4(a) Resistance 3.8.2 Suspended scaffolding 3.9.16(1e) HAIR Protection 2.10.2(c) HAND Protection 2.10.10 HANDLING AREA Forbidden 2.16.2 HAZARD SIGNS 1.1(32) HAZARDOUS PRODUCTS INFORMATION REGULATION (chapter S-2.1, r. 8.1) 3.16.10 HAZARDOUS PRODUCTS REGULATIONS (SOR/2015-17) 3.23.13 HEPA 3.23.15; 3.23.16 HIGH-EFFICIENCY FILTER Definition 1.1(21.2)HIGHWAY SAFETY CODE (chapter C-24.2) 2.18.1-2.18.2; 3.10.4(2a); 4.3.1.1; 10.1.1; 10.3.1 HOISTING Accessory 2.15.1 Material 3.10.9; 3.16.3-3.16.7 Scaffolding cable 3.9.13 Signal 8.11.3 HOISTING APPARATUS Definition 1.1(2); 2.15.0.1 Demolition 3.18.2 Digger derrick 2.15.15 Job-site elevator 2.15.14 Mast-climbing transport platform 2.15.18; 2.15.18; 2.15.19; 2.15.20 Material 2.15.11; 3.10.9

Minimum age 2.15.10 Mobile elevating work platform 1.1(29.01); 2.15.16; 2.15.17 Personnel hoisting apparatus 1.1(2.1), 2.15.2 Safety harness $3.1\overline{0.9}(3)$ Scaffolding 3.9.4(3); 3.9.16(c) self-propelled vehicle 1.1(36.1) Supervision 2.15.9 Vehicle-mounted aerial device 2.15.13 HOISTING CABLE Scaffolding 3.9.13 HOISTING OF LOADS Safety measures 2.15.6 HOOD Mandatory 2.10.11 HYDRANT 3.4.1 HYDRAULIC SHOVEL 3.10.3.3; 3.10.3.4 ICE Concrete formwork 6.5.6 Roadway 3.2.4(b) Scaffolding 3.9.14(1c) INQUIRY Inspector 2.2.2 Safety officer 2.5.4(3f) INSPECTOR Accomodation 2.4.2(j) Communication of findings 2.2.4 Concentration of gaz 8.3.2; 8.3.12 Decision appealed from 2.2.3 Temporary work 2.2.6 INTERNAL COMBUSTION MOTOR Utilization 3.10.17 INTERPRETATION 4.1.11 ISO Definition 1.1(23.1)JACK Capacity 2.13.1 Concrete formwork 6.6.1; 6.7.4; 6.9.2 Hoisting 2.15.8 See also Lifting jack

© Éditeur officiel du Québec, 2025 Toute reproduction interdite JEWELLERY Prohibition 2.10.2(b) JIB CRANE Hazard sign 3.2.5 JOB SITE COMMITTEE Communication of inspection 2.2.4 Composition 2.5.2(2)Duty 2.5.2(3) Establishment 2.5.1 Meeting 2.5.2(4); 2.5.2(5) JOB SITE ELEVATOR See Elevator JOIST Concrete formwork 6.6.2 JUMBO Definition 1.1(24)Drilling 8.5.4-8.5.5 LADDER Access to a work area 3.5.1-3.5.2 Commercial Standard 3.5.3 Elevator shaft 3.5.6.1 Inclination App. 0.1 Lighting 8.10.3 Maximum length 3.5.5 Platform of jumbo 8.5.4(b) Repair Storage 3.5.9 Site-fabricated Construction 3.5.4 Underground work site 8.7.1-8.7.4; 8.7.7 Utilization 3.5.6; 3.5.8 LAMP Miner 8.10.1 LANDING Scaffolding 3.9.11(a) LANYARD Definition 1.1(12.0)Fall arresting device standard 2.10.12; 3.10.9(3) LEG PAD

Mandatory 2.10.11

LICENCE Blending of ANFO 4.6.16 LIFE JACKET 2.10.13 See also Work above or near water LIFE LINE Definition 1.1(7.1) Standard 2.10.12 (1d); 3.10.9(3) Suspended scaffolding 3.9.16(1f) LIFTING See Hoisting LIFTING JACK Positive stop 2.15.8(2)LIGHTING Traffic lane 3.2.4(f)Underground work site 8.10.1-8.10.4; 8.11.9 Vehicle 3.10.11 LOAD Handling 2.15.6 Load-bearing part 1.1(28.1) Storage 3.16.1(2) See also Long load; Rated load LOADER 3.10.3.1 LOADER FOR LIFTING PURPOSES 3.10.3.3 LOADER USED FOR HANDLING PURPOSES 3.10.3.2 LOADING AREA Definition 1.1(37) LOCKOUT AND OTHER ENERGY CONTROL METHODS Access 2.20.3 Application 2.20.2; 2.20.14 2.20.1 Definitions Coordination of measures 2.20.10 Cut of the lock 2.20.12 Equivalent safety of a method Analyze 2.20.4 Information 2.20.8 Material 2.20.11 Procedures 2.20.5; 2.20.6; 2.20.7 Record 2.20.11 Specific control mode 2.20.13 Written authorization 2.20.9

© Éditeur officiel du Québec, 2025 Toute reproduction interdite LOCOMOTIVE Explosive 8.4.5 Underground work site 8.8.3 LODGING Application of Code 2.1.1 LOG BOOK Mobile crane 2.15.7.2; App. 9 Tower crane 2.15.7.4; App. 10 LONG LOAD 3.10.19 LOWER EXPLOSIVE LIMIT Explosive actuated tool 7.3.4 LUMBER Piling 3.16.9 Stacking of waste material 3.2.2(5)Underground work site 8.1.3 LUNCH ROOM Employer's obligation 3.2.9 MANHOLE Protection 10.4.3 MANUFACTURER Definition 1.1(20)MASONRY Piling of materials 3.16.10 MAST-CLIMBING WORK PLATFORM 1.1(14.2) Plans 2.4.1(2.m) Standards 3.9.22 MATERIAL HOIST Plan 2.4.1(3)MATERIALS Equivalence 2.3.1-2.3.2 Non-combustible 1.1(25) MEAL Construction site 3.2.9 MECHANICAL SHOVEL Demolition 3.18.3 MEDICAL BRACELET Use 2.10.2(b)

MINIMUM AGE Boatswain's chair 3.9.17(1a) Demolition 3.18.1(12) Digging and excavation 3.15.10 Explosive actuated tool 7.2.2 Hoisting apparatus 2.15.10 Shot-firer 4.2.3 Suspended scaffolding 3.9.16 Underground work 8.13.1

MISFIRE Definition 1.1(34.1) Re-firing 4.9.2 Shot-firer's duties 4.9.1

MITTEN Mandatory 2.10.10

MOTOR VEHICLE SAFETY ACT 8.3.7

MOTORIZED SCAFFOLDING 1.1(15.01) Standards 3.9.25

NAIL Protruding 3.2.3 Scaffolding assembly 3.9.9(a)

NATIONAL FIRE PROTECTION ASSOCIATION See NFPA (NATIONAL FIRE PROTECTION ASSOCIATION)

NATIONAL INSTITUTE FOR OCCUPATIONAL SAFETY AND HEALTH See NIOSH (NATIONAL INSTITUTE FOR OCCUPATIONAL SAFETY AND HEALTH) NFPA (NATIONAL FIRE PROTECTION ASSOCIATION) 1.1(26)

NIOSH (NATIONAL INSTITUTE FOR OCCUPATIONAL SAFETY AND HEALTH) 8.3.1; 3.23.15

NITROCARBONITRATE See ANFO (AMMONIUM NITRATE AND FUEL OIL)

NITROGEN DIOXIDE Underground work site 8.3.2; 8.3.12

NITROGLYCERINE See Dynamite

NOISE Calculator 1.1(7.0.0) Daily noise exposure level 1.1(26.1) DBA 1.1(13.2) 1.1(13.3) DBC Equivalent continuous sound pressure level (dBA) 1.1(26.2) Impulse noise 1.1(3.1)Noise 2.21-2.21.14 Peak sound pressure level 1.1(29.2) NON-COMBUSTIBLE CONSTRUCTION Definition 1.1(11) NOTICE Concrete formwork 6.3.1(d) Correction 2.2.4 Defect 2.5.4(3d) Demolition 3.18.1(1)Emergency repair 2.4.1(1.2)Explosive actuated tool 7.6.1 Opening and closing of a site 2.4.1(1)OBSTRUCTION Demolition 3.18.5(b) Roadway 3.2.4(a) Underground work site 8.3.5.1; 8.7.9(d) OIL Heating 3.11.3 Roadway 3.2.4(b) Underground work site 8.1.5 OIL PIPELINE 3.10.3(8) **OPERATOR** Heavy equipment 3.10.3(5); 3.10.4 OPERATOR'S CERTIFICATE Explosive actuated tool 7.2.1; Schedule 8 ORDRE DES INGÉNIEURS DU QUÉBEC 1.1(23)OUTRIGGER SCAFFOLDING 3.9.15 **OVERSLEEVE** Mandatory 2.10.11 **OXY-CATALYST EXHAUST PURIFIER** Underground work site 8.3.12

OXYGEN Use 3.16.8(5) OXYGEN CYLINDER Storage 3.13.4 PAPER Underground work site 8.1.3 PARKING Underground work site 8.8.8 PEDESTRIAN Protection 2.7.1 PERSONAL FLOATATION DEVICE 2.10.13 See also Work above or near water PERSONAL PROTECTIVE EOUIPMENT Against sandblasting 3.20.1; 3.20.3 Eye 2.10.5 Face 2.10.5 Feet 2.10.6 Guardrail 2.9.2 Hand 2.10.10 Head 2.10.3-3.10.4 Life jacket 2.10.13 Mandatory 2.10.11 Other part than the hand 2.10.11 Respiratory passage 2.10.8 Safety belt 2.10.14 Safety net 2.9.3 Utilization 2.10.2 PESTICIDE Simultaneous transportation of workers 2.18.7(1) PILE Head 2.16.4 Protection of workers 3.15.7 Solidity 2.16.1 PILE-HAMMER 2.16.3 PILING Bagged material 3.16.4 Bound object 3.16.5 Lumber 3.16.9 Masonry material 3.16.10 Method 3.16.2 Pipe 3.16.3

PIPE Compressed air 8.5.3-8.5.7 Fire 3.4.6 Piling 3.16.3 Steam 3.12.4 Underground work site 8.1.1(2b) PIPELINE Heavy equipment 3.10.3(8) PLAN Shoring drawing 6.1.1-6.1.3 Transmission 2.4.1(2) PLATFORM Jumbo 8.5.4 Plan 2.4.1(2g); 2.4.1(2h) Suspensed scaffolding 3.9.16(1d) Temporary structure 3.3.1 Use and construction 3.7.1; 3.7.3 PORTABLE GENERATOR Authorization for firing 4.7.4 PORTABLE TOOLS Explosive actuated tool 7.1.1.1-7.1.1.16 General 7.0-7.0.8 Nailing gun 7.1.2.1-7.1.2.4 Saws 7.1.3.1.1-7.1.3.2.6 POST Concrete formwork 6.6.2 Hoisting of material 3.10.9(1)Scaffolding 3.9.15 POSTING Standardized signal 8.11.5 PRINCIPAL CONTRACTOR Findings of inspection 2.2.4 Obligations 2.8.1 Opening and closing of site 2.4.1(1)See also Employer PROPANE GAS CYLINDER Storage 3.13.10 PROPPING Concrete formwork Bearing surface 6.5.1-6.5.7 Dismantling 6.9.1-6.9.2 Exception 6.3.1

Horizontal bracing 6.7.1-6.7.4 Load 6.4.1 Multistory structure 6.8.1 Plan 2.4.1(2b); 6.1.1-6.1.3 Preparation 6.2.1 Support part 6.6.1-6.6.2 Damaged building 2.14.1 Demolition 3.18.1(4); 3.18.1(5) See also Shoring PROTECTION AGAINST FIRE 2.4.1(1.3) PROTECTION OF WORKERS See Safety PROTECTIVE APRON 2.10.11 PROTECTIVE CLOTHING 3.23.1.1 PROTECTIVE SCREEN Welding and cutting 3.14.4 PROTECTIVE STRUCTURE Heavy equipment 3.10.3(3)-3.10.3(5); 3.10.10 PROTECTIVE WALL 1.1(25.1) Demolition 3.18.3; 3.18.5 Protection of public 2.7.1-2.7.2 PUMP JACK SCAFFOLD 1.1(14.1) Standards 3.9.23 RAILROAD 8.8.4 RAMP Temporary structure 3.3.1 Utilization and construction 3.7.1-3.7.2 RATED LOAD Definition 1.1(10) Employer's obligation 2.15.3(a)Hoisting apparatus 3.10.9(1) Lifting jack 2.15.8 Plate 2.15.5 Posting 2.15.2 See also Load RECIPIENT Hazardous substance 3.16.8(3) RECORD Of shots 4.7.10 Work in underground site 8.12.3

RECREATIONAL FACILITY Application of the Code 2.1.1

REFRIGERATION Underground work site 8.3.12

REGULATION RESPECTING OCCUPATIONAL HEALTH AND SAFETY IN MINES (chapter S-2.1, r. 14) 8.6.2

REGULATION UNDER THE ACT RESPECTING EXPLOSIVES (chapter E-22, r. 1) 4.3.1; 4.3.7; 4.4.1; 4.4.2

REGULATION RESPECTING THE PROFESSIONAL QUALIFICATION OF CONTRACTORS AND OWNER-BUILDERS (chapter B-1.1, r. 9) 2.4.2

REGULATION RESPECTING OCCUPATIONAL HEALTH AND SAFETY (chapter S-2.1, r. 13) 2.10.8; 3.2.10; 3.2.14; 3.10.17; 3.21.2(c) 3.21.3(b); 3.23.0.1; 3.23.16(4); 4.1.10; 8.3.1

REPAIR Emergency 2.4.1(1.2) Raised part blocked 3.10.18

RESCUE Equipment 2.10.13 Underground work site 8.12.1-8.12.2

RESISTANCE Guardrail 3.8.2

RESPIRABLE ASBESTOS FIBRE Definition 1.1(21.1)

RESPIRATOR 2.10.8; 3.15.9; 3.21.2; 3.23.5; 3.23.14.1; 3.23.15; 3.23.16; 3.23.16.1; 8.12.2

RESPIRATORY PASSAGE Protection 2.10.8

REVERSE GEAR Automatic reset backup alarm 3.10.12

ROLL OF PAPER Piling 3.16.5(2)

ROLLER Utilization 3.10.3(7)

ROOF Vehicle 2.18.4(2c) RUNWAY Temporary structure 3.3.1 Use and construction 3.7.1; 3.7.3 SAE Definition 1.1(31.01)SAFETY Device 3.10.13 Employer's obligation 2.4.1(6); 2.4.2 Equivalence of materials 2.3.2 See also Personal protective equipment SAFETY BELT Heavy equipment 3.10.3(5) Standards 2.10.14 Utilization 2.10.14 SAFETY COURSE Employer's obligation 2.4.2(i) SAFETY DEVICE 3.10.13 SAFETY FACTOR Definition 1.1(21)SAFETY FUSE Explosive 4.6.2 SAFETY HARNESS 2.10.12 Standard 3.10.9(3) SAFETY HAT Mandatory 2.10.3 SAFETY NET 2.9.3 SAFETY OFFICER Duty 2.5.4 Number 2.5.3 SAFETY ORDER 2.5.4(3e) SAFETY REPRESENTATIVE Communication of an inspector 2.2.4 SALUBRITY Employer's obligation 2.4.1(6)

SAND

Traffic lane 3.2.4(c) SAND BLASTING Maintenance of equipment 3.20.4 Personal protective equipment 3.20.1 Air flow 3.20.3 Ventilation 3.20.5 SAND LAYER Demolition 3.18.2 SAW 3.10.15 SCAFFOLDING Access 3.9.11 Anti-slip substance 3.9.14(1c) Assembly 3.9.9 Boatswain's chair 3.9.17 Bricklayer 3.9.21 Concrete form 6.6.1; 6.7.3 Danger sign 3.2.5 Demolition 3.18.2 Erection and dismantling 3.9.4 Falls 3.9.4(5) Hoisting cable 3.9.13 Inspection 3.9.12 Ladder jack 3.9.18 Material 3.9.3 Metal 3.9.20 Minimum age 3.9.16 Mobile 3.9.19 Mooring 3.9.10 Outrigger 3.9.15 Plan 2.4.1(2i); 2.4.1(2d)-2.4.1(2f) Platform 3.9.8 Putlog 3.9.7 Reformed piece 3.9.14(4) Shoring 1.1(15); 2.13.3; 2.13.3.1 Standards 3.9.15-3.9.25 Stringer 3.9.16 Suspended 3.9.16 Temporary structure 3.3.1 Upright 3.9.5 Utilization 3.9.1-3.9.2; 3.9.14 See also Motorized scaffolding; Outrigger scaffolding; Winch scaffolding SCALING

Definition 1.1(31) Underground work site 8.2.4-8.2.6 SEAT Vehicle 2.18.4(2a); 2.18.4(3) SELF-PROPELLED VEHICLE 3.10.2 SEMI-TRAILER Transportation of persons 2.18.6 SEWER Emergency repair 2.4.1(1.2)SHAFT Definition 1.1(30)Excavation work 3.15.8(2) Explosive 8.4.4 Lighting 8.10.3 Shoring of bank 3.15.8(1)Signal 8.11.1-8.11.3; 8.11.6-8.11.8 Sinking of 8.11.1-8.11.2; 8.11.7 Stair and ladder 8.7.1-8.7.3 Transportation 8.6.1-8.6.2 SHELTER Underground work site 8.7.8-8.7.9 SHORING Caisson and pile 3.15.7 Damaged building 2.14.1 Digging and excavation 3.15.3 Plan 2.4.1(2a) Scaffold 2.13.3.1 Shaft 3.15.8 Underground work site 8.2.3; 8.2.9 See also Propping SHORING SUPPORT 8.2.7-8.2.8 SHOT App. 2.2 Blasting logbook 4.7.10 See also Explosive SHOT-FIRER Blasting area 4.8.1; 4.8.2 Candidate 4.2.3 Certificate 4.2.1; ; 4.2.1.1; 4.2.9; 4.2.10; 4.2.11 Checking of circuit 4.8.4 Examination 4.2.3; Helper 4.2.2 SIDEWALK Demolition 3.18.1(7) Protection of public 2.7.1

See also Covered sidewalk

© Éditeur officiel du Québec, 2025 Toute reproduction interdite SIGNAL Demolition 3.18.1(10)Emergency 8.11.9 Explosive 4.7.6 Hazard 3.2.5 Legible copy 8.11.5 Method App. 5.1 Shaft 8.11.1-8.11.3; 8.11.6-8.11.8 Explosive sign 4.3.8 See also Warning light SIGNAL PERSON 2.8.3; 2.8.4; 2.8.5 SILL Concrete formwork 6.5.2-6.5.3 SINK Construction site 3.2.8 SKIDDER 3.10.2 SLIPPING Prevention 3.2.4(c)SNOW Concrete formwork 6.5.6 Roadway 3.2.4(b) Scaffolding 3.9.14(1c) SOCIETY OF AUTOMOTIVE ENGINEERS 1.1(31.01) STAIR Construction 3.6.2 Lighting 8.10.3 Scaffolding 3.9.11(b) Steel frame 3.6.3 Temporary 3.6.4 Underground work site 8.7.1-8.7.3 Utilization 3.6.1 STANDARDS Body Belts and Saddles for Work Positioning and Travel Restraint (CAN/CSA Z259.1) 2.10.14 Canadian Electrical Code, Part 1, 16th Edition (CSA Standard C22.10-1992) 3.14.6 Certification of Companies for Fusion Welding of Steel (CSA W47.1) 3.9.24 Certification of Welding Inspection Organizations (CSA Standard W178.1) 1.1 (26.1); 3.9.24; 3.9.25 Code for Tower Cranes (CSA Z248-1975) 2.15.7.4

Code for the Engineering Design of Wood (DCA 086-1976) 6.4.2 Eye and Face Protectors (CAN/CSA Z94.3) 2.10.5 Flammable and Combustible Liquids Code (NFDA 30) 8.1.2 Full Body Harness (CAN/CSA Z259.10) 2.10.12; General Purpose Electric Overhead Travelling cranes (CSA B167-1964) 2.15.7.5 Heavy equipment 3.10.3 Industrial protective headwear — Performance, selection, care, and use CAN/CSA Standard 2.10.3Z94.1 Installation Code for Oil Burning Equipment (CSA B139-1971) 3.11.6 ISO Standard 6165:2012, Earth-moving machinery – Basic types – Identification and terms and definitions 3.10.12 ISO Standard 9533:2010, Earth-moving machinery - Machine-mounted audible travel alarms and forward horns - Test methods and performance criteria 3.10.12 Low lift and High Lift Trucks (CSA B335.1-1977) 2.15.7.6 Mast-climbing Work Platforms (ANSI/SIA A 92.9-1993) 3.9.24 Mast-climbing Work Platforms (ISO 16369) 3.9.25 Minimum Performance Criteria for Falling object Protective Structures(FOPS) (SAE J23101-1976) 3.10.3(3)Natural Gas Installation Code (CAN/CGA Standard B.149.1-M91) 3.11.7 NF EN 1.1(25.2) Non-destructive examination 1.1 (17.2) CAN/CGSB-48.9712 Standard, Non-Destructive Testing-Qualification and Certification of Personnel Portable Fire Extinguishers (NFPA 10-1975) 3.4.4: 8.1.1 Portable Fire Extinguishers (NFPA 10-1988) 3.4.4; 8.1.1; 8.1.6 Portable Ladders (CSA Z-11-1969) 3.5.3; 3.5.7 Propane Installation Code (CAN/CGA Standard B.149.2-M91) 3.11.7 Protective Footwear, (CAN/CSA Z-195) 2.10.6 CSA Standard Z94.4-93 Selection, Use, and Care of Respirators 3.23.14.1 SAÊ Standard J994, Alarm – Backup – Electric Laboratory Performance Testing 3.10.12 Safety Code for Mobile Cranes (CSA Z150-1974) 2.15.4: 2.15.7.2 Safety Code for Suspended Powered Platforms (CAN3-Z271-M84)) 3.9.16(2); 3.9.17(2)
Safety harness 2.10.12 (CAN3-Z271-M84)) 3.9.16(2); 3.9.17(2) Safety in Welding, Cutting, and Allied Processes (CAN/CSA Standard W117.2-M87) 3.14.2; 3.14.3 Standard Method of Test for Determination of Non-Combustibility in Building Materials (CAN4-S114) 1.1 (25) Steel Structures for Building Limit States design CAN3 (S16.1-M78) 6.4.2 Steel Structures for Buildings (CSA S16-1969) 6.4.2Steel Wire Rope for General Purpose and for mine Hoisting and Mine Haulage (CSA G4-1976) 3.9.13 STEAM SUPPLY Lines 3.12.2-3.12.3 Pile driving engine 3.12.4 Safety 3.12.1 System 3.12.5 STEEL Concrete formwork 6.4.2 Staking up of waste material 3.2.2(5)STEEL STRUCTURE ERECTION OR DISMANTLING WORK Scope 3.24.1 Means of access 3.24.3 Rescue following a fall 3.24.4 Presence of a first-aider 3.24.5 Protuding parts 3.24.6 Plans and procedures 3.24.7 Telecommunications system 3.24.2 ERECTION OF A STEEL STRUCTURE Preparation of the work area 3.24.8 Erection plans 3.24.9 Erection procedure 3.24.10 Anchor rods 3.24.11 Prior certification: 3.24.12 Anchor rod modification or repair 3.24.13 Stacks of shims 3.24.14 Hoisting procedure 3.24.15 Component weight 3.24.16 Hoist hooks 3.24.17 Beam-column connection of a multi-span steel structure 3.24.18 Erection of an open web steel joist 3.24.19 Space between girts 3.24.20

DISMANTLING OF A STEEL STRUCTURE Obligations of the principal contractor 3.24.21 Dismantling plan 3.24.22 STEPLADDER Standard 3.5.7 STILT 3.19.1 STOP Explosive 4.3.8 STOP BLOCK Underground work site 8.8.5 STORAGE Chemical substance 3.16.8 Construction site 3.2.1 Dangerous substance 3.16.8 Detonating relay 4.4.4 Detonator 4.4.4 Flammable liquid 8.1.2 Propane gas cylinder 3.13.10 Oxygen cylinder 3.13.4 See also Explosive STORAGE'S CASE Explosive actuated tool Content 7.5.1 STRENGTH Concrete formwork 6.4.2 Guardrail 3.8.2 STRINGER Concrete formwork 6.6.2 Scaffolding 3.9.6 **SUPERVISION** Demolition 3.18.1(11) Shot-firer 4.2.2 SUPPLY OF GAS UNDER PRESSURE Gas under pressure 3.13.5-3.13.9 Standard 3.13.5 Using 3.13.6 Piping 3.13.1-3.13.3 Propane gas cylinder 3.13.10 Storage of oxygen cylinder 3.13.4

SUSPENDED SCAFFOLDING Definition 1.1(15.1)Standard 3.9.16(2) SWINGING WEIGHT 3.18.3 SWITCH Electricity 2.11.8 TABLE Lunch room 3.19(e)TAMPING ROD AND PUNCH Loading of explosive 4.6.7 TELEPHONE Shaft 8.11.2 Tunnel 8.11.8 TEMPORARY HEATING See Construction site; Heating TEMPORARY WORK 3.3.1-3.3.4 TIMBER SHORES Quality of timber 2.13.2 Size 2.13.2 Utilization 2.13.2 TOILET FACILITY Construction site 3.2.7 TRAFFIC Control 2.8.1-2.8.5 Non-skid product 3.2.4(c) Underground work site 8.3.5.1; 8.7.8 TRAFFIC SIGN Work on roads open to traffic 10.3.1 TRAILER Transportation of persons 2.18.6 TRAINING Explosive actuated tool 7.2.1; Schedule 8 Extinguisher Driver of a self-propelled vehicle 8.1.6 TRANSPORTATION Detonator 4.3.13 Explosive Authorized person 4.3.2; 4.3.9 Carrier 8.4.5

Extinguisher 4.3.5 Fueling 4.3.8 Locomotive 8.4.5 Radio transmitter 4.3.10-4.3.12 Vehicle 4.3.1-4.3.5; 4.3.13 Loading 4.3.7 Material 3.16.1(1); 3.16.3-3.16.7 Shaft 8.6.1-8.6.2 Stop 4.3.8 Underground work site 8.4.3 Workers Conformity of vehicle 2.18.1; 2.18.4 Driver 2.18.2 Passenger 2.18.3 Simultaneous transportation of materials 2.18.7Trailer and semi-trailer 2.18.6 Truck 2.18.5 TRAVELLING CRANE Standard 2.15.7.5 TRENCH Definition 1.1(34) See also Digging and excavation TRUCK Transportation of workers 2.18.5 TUNNEL Definition 1.1(35) Telephone 8.11.8 **TWO-BLOCKING** Definition 1.1(27)ULC (UNDERWRITER'S LABORATORIES OF CANADA) Definition 1.1(36)Fire extinguishers 3.4.4 UNDERGROUND WORK SITE Air quality Concentration 8.3.2; 8.3.11 Purifier system 8.3.12 Standard 8.3.1 Supply 8.3.1; 8.3.4; 8.5.3 Ventilation 8.3.3 Clothing 8.9.1 Code of signal App. 4 Definition 1.1(9)Drilling

Air Line 8.5.8 Air hose 8.5.3; 8.5.7 Bit 8.5.2; 8.5.6 Jumbo 8.5.4-8.5.5 Position of worker 8.5.1 Emergency 8.12.1-8.12.2 Emission control device 8.3.7(b) Evacuation 8.12.1 Extinguisher 8.1.6 Fire Building 8.1.1; 8.1.7 Extinguisher 8.1.1; 8.1.6; 8.1.7 Flammable liquid 8.1.2-8.1.5 Flammable substance 8.1.2-8.1.5 Fuel 8.3.9 Gas 8.1.4; 8.3.11-8.3.12; 8.3.1.8.3.2 Gasoline 8.1.4 Grease 8.1.5 Ground stability Bolting 8.2.10 Employer's obligation 8.2.1-8.2.2 Scaling 8.2.4-8.2.6 Shoring 8.2.3; 8.2.9 Support 8.2.7-8.2.8 Hygiene 8.9.1; 8.9.4 Ladder 8.7.1-8.7.4; 8.7.7 Lighting 8.10.1-8.10.4 Locomotive 8.8.3 Lumber 8.1.3 Minimum age 8.13.1 Obstacle 8.3.5.1; 8.7.9(d) Oil 8.1.5 Paper 8.1.3 Parking of vehicle 8.8.8 Pipe 8.1.1(2b) Refrigeration 8.3.12 Rescue 8.12.1-8.12.2 Scaling 8.2.4-8.2.6 Shoring 8.2.3; 8.2.9 Signal Copy 8.11.5 Sinking of shaft 8.11.1-8.11.13; 8.11.6-8.11.8 Standard 8.11.4 Stair 8.7.1-8.7.3 Stop block 8.8.5 Traffic Abandoned zone 8.3.5 Clearance space 8.7.5-8.7.6 Fuel 8.3.9 Ladder or stair 8.7.1-8.7.4; 8.7.7 Locomotive 8.8.3 Obstruction 8.3.5.1 Passenger wagon 8.8.6-8.8.7

Railroad 8.8.4 Shelter 8.7.8-8.7.9 Vehicle 8.3.6-8.3.8 Warning system 8.3.10 Traffic lane 8.3.5.1; 8.7.7 Transportation in a shaft 8.6.1-8.62 Vehicle 8.1.6; 8.3.6-8.3.8; 8.8.2 Wagon 8.8.6-8.8.7 Warning device 8.3.10; 8.8.2-8.8.3 Waste material 8.1.3 Water supply 8.1.1(2b) See also Construction site; Explosive UNDERWRITERS LABORATORIES OF CANADA See ULC (UNDERWRITER'S LABORATORIES OF CANADA) UNTAMPING 4.5.9.1; 4.6.18 VALVE Air supply 8.8.1 VAPOR See Dangerous substance VEHICLE Explosive 4.3.1-4.3.5 Transportation of workers 2.18.1-2.18.7 Underground work site 8.1.6; 8.3.6-8.3.8; 8.8.2 Utilization 3.10.2; 3.10.4(2) VENT PIPE Heating 3.11.3 VENTILATION Toilet facility Area of great depth 2.4.1(6d) VOLTAGE Employer's responsability 5.2.4 WAGON Underground work site 8.8.6-8.8.7 WALKWAY Plan 2.4.1(2g) Utilization and construction 3.7.1 WALL Demolition 3.18.3 Protective 2.7.1-2.7.2

WARNING DEVICE Firing of explosive App. 2.4 Hoisting apparatus 2.15.1(c) Mandatory equipment 3.10.12 Underground work site 8.3.10; 8.8.2-8.8.3 WARNING SIGN Electric line 5.3.1 WARNING LINE 2.9.4.0; 2.9.4.1 WASTE MATERIAL Construction site 3.2.2 Demolition 3.18.1(3)Enclosure 3.2.2(2c) Roadway 3.2.4(e) Underground work site 8.1.3 WATER See Underground work WATER PIPE Freezing with a liquified gas 2.19.1-2.19.3 WATER SUPPLY Drinking water 3.2.6 Fire protection 3.4.6 Underground work site 8.1.1(2b) WELDER 3.7.1(g); 3.7.1(h) WELDING AND CUTTING Chassis or hood of welding machine 3.14.6 Dangerous substance 3.14.1; 3.14.3 Portable welding machine 3.14.7 Prohibition 3.14.8 Protective screen 3.14.4 Standard 3.14.2; 3.14.3; 3.14.6 Welding torch 3.14.5 WINCH SCAFFOLDING 1.1(14.3) Standards 3.9.24 WIRE MESH Guardrail 3.8.5; 3.9.16(1e) WORK ABOVE OR NEAR WATER Boat 11.4 Certificate 11.2 Contractor 2.4.4; 11.1 Definition 1.1(34.0.1)

Equipment 11.3 Life buoys 11.5 Life jacket 2.10.13 Personal floatation device 2.10.13 Sound alarm system 11.6 WORK CARRIED OUT OUTSIDE 3.23.1.1 WORK IN CONFINED AIR 3.21.1-3.21.5 Concentration reading 3.21.2-3.21.3 Contaminant Flammable product 3.21.2(b) Other 3.21.2(c); 3.21.2(d)Oxygen 3.21.2(a) Emergency 3.21.6 Prevention from entering 3.21.5 Ventilation 3.21.4 WORK IN ISOLATED LOCATION 3.22.1 WORK LIABLE TO PRODUCE ASBESTOS DUST **EMISSIONS** Application of the Code 3.23.1 Before undertaking work Obligation 3.23.8 Cleaning of the work area 3.23.12, 3.23.12.1 Compressed air 3.23.5 Containing asbestos 3.23.0.1 Debris of materials containing asbestos 3.23.10 Determination of types of asbestos 3.23.3 Drop sheets 3.23.11 Exhaust ventilation system 3.23.15(9.1)(10), 3.23.16.1 Friable materials 3.23.4; 3.23.9 Handing and removing 3.23.16.1 Obligation 3.23.16.1 Glove bag 3.23.15(9.2) Label to receptacle 3.23.13 Moving kilns, boilers or similar devices 3.23.9.1 Protective clothing 3.23.1.1 Protective shoes 3.23.14 Removal materials liable to generate asbestos dust 3.23.3.2 Respirator 3.23.14.1 Sites with high-risk work 3.23.2(3) Employer's obligation 3.23.16 Sites with low-risk work 3.23.2(1) Employer's obligation 3.23.14.1 Sites with moderate-risk work 3.23.2(2) Employer's obligation 3.23.15 Smoking or eating 3.23.6 Training and information program 3.23.7

Use of crocidolite or amosite 3.23.3.1 Wetting agent 3.23.1.1, 3.23.8, 3.23.9; 3.23.10 Work area Smoking, eating, drinking or chewing any substances 3.23.6 Work carried out outside 3.23.1.1

WORK ON ROADS OPEN TO TRAFFIC Application 10.2.1 Electric lighting 10.4.2 Flag person 10.3.2 High-visibility safety apparel 10.4.1 Protection of manholes 10.4.3 Public highway 10.1.1 Road vehicle 10.1.1 Traffic signs 10.3.1

WRECKING See Demolition

PROVISIONS LISTED IN SECTION 2 OF THE REGULATION RESPECTING OCCUPATIONAL HEALTH AND SAFETY AND APPLICABLE TO CONSTRUCTION SITES

Excerpt of Regulation respecting occupational health and safety

chapter S-2.1, r. 13

Act respecting occupational health and safety (chapter S-2.1, s. 223).

2. Scope: Notwithstanding any provisions to the contrary, this Regulation applies to all establishments.

Sections 1 to 5, 17, 40, 42, 44 to 48, 64 and 65, subparagraphs 1 to 3 of the first paragraph and the second paragraph of section 66, sections 107 to 111, 113 to 115, 121 to 124 and 144, the first paragraph of section 145 and sections 148 to 151 and Division XXVI.1 also apply, with the necessary modifications, to construction sites or, if applicable, to categories of sites specified therein.

O.C. 885-2001, s. 2; O.C. 119-2008, s. 8; O.C. 425-2010, s. 1; O.C. 428-2015, s. 8; O.C. 287-2021, s. 1.

Provisions listed in section 2 of the Regulation respecting occupational health and safety

1. Definitions: In this Regulation,

"ACNOR" means the Canadian Standards Association or the Association canadienne de normalisation;

"aerial basket lifting device" means any elevator equipped with an extendable/retractable or jointed arm designed to be fitted with a carrier and used to lift workers or supplies by means of a basket on work sites;

"air recirculation" means local exhaust ventilation by extraction, filtering of the air and redistribution of the filtered air in a work area;

"all-terrain vehicle" means any passenger vehicle designed for sports driving off public highways and whose net weight does not exceed 450 kg;

"ANSI" means the American National Standards Institute;

"asbestos" means the fibrous form of mineral silicates belonging to rock-forming minerals of the serpentine group, namely chrysotile, and the amphibole group, namely actinolite, amosite, anthophyllite, crocidolite, tremolite or any mixture containing one or more of these minerals;

"asbestos dust" means airborne asbestos particles or deposited asbestos particles liable to become airborne in the work area;

"ASME" means the American Society of Mechanical Engineers;

"ASTM" means the American Society for Testing and Materials;

"calculator" means a calculation tool that can be used to evaluate the daily noise exposure level ($L_{\text{EX,8h}}$ or $L_{\text{ex,8h}}$) for the purpose of reducing the duration of workers' daily exposure to noise;

"CEN" means the European Committee for Standardization;

"CGA" means the Canadian Gas Association or the Association canadienne du gaz;

"confined space" means any space that is completely or partially enclosed, such as a reservoir, a silo, a vat, a hopper, a chamber, a vault, a pit, including a pit and a reception pit for manure, a sewer, a pipe, a chimney, an access shaft, a truck or freight car tank, or a wind turbine blade, and that presents one or more of the following risks due to the confinement:

(1) a risk of asphyxia, intoxication, loss of consciousness or judgment, fire or explosion associated with the atmosphere or internal temperature;

(2) a risk of being buried;

(3) a risk of drowning or being carried away due to the level or flow of a liquid;

"CSA" means the Canadian Standards Association or the Association canadienne de normalisation;

"daily noise exposure level" means the equivalent continuous sound pressure level (dBA) for an 8-hour working day. It results from measurements that include all the types of noise present, including impulse noises;

"dBA" means an A-weighted decibel measurement - the weighting reduces the significance of extreme frequencies, in particular low frequencies below 200 Hz, and increases the significance of frequencies around 2 500 Hz. A-weighting must be used for all measurements to evaluate $L_{\text{EX,8h}}$ or $L_{\text{ex,8h}}$;

"dBC" means a C-weighted decibel measurement - the weighting reduces the significance of frequencies at or below 31 Hz and increases the significance of frequencies at or above 8 000 Hz. C-weighting must be used for all measurements to evaluate peak sound pressure level;

"EN" a European standard issued by the European Committee for Standardisation;

"equivalent continuous sound pressure level (dBA)" means the A-weighted continuous sound pressure level measured over a given period of time. It is identical to the sound pressure level of a constant noise having the same total A-weighted sound energy over the same period of time. It results from measurements that include all the types of noise present, including impulse noises. In the formulas used to calculate daily noise exposure level, it corresponds to $L_{p,A,eqTe}$ or $L_{eq,t}$, which is the A-weighted equivalent continuous sound pressure level for the duration of the working day in hours (T_e or T_w);

"free fall distance" means the vertical distance measured from the beginning of a fall, from the harness D-ring to which the fall arrest connecting device is attached, to the point where the fall arrest system begins to apply force to stop the fall;

"friable material" means material that can be crumbled, pulverized or powdered by hand pressure when dry or that is crumbled, pulverized or powdered;

"heat stress" means heat unbalance in a worker caused by working in a hot environment;

"high-efficiency filter" means any filter capable of filtering particles 0.3 μ m in size at an efficiency rate of at least 99.97%;

"hoisting apparatus" includes cranes, travelling cranes, gantries, winches, blocks, lift trucks, aerial basket lifting devices, work platform lifts, screw-type jacks, rack-type jacks and other similar apparatus but does not include elevators and dumb-waiters;

"impulse noise" means a noise of short duration (generally less than one second), peaking at a high level and characterized by a sharp increase and rapid decrease in sound level. The parameter used to measure an impulse noise is the C-weighted peak sound pressure level;

"IEC" means the International Electrotechnical Commission;

"instructor" means a person in charge of the practical training and communication of theoretical knowledge required for the acquisition of occupational skills;

"ISO" means the International Organization for Standardization;

"lanyard" means a rope or strap fastened at one end to a safety harness and at the other end to an anchorage system or other component of a fall arrest connecting device;

"lifeline" means a synthetic fibre rope, a steel wire rope or a strap attached to an anchorage system and used to guide a rope grab;

"machine" means an assembly, fitted with or intended to be fitted with a drive system other than directly applied human or animal effort, consisting of linked parts or components, at least one of which moves, and which are joined together for a specific application;

"NF EN" means the European standard, the French version of which (NF) is published in France by the Association française de normalisation;

"NFPA" means the National Fire Protection Association;

"NIOSH" means the National Institute for Occupational Safety and Health;

"peak sound pressure level" means the instantaneous peak sound pressure level measured in C-weighted decibels;

"rated load" means the maximum load set by the manufacturer or an engineer;

"respirable asbestos fibre" means asbestos fibre having a ratio of length to diameter of more than 3:1; only fibres longer than 5 µm must be taken into account for measurement purposes;

"respiratory zone" means the zone within a hemisphere having a 300 mm radius extending in front of the face and measured from the midpoint of an imaginary line joining the ears;

"SAE" means the Society of Automotive Engineers;

"safety factor" means the ratio between the rupture load and the working load;

"self-propelled vehicle" means a motor vehicle mounted on wheels, on tracks or on rails, used for the transportation of objects or materials, or for towing or pushing trailers or materials, with the exception of an all-terrain vehicle or an elevating or lifting device;

"stationary work station" means any work station in which a worker is required to perform his duties for at least 4 hours of his working day over a usual work surface of 30 m² or less;

"washroom" means any room containing one or several toilets, urinals, sinks or showers to meet the sanitary needs of the workers of an establishment;

"work station" means any place, including a vehicle occupied by a worker to perform his work;

"ULC" means the Underwriters' Laboratories of Canada or the Laboratoires des assureurs du Canada.

```
O.C. 885-2001, s. 1; O.C. 510-2008, s. 1; O.C. 1411-2018, s. 1;
O.C. 49-2022, s. 1; O.C. 1223-2021 s. 1; O.C. 644-2022, s. 1;
O.C. 821-2023, s. 1; O.C. 781-2021, s. 1; O.C. 43-2023, s. 1; O.C. 1112-
2023, s. 1.
```

3. Purpose: The purpose of this Regulation is to establish standards pertaining in particular to the quality of air, temperature, humidity, heat stress, lighting, noise and other contaminants, sanitary facilities, ventilation, hygiene, sanitation and cleanliness in establishments, area conditions, storage and handling of dangerous substances, machine and tool safety, certain high risk tasks, individual protective equipment and the transportation of workers to ensure the quality of the work environment, to safeguard the health of workers and to ensure their safety and physical well-being.

O.C. 885-2001, s. 3.

4. Employer's obligations: The employer shall comply with the standards set hereunder, with the exception of those of sections 312.5 and 339.

O.C. 885-2001, s. 4; O.C. 425-2010, s. 2.

5. Operational status of equipment: Any equipment used or installed in an establishment for purposes of preventing the emission of gases, dusts, fumes and vapours, to ensure proper conditions for lighting, ventilation, temperature, salubrity and hygiene prescribed hereunder or to ensure that noise or heat stress conditions comply with the requirements hereunder, shall always be in operational

condition and shall give optimal performance during the establishment's business hours in such manner as to provide the performance for which it was designed.

O.C. 885-2001, s. 5.

17. Cleaning: Subject to section 326, the upkeep of the work premises of an establishment shall be ensured through vacuuming, wet mopping or any other method that controls and reduces to a maximum the stirring up of dust. $\overline{O.C. 885-2001, s. 17}$.

40. No worker in an establishment shall be exposed to:

(1) a concentration of airborne oxygen below 19.5% in volume at normal atmospheric pressure;

(2) gases, fumes, vapours, dusts or mists, beyond the limits provided for in Schedule I.

Subparagraph 2 of the first paragraph also applies to a work station located in a vehicle, wherever situated.

O.C. 885-2001, s. 40; O.C. 49-2022, s. 3.

42. Carcinogenic and isocyanate substances: When a worker is exposed to a substance identified in Schedule I as having a known or suspected carcinogenic effect on humans or being diisocyanate or isocyanate oligomers, such exposure shall be reduced to a minimum, even when it remains within the standards in that Schedule.

O.C. 885-2001, s. 42.

44. Methods: Dusts, gases, fumes, vapours and mists found in the workplace environment shall be measured in the respiratory zone of workers or, if this proves to be impossible owing to the lack of equipment for taking a sampling in this zone, then outside the breathing zone but in a place located as close as possible to such zone.

These dusts, gases, fumes, vapours and mists found in the workplace environment shall be sampled and analyzed to obtain an accuracy equivalent to that obtained by applying the methods described in the Sampling Guide for Air Contaminants in the Workplace published by the Institut de recherche Robert-Sauvé en santé et sécurité du travail du Québec.

The sampling strategy for these contaminants shall be carried out in accordance with common practices in industrial hygiene as summarized in the aforementioned guide. $\overline{\text{O.C. 885-2001, s. 44.}}$

45. Respirator: The employer must provide the worker with a respirator in the following cases:

(1) during the period required to implement a measure provided for in section 41;

(2) in case of an emergency where the values provided for in section 40 are not complied with;

(3) if no measure makes it possible to comply with the values provided for in section 40.

O.C. 885-2001, s. 45; O.C. 49-2022, s. 6.

45.1. Every respirator provided by the employer must be certified by the NIOSH or the CSA.

When providing such a device, the employer must draft and apply a respiratory protection program in compliance with CAN/CSA Standard Z94.4-11, Selection, Use and Care of Respirators, as published in September 2016.

O.C. 49-2022, s. 7; O.C. 280-2024, s. 1.

46. Prohibition: Notwithstanding section 45.1, an employer may not provide the worker with a self-contained or air-supplied protective respiratory apparatus equipped with an automatic device which interrupts or restricts the air supply in the part of the apparatus covering the face.

O.C. 885-2001, s. 46; O.C. 49-2022, s. 8.

47. (*Revoked*).

O.C. 885-2001, s. 47; O.C. 49-2022, s. 9.

48. Air supply: Compressed breathing air for supplied-air respirators or self-contained respiratory protective apparatuses must comply with CSA Standard CAN/CSA-Z180.1-00, Compressed Breathing Air and Systems. Systems that produce, store and distribute air must comply with the standard that applies to them.

Samples of compressed breathing air shall be taken and analyzed to obtain an accuracy equivalent to that obtained by applying the methods described in the Sampling Guide for Air Contaminants published by the Institut de recherche Robert-Sauvé en santé et sécurité du travail du Québec. The analyses must be made at least every 6 months, except for ambient air systems. The results of these analyses shall be entered in a register that shall be kept for a period of at least 5 years. Breathable compressed air supply and distribution systems shall be maintained in compliance with the manufacturers' instructions. The date on which such maintenance is performed as well as the name of the person who performed it shall be recorded by the employer in a register that shall be kept for a period of at least 5 years.

O.C. 885-2001, s. 48; O.C. 915-2011, s. 1; O.C. 1104-2015, s. 1; O.C. 49-2022, s. 10.

64. Lead: The recovery of lead or lead products and other related operations shall be performed inside an establishment in compliance with the requirements under section 107.

O.C. 885-2001, s. 64.

65. Floor: In any establishment where lead, mercury or their compounds are handled, stored or used in either solid or liquid form, the floor covering shall be made of a non-porous material.

O.C. 885-2001, s. 65.

66. Protective clothing: The employer shall make sure that workers wear protective clothing used exclusively for their work when performing any of the following activities:

(1) the recovery or melting of lead or lead products;

(2) the manufacturing of lead batteries;

(3) the manufacturing of lead powders or salts, chlorine, fluorescent lamps or caustic soda where workers must handle lead or mercury;

not applicable

(4) any work involving exposure to crocidolite asbestos, amosite or any other type of amphibole;

(5) any work involving exposure to chrysotile asbestos fibres that cannot be contained within the exposure value levels specified in Schedule I.

Before reuse, the employer shall ensure that such clothing has been cleaned with a vacuum equipped with a high-efficiency filter, unless the clothing has been washed. $\overline{O.C. 885-2001, s. 66}$.

107. Local ventilation: Any localized source at a stationary work station that emits dusts, gases, fumes, vapours or mists shall be equipped with a local exhaust ventilation system for trapping the dusts, gases, fumes, vapours or mists at their source.

O.C. 885-2001, s. 107.

108. Recirculation of air: Any air recirculation system shall be designed so that:

(1) the concentration of dusts, fumes, gases, vapours and mists in any work station is lower than the weighted average exposure value permissible in the work environment and the permissible recirculation concentration provided for in Schedule I;

(2) a duct is provided for evacuating contaminated air outside the establishment in case the air filtering system breaks down or is not working properly;

(3) no dusts, fumes or mists are discharged into a room where no dusts, fumes or mists were present before the air recirculation system is put into operation; and

(4) there is no recirculation of gases, vapours, mists, fumes or dusts which are identified under Schedule I as a substance whose recirculation is prohibited.

O.C. 885-2001, s. 108.

109. Fresh air intake: Subject to section 108, an establishment ventilated mechanically shall be equipped with a fresh air intake system designed to replace the volume of air evacuated from the work environment with fresh air from the atmosphere.

The fresh air intake shall be situated so that no air already evacuated from an establishment is reintroduced. $\overline{\text{O.C. 885-2001, s. 109.}}$

110. Adjacent facilities: All establishments shall be designed, built, equipped and operated so that they do not emit gases, dusts, fumes, vapours, odours or mists through ceilings, walls, floors, corridors, stairwells, or freight or passenger elevator hoistways into any building or facility adjacent to the establishment.

O.C. 885-2001, s. 110.

111. Ventilation of change rooms and toilets: During the hours of operation of an establishment, the change rooms and washrooms shall be ventilated toward the outside of the establishment, either naturally in accordance with section 102, or mechanically by extraction in accordance with the standards prescribed in the following table:

Place		Ventilation (in cubic metres of air per hour)
Change rooms	hooks or lockers for street clothes or unsoiled work clothes	18 m ³ /h per square metre of the room's surface area.
	hooks or lockers for damp work clothes (drying facilities)	the greater of: 36 m ³ /h per square metre of the room's surface area, and 12 m ³ /h per locker.
Toilets and urinals		the greater of: $-36 \text{ m}^3/\text{h}$ per square metre of the room's surface area, and $-45 \text{ m}^3/\text{h}$ per toilet or urinal, but not less than 350 m ³ /h.
Showers		the greater of: $-36 \text{ m}^{3}/\text{h}$ per square metre of the room's surface area, and $-90 \text{ m}^{3}/\text{h}$ per shower head, but not less than 350 m ³ /h.

Where a washroom is ventilated naturally, the ventilation area per toilet shall be 0.1 m^2 .

O.C. 885-2001, s. 111.

113. Combustion products: Except in the cases provided for in sections 114 and 115, combustion products vented by the air heating facilities of an establishment shall be evacuated directly outside the establishment by means of a duct.

O.C. 885-2001, s. 113.

114. Infrared heating: In any establishment heated by a gas-fired infrared device, air contaminated by combustion gases shall be evacuated outside by natural or mechanical ventilation at the minimum rate of

O.C. 885-2001, s. 114.

115. (Revoked).

O.C. 885-2001, s. 115; O.C. 889-2020, s. 1.

121. Compulsory measurements: In any establishment employing 50 workers or more where workers are exposed to heat stress conditions in which the heat stress index reaches or exceeds the continuous work curve in the graph in Schedule V, this index shall be measured twice a year, once during the summer, at each work station where the index is reached or exceeded.

The measurements obtained in accordance with the first paragraph shall be entered in a register. The register shall be kept for at least 5 years.

O.C. 885-2001, s. 121.

122. Method: For the purposes of this Regulation, the heat stress index is measured by the Wet BulbGlobe Temperature Index (W.B.G.T. method) as established in Schedule V.

O.C. 885-2001, s. 122.

123. Index exceeds the continuous work curve: In any establishment where workers are exposed to heat stress conditions such that the heat stress index exceeds the continuous work curve in the graph in Schedule V, the employer shall ensure that the workers thus exposed undergo medical supervision and shall provide them with water at a temperature of between 10 °C and 15 °C, and one shower per 15 exposed workers.

O.C. 885-2001, s. 123.

124. Special measures: In any establishment where workers are exposed to heat stress conditions such that the heat stress index exceeds the continuous work curve in the graph in Schedule V, the following measures shall be taken:

(1) re-equip the exposed work station with reflecting screens, additional insulation or ventilation to reduce the heat stress index of the work station to a value less than or equal to the values of the continuous work curve;

(2) if the application of paragraph 1 proves impossible or does not allow the continuous work curve to be reached, control the work load, the time of exposure and the rest time in accordance with the alternate work-rest regimen prescribed for that purpose in Schedule V;

(3) if the application of paragraphs 1 and 2 proves impossible or does not allow the continuous work curves indicated in the graph in Schedule V to be reached or while waiting for the alterations required under paragraph 1 to be done, ensure that the workers wear appropriate individual equipment in accordance with the nature of the heat stress. $\overline{O.C. 885-2001, s. 124}$.

144. Ionizing radiation: Workers exposed to ionizing radiation shall be monitored by dosimetry.

In the event of an overdose, workers thus exposed shall undergo medical examinations at more or less regular intervals, depending on the duration of exposure.

O.C. 885-2001, s. 144.

145. Drinking water: Any establishment shall provide workers with drinking water whose quality complies with the Regulation respecting the quality of drinking water (chapter Q-2, r. 40).

not aplicable

The quantity of drinking water provided to the workers must be sufficient to meet their daily physiological and personal hygiene needs while taking into account, in

Without limiting the scope of the second paragraph, the quantity must at least enable each worker to drink 1 litre of drinking water, wash their hands 4 times over a period of 8 hours and take a shower once a day, when this Regulation requires that it be put at the disposal of the workers. The quantity must also ensure the proper functioning of emergency showers, if applicable.

O.C. 885-2001, s. 145; O.C. 287-2021, s. 2.

146. (*Revoked*).

O.C. 885-2001, s. 146; I.N. 2020-01-01; O.C. 287-2021, s. 3.

148. Bottled water: Any bottled water distributed in an establishment shall comply with the stipulations in the Regulation respecting bottled water (chapter P-29, r. 2). $\overline{O.C. 885-2001, s. 148}$.

149. Distributors: All establishments shall be equipped with distributors of drinking water intended for consumption by the workers in a proportion of one distributor per group of 75 workers and an additional distributor for any fraction of that number above 75 workers. In an establishment with less than 75 workers, at least one drinking water distributor shall be provided.

Drinking water distributors shall be easy to clean and made of leakproof material. They shall be kept free from any source of water contamination.

O.C. 885-2001, s. 149.

150. Water unsafe for drinking: Any drinking water distribution system intended for workers' consumption shall be designed and installed to eliminate any possibility of cross-connection or contamination with any piping system likely to contain water that is unsafe for drinking.

Any tap for water that is unsafe for drinking shall be identified.

O.C. 885-2001, s. 150.

151. Paper cups: Except where workers are provided with water fountains, they shall have at their disposal sanitary individual disposable paper cups.

The use of a common glass or cup is prohibited.

When workers are provided with paper cups, a refuse container shall be placed less than 2 m from the drinking water distributor.

O.C. 885-2001, s. 151.

Regulation respecting prevention mechanisms specific to construction sites

chapter S-2.1, r. 8.2

Act respecting occupational health and safety (chapter S-2.1, s. 58).

CHAPTER I SCOPE

S.Q. 2021, c. 27, s. 243.

1. For the purposes of the Act respecting occupational health and safety (chapter S-2.1), this Regulation determines the rules applicable on construction sites with regard to the job-site committee, the health and safety representative and the health and safety coordinator.

S.Q. 2021, c. 27, s. 243.

CHAPTER II JOB-SITE COMMITTEE

S.Q. 2021, c. 27, s. 243.

DIVISION I

COMPOSITION OF JOB-SITE COMMITTEE AND DESIGNATION OF MEMBERS

S.Q. 2021, c. 27, s. 243.

2. The maximum number of employers' representatives on the job-site committee must be equal to the number of health and safety representatives and representatives of each of the representative associations that sit on the committee.

If the number of employers present on the construction site exceeds the maximum number of representatives prescribed in the first paragraph, the employers' representatives on the committee are respectively those of the employers that employ the greatest number of workers present on the construction site.

S.Q. 2021, c. 27, s. 243.

3. Where two or more health and safety representatives or two or more health and safety coordinators are designated on a construction site, the number of representatives or coordinators on the committee is equal to the minimum number prescribed in sections 13 and 16, according to the category of construction site.

S.Q. 2021, c. 27, s. 243.

4. The health and safety representatives on the jobsite committee are designated by all the representative associations.

Failing that, they are designated by a majority of the construction workers present on the construction site. $\overline{S.Q. 2021, c. 27, s. 243}$.

DIVISION II RULES OF OPERATION FOR JOB-SITE COMMITTEE

S.Q. 2021, c. 27, s. 243.

5. The job-site committee holds its first meeting within 14 days after the date on which the work begins. S.Q. 2021, c. 27, s. 243.

6. Despite the minimum frequency of meetings prescribed in the first paragraph of section 207 of the Act, the job-site committee of a construction site employing 100 workers or more must meet at least once a week.

S.Q. 2021, c. 27, s. 243.

7. The agenda of a job-site committee meeting is determined by the principal contractor.

Any committee member may, at the beginning of the meeting and with the other members' agreement, propose amendments to the agenda.

S.Q. 2021, c. 27, s. 243.

8. The quorum at a meeting is at least one representative of the principal contractor, at least one employer's representative and at least half the members referred to in paragraphs 3 and 4 of section 205 of the Act who represent workers.

S.Q. 2021, c. 27, s. 243.

9. Any vacancy on the job-site committee must be filled not later than 14 days after the committee is informed of it if the construction site employs at least 20 workers or not later than 7 days if the construction site employs at least 100 workers.

A vacancy is filled according to the method of designation prescribed for designating the member to be replaced, if any.

S.Q. 2021, c. 27, s. 243.

10. The principal contractor must draw up the minutes of the job-site committee's meetings.

At each meeting, the committee adopts the minutes of the previous meeting. Adopted minutes are kept by the principal contractor, in a register established for that purpose, for at least one year after the date on which the work ends.

Committee members may, by request to the principal contractor, obtain copies of the committee's minutes. $\overline{S.Q. 2021, c. 27, s. 243}$.

DIVISION III

TRAINING OF JOB-SITE COMMITTEE MEMBERS S.Q. 2021, c. 27, s. 243.

11. A job-site committee member must obtain a certificate for at least one hour of theoretical training issued by the Commission or by a body recognized by it.

The training must pertain, in particular, to the following subjects:

(1) prevention mechanisms applicable on a construction site;

(2) the role of the job-site committee and its rules of operation;

(3) follow-up on the prevention program;

(4) analysis of accident notices and follow-up on them;

(5) follow-up on suggestions and complaints regarding occupational health and safety received from construction workers, representative associations, the joint sector-based construction association referred to in section 99 of the Act, employers and the principal contractor; and

(6) follow-up on reports regarding inspections carried out on the construction site.

A member who holds a health and safety coordinator training certificate or a health and safety representative training certificate under section 15 is not required to take such training.

S.Q. 2021, c. 27, s. 243.

CHAPTER III HEALTH AND SAFETY REPRESENTATIVE

S.Q. 2021, c. 27, s. 243.

12. The minimum amount of time that a health and safety representative may devote daily to the exercise of his or her functions, except the functions referred to in paragraphs 2, 6 and 7 of section 210 of the Act, is as follows, according to the number of workers present on the construction site:

(1) 10 to 24 workers: 1 hour;

(2) 25 to 49 workers: 3 hours;

(3) 50 to 74 workers: 4 hours;

(4) 75 to 99 workers: 6 hours; and

(5) 100 workers and more: 8 hours.

S.Q. 2021, c. 27, s. 243.

13. The minimum number of health and safety representatives designated in accordance with section 212.1 of the Act is as follows, according to the number of workers present on the construction site:

(1) 100 to 199 workers: 1;

(2) 200 to 599 workers: 2;

(3) 600 to 899 workers: 3;

(4) 900 to 1,199 workers: 4; and

(5) 1,200 workers and more: 5. <u>8.0. 2021, c. 27, s. 243.</u>

14. A health and safety representative designated in accordance with section 209 of the Act must obtain a certificate for at least three hours of theoretical training issued by the Commission or by a body recognized by it.

The training must pertain, in particular, to the following subjects:

(1) prevention mechanisms applicable on a construction site;

(2) the representative's role, functions and responsibilities;

(3) inspection of workplaces;

(4) assistance to workers in the exercise of their rights recognized by the Act and the regulations;

(5) the representative's role during an inspector's visit; and

(6) accident investigation and analysis of reported incidents.

S.Q. 2021, c. 27, s. 243.

15. A health and safety representative designated in accordance with section 212.1 of the Act must obtain a certificate for at least 40 hours of theoretical training issued by the Commission or by a body recognized by it.

In addition to the subjects listed in the second paragraph of section 14, the training must pertain to the prevention program and the operation of a job-site committee.

S.Q. 2021, c. 27, s. 243.

CHAPTER IV

HEALTH AND SAFETY COORDINATOR

S.Q. 2021, c. 27, s. 243.

16. The minimum number of health and safety coordinators designated in accordance with section 215.1 of the Act is as follows, according to the number of workers present on the construction site:

- (1) 100 to 199 workers: 1;
- (2) 200 to 599 workers: 2;
- (3) 600 to 899 workers: 3;
- (4) 900 to 1,199 workers: 4; and

(5) 1,200 workers and more: 5.

S.Q. 2021, c. 27, s. 243.

17. A health and safety coordinator must obtain a certificate for at least 240 hours of theoretical training issued by the Commission or by a body recognized by it.

The training must pertain, in particular, to the following subjects:

(1) the legislative and regulatory framework for occupational health and safety applicable to a construction site; (2) prevention mechanisms applicable on a construction site;

(3) the coordinator's role and general functions, including coordinating a job-site committee;

(4) preparing and updating a prevention program specific to a construction site;

(5) the coordinator's role during an inspector's visit on the construction site;

(6) the main safety measures applicable on a construction site, taking into account the priorities for action established by the Commission;

(7) the main occupational health rules applicable on a construction site;

(8) the occupational health and safety management audit;

(9) inspection of workplaces;

(10) accident investigation and analysis of reported incidents;

(11) preparation of work directives specific to a construction site; and

(12) interpersonal relations and communication skills. s.q. 2021, c. 27, s. 243.

CHAPTER V

TRANSITIONAL PROVISION

S.Q. 2021, c. 27, s. 243.

18. A person who, on 31 December 2022, holds an attestation of safety officer delivered by the Commission under paragraph c of subsection 2 of section 2.5.4 of the Safety Code for the construction industry (chapter S-2.1, r. 4) and who is designated health and safety representative or health and safety coordinator is not required to obtain the training certificates required under sections 15 and 17. $\overline{s.0.2021, c. 27, s. 243}$.